Linköping Studies in Science and Technology. Dissertations. No. 985

# **Linear Models of Nonlinear Systems**

**Martin Enqvist** 



Department of Electrical Engineering Linköpings universitet, SE–581 83 Linköping, Sweden Linköping 2005

## Linear Models of Nonlinear Systems

© 2005 Martin Enqvist

maren@isy.liu.se www.control.isy.liu.se Division of Automatic Control Department of Electrical Engineering Linköpings universitet SE-581 83 Linköping Sweden

ISBN 91-85457-64-7

ISSN 0345-7524

Printed by LiU-Tryck, Linköping, Sweden 2005

To Ann-Christine

# Abstract

Linear time-invariant approximations of nonlinear systems are used in many applications and can be obtained in several ways. For example, using system identification and the prediction-error method, it is always possible to estimate a linear model without considering the fact that the input and output measurements in many cases come from a nonlinear system. One of the main objectives of this thesis is to explain some properties of such approximate models.

More specifically, linear time-invariant models that are optimal approximations in the sense that they minimize a mean-square error criterion are considered. Linear models, both with and without a noise description, are studied. Some interesting, but in applications usually undesirable, properties of such optimal models are pointed out. It is shown that the optimal linear model can be very sensitive to small nonlinearities. Hence, the linear approximation of an almost linear system can be useless for some applications, such as robust control design. Furthermore, it is shown that standard validation methods, designed for identification of linear systems, cannot always be used to validate an optimal linear approximation of a nonlinear system.

In order to improve the models, conditions on the input signal that imply various useful properties of the linear approximations are given. It is shown, for instance, that minimum phase filtered white noise in many senses is a good choice of input signal. Furthermore, the class of separable signals is studied in detail. This class contains Gaussian signals and it turns out that these signals are especially useful for obtaining approximations of generalized Wiener-Hammerstein systems. It is also shown that some random multisine signals are separable. In addition, some theoretical results about almost linear systems are presented.

In standard methods for robust control design, the size of the model error is assumed to be known for all input signals. However, in many situations, this is not a realistic assumption when a nonlinear system is approximated with a linear model. In this thesis, it is described how robust control design of some nonlinear systems can be performed based on a discrete-time linear model and a model error model valid only for bounded inputs.

It is sometimes undesirable that small nonlinearities in a system influence the linear approximation of it. In some cases, this influence can be reduced if a small nonlinearity is included in the model. In this thesis, an identification method with this option is presented for nonlinear autoregressive systems with external inputs. Using this method, models with a parametric linear part and a nonparametric Lipschitz continuous nonlinear part can be estimated by solving a convex optimization problem.

# Sammanfattning

Linjära tidsinvarianta approximationer av olinjära system har många användningsområden och kan tas fram på flera sätt. Om man har mätningar av in- och utsignalerna från ett olinjärt system kan man till exempel använda systemidentifiering och prediktionsfelsmetoden för att skatta en linjär modell utan att ta hänsyn till att systemet egentligen är olinjärt. Ett av huvudmålen med den här avhandlingen är att beskriva egenskaper för sådana approximativa modeller.

Framförallt studeras linjära tidsinvarianta modeller som är optimala approximationer i meningen att de minimerar ett kriterium baserat på medelkvadratfelet. Brusmodeller kan inkluderas i dessa modelltyper och både fallet med och utan brusmodell studeras här. Modeller som är optimala i medelkvadratfelsmening visar sig kunna uppvisa ett antal intressanta, men ibland oönskade, egenskaper. Bland annat visas det att en optimal linjär modell kan vara mycket känslig för små olinjäriteter. Denna känslighet är inte önskvärd i de flesta tillämpningar och innebär att en linjär approximation av ett nästan linjärt system kan vara oanvändbar för till exempel robust reglerdesign. Vidare visas det att en del valideringsmetoder som är framtagna för linjära system inte alltid kan användas för validering av linjära approximationer av olinjära system.

Man kan dock göra de optimala linjära modellerna mer användbara genom att välja lämpliga insignaler. Bland annat visas det att minfasfiltrerat vitt brus i många avseenden är ett bra val av insignal. Klassen av separabla signaler detaljstuderas också. Denna klass innehåller till exempel alla gaussiska signaler och just dessa signaler visar sig vara speciellt användbara för att ta fram approximationer av generaliserade wiener-hammersteinsystem. Dessutom visas det att en viss typ av slumpmässiga multisinussignaler är separabel. Några teoretiska resultat om nästan linjära system presenteras också.

De flesta metoder för robust reglerdesign kan bara användas om storleken på modellfelet är känd för alla tänkbara insignaler. Detta är emellertid ofta inte realistiskt när ett olinjärt system approximeras med en linjär modell. I denna avhandling beskrivs därför ett alternativt sätt att göra en robust reglerdesign baserat på en tidsdiskret modell och en modellfelsmodell som bara är giltig för begränsade insignaler.

Ibland skulle det vara önskvärt om en linjär modell av ett system inte påverkades av förekomsten av små olinjäriteter i systemet. Denna oönskade påverkan kan i vissa fall reduceras om en liten olinjär term tas med i modellen. En identifieringsmetod för olinjära autoregressiva system med externa insignaler där denna möjlighet finns beskrivs här. Med hjälp av denna metod kan modeller som består av en parametrisk linjär del och en ickeparametrisk lipschitzkontinuerlig olinjär del skattas genom att man löser ett konvext optimeringsproblem.

# Acknowledgments

First of all, I would like to thank my supervisor Professor Lennart Ljung for introducing me to the fascinating field of system identification and for giving me excellent guidance during the work with this thesis. I have really enjoyed discussing my research topics with him and his wide knowledge and truly remarkable intuition for identification and control has been a great source of inspiration for me. Furthermore, spending time with Lennart Ljung is also nice because he has many interests besides research. During the last five years, I have had the privilege to learn about many interesting and useful topics such as where to find nice beaches in the world, how to deal with pickpockets and the habits of parrots.

I would also like to thank Dr. Jacob Roll, Professor Torkel Glad and Professor Anders Helmersson for our fruitful cooperation on some of the publications that this thesis is based on.

I am very grateful to Markus Gerdin, Dr. Jacob Roll, Johan Sjöberg and Erik Wernholt for proofreading this thesis and to Dr. Ola Härkegård and Dr. Mikael Norrlöf for proofreading earlier versions of the material in it. They have all given me many valuable comments on my work and suggestions for improvements.

Furthermore, I would like to thank Gustaf Hendeby for helping me with many LATEX issues and for designing a nice thesis layout and Henrik Tidefelt for developing the convenient MetaPost block diagram package.

I would also like to thank Ulla Salaneck for helping me with all sorts of things and everyone else in the Automatic Control group for providing such a nice working atmosphere. Besides the ones already mentioned, I would in particular like to thank also Frida Eng, Erik Geijer Lundin, Dr. David Lindgren and Thomas Schön for always having time to discuss various issues with me.

In addition, I would like to thank Professor Johan Schoukens and Professor Pertti Mäkilä for giving me valuable comments on my work and for letting me use some of their time for discussions. They have really helped me understand my field of research better.

This work has been supported by the Swedish Research Council, which is hereby gratefully acknowledged.

I would also like to thank my parents and my brother for supporting me and for always being interested in what I am doing. Finally, I would like to thank Ann-Christine for all the encouragement, support and love she has given me. I love you!

> Martin Enqvist Linköping, October 2005

# Contents

1	Intr	duction	1
	1.1	Systems and Models	2
	1.2	Motivating Examples	3
	1.3	Outline of the Thesis	6
	1.4	Contributions	7
2	Prel	minaries	11
	2.1	Linear Systems and Stochastic Processes	11
	2.2	Nonlinear Systems	15
	2.3	System Identification	16
		2.3.1 Prediction-Error Methods	16
		2.3.2 A Prediction-Error Method for Random Multisines	18
	2.4	Separable Processes	21
3	Met	ods for Linearization of Nonlinear Systems	27
	3.1	Deterministic Approaches	27
	3.2	Stochastic Approaches	29
		3.2.1 Results for Static Nonlinearities	29
		3.2.2 Results for Hammerstein and Wiener Systems	30
		3.2.3 Results for General Nonlinear Systems	31
Ι	LT	-SOEs	33
Ι	LT	-SOEs	33
I 4	LT The	-SOEs Notion of LTI Second Order Equivalents	33 35
I 4	<b>LT</b> <b>The</b> 4.1	-SOEs Notion of LTI Second Order Equivalents Assumptions on the Input and Output Signals	<b>33</b> <b>35</b> 35

	4.3	The Output Error Model Type for Periodic Inputs	46
	4.4	The General Error Model Type	51
	4.5	Interpretations of the GE-LTI-SOE	60
	4.6	Assumptions on the Noise	64
5	Basi	c Properties of LTI-SOEs	67
	5.1	Additive Noise	67
	5.2	Even and Odd Nonlinearities	68
	5.3	Sums of Nonlinear Systems	71
	5.4	Minimum Phase Input Filters	72
	5.5	Properties for Regular OE-LTI-SOEs	75
		5.5.1 Optimality Properties	75
		5.5.2 Spectral and Residual Analysis	77
		5.5.3 Closed-loop Identification	79
	5.6	LTI-SOEs with a General Error Model	81
6	NFI	R Systems with Separable Inputs	85
	6.1	Separability	86
	6.2	Separable Random Multisines	89
	6.3	Higher Order Separability	91
		6.3.1 Definition and Basic Properties	91
		6.3.2 Higher Order Separability and OE-LTI-SOEs	92
		6.3.3 Identification of Generalized Hammerstein Systems	99
7	NFI	R Systems with Gaussian Inputs 10	01
	7.1	OE-LTI-SOEs of NFIR Systems with Gaussian Input Processes 10	02
	7.2	Applications	05
		7.2.1 Structure Identification of NFIR Systems	05
		7.2.2 Identification of Generalized Wiener-Hammerstein Systems 10	06
8	Alm	ost Linear Systems	11
	8.1	Almost Linear Systems	11
	8.2	A Convergence Result	17
	8.3	Almost Linear NFIR Systems	20
9	Disc	ussion 1	23
II	Id	entification and Control 12	25
10	Rob	ust Control 12	27
	10.1	Robust Control for Constrained Inputs	29
	10.2	Robust Control Using LTI Models	32
	10.3	Discussion	36

11	Mixed Parametric Nonparametric Identification	137
	11.1 Identification of NARX and NOE Systems	139
	11.2 Consistency	142
	11.3 Examples	147
	11.3.1 NARX Models	147
	11.3.2 NOE Models	149
	11.4 A General Perspective	153
	11.5 Discussion	154
12	Conclusions	155
Bil	bliography	157
A	Calculations for Example 4.3	165
В	MATLAB Code	169
	B.1 Example 5.2	169
	B.2 Example 7.2	170
	B.3 Example 7.3	170

# Notation

# Symbols, Operators and Functions

$\mathbb{N}$	the set of natural numbers $(0 \in \mathbb{N})$
$\mathbb{Z}$	the set of integers
$\mathbb{Z}_+$	the set of positive integers
$\mathbb{R}$	the set of real numbers
$\mathbb{C}$	the set of complex numbers
$L^1(\mathbb{R}^n)$	the space of functions f such that $\int_{\mathbb{R}^n}  f(x)  dx < \infty$
$\in$	belongs to
$A \subset B$	A is a subset of B
$A \setminus B$	set difference, $\{x \mid x \in A \land x \notin B\}$
$\operatorname{card}(A)$	the cardinality of the set $A$
	equal by definition
$\prec$	component-wise inequality (for vectors),
	negative semidefiniteness (for a matrix A with $A \preccurlyeq 0$ )
$\arg\min f(x)$	value of x that minimizes $f(x)$
v	$\sqrt{v^T v}$
q	the shift operator, $qu(t) = u(t+1)$
$(x(t))_{t=0}^M$	the sequence $x(0), x(1), \ldots, x(M)$
$\ x\ $	$\sqrt{\sum_{t=0}^{\infty} x(t)^T x(t)}$
$\ x\ _N$	$\sqrt{\sum_{t=0}^{N} x(t)^T x(t)}$
$[G(z)]_{\text{causal}}$	the causal part of the transfer function $G(z)$
$\mathrm{E}(X)$	expected value of the random variable $X$

E(X Y)	conditional expectation of $X$ given $Y$
$R_u(\tau)$	covariance function of the signal $u$
$R_{yu}(\tau)$	cross-covariance function of the signals $y$ and $u$
$\Phi_u(z)$	z-spectrum of the signal u
$\Phi_u(e^{i\omega})$	spectral density function of the signal $u$
$\Phi_{yu}(z)$	z-cross-spectrum of the signals $y$ and $u$
$\Phi_{yu}(e^{i\omega})$	cross-spectral density function of the signals $y$ and $u$
$\theta$	parameter vector
r(t)	reference signal at time $t$
u(t)	input signal at time t
y(t)	output signal at time t
$y_{nf}(t)$	noise-free output signal at time $t$
w(t)	noise signal at time $t$
$\eta_0(t)$	OE-LTI-SOE residual at time $t$
$\varepsilon_0(t)$	GE-LTI-SOE residual at time $t$
$G_{0,OE}(z)$	Output Error Linear Time-Invariant Second Order Equivalent
$G_{0,GE}(z)$	system model part of the General Error Linear Time-Invariant Sec-
	ond Order Equivalent
$H_{0,GE}(z)$	noise model part of the General Error Linear Time-Invariant Second
	Order Equivalent

# Abbreviations and Acronyms

AutoRegressive (system) with eXternal input
Discrete Fourier Transform
Finite Impulse Response
General Error
Linear Time-Invariant
Nonlinear AutoRegressive (system) with eXternal input
Nonlinear Finite Impulse Response
Nonlinear Output Error
Output Error
Quadratic Programming
Second Order Equivalent
with probability one

# Assumptions

A1	Standard assumptions on the input (see page 36)
A2	Standard assumptions on the output (see page 36)
A3	Assumptions on the input and output signals for periodic inputs (see page 36)
A4	Assumption used in the definition of GE-LTI-SOEs (see page 37)
A5	Assumptions on the noise (see page 64)
A6	Assumptions on Gaussian probability density functions (see page 102)
A7	Assumptions on the input and output signals for Gaussian inputs (see page 102)

# Introduction

Mathematical modeling of real-life systems is a very common methodology in science and engineering. It is used both as a means for achieving deeper knowledge about a system and as an engineering tool, e.g., as a basis for simulations or for design of controllers. Sometimes, it is possible to construct a model of a system from physical laws and principles. However, in other cases this is not possible, either because of a lack of knowledge of the studied system or because physical modeling is considered too time consuming. In these cases, system identification can be a way of solving the modeling problem.

System identification deals with the problem of how to estimate a model of a system from measured input and output signals. Usually, only estimation problems for dynamic systems, i.e., systems with some kind of memory, are called system identification. A system can be linear or nonlinear and, depending on the type of system, linear or nonlinear models can be estimated. In practice, linear models are very common and they are often used also when the true system is nonlinear. In these cases, the model can only give an approximate description of the system. It is therefore interesting to understand in what way an estimated linear model can approximate a nonlinear system and how this approximation depends on the properties of the true nonlinear system and of the input signal used. The main objective of this thesis is to give some answers to this problem. Furthermore, some robustness issues concerning system identification and automatic control are also discussed.

The field of automatic control concerns methods for changing the behavior of a dynamic system and a device designed for this purpose is called a controller. For example, controllers can be used to stabilize a system, to make a system less sensitive to disturbances or to change the response of a system to an external signal. Usually, a control method is designed for a particular class of control problems, which is often defined by a number of mathematical assumptions about the system. However, many methods are applied also to real-life systems that do not satisfy all these assumptions. Hence, it is important to investigate the robustness of a control method with respect to erroneous assumptions about the system. When a controller is designed based on a mathematical model, it is said to be robust towards model errors if the differences between the model and the true system cannot cause instability. This type of robustness of a controller is discussed here.

This chapter contains a brief discussion about systems and models and some motivating examples. These examples describe some of the phenomena or problems that can occur when a linear model of a nonlinear system is estimated. Furthermore, an outline of the thesis is given and finally, the main contributions are listed.

## 1.1 Systems and Models

A very important notion in system identification is the difference between a *system* and a *model*. In a wide sense, a system is any kind of physically or conceptually bounded object. Examples of systems are the solar system, a human brain cell and an electrical motor. A system is usually affected by external signals. For example, the solar system is affected by the gravity of other stars, a human brain cell is affected by neighboring cells and by the composition of the blood, and an electrical motor is affected by the voltage over its winding. External signals with a desired effect on the system are called *inputs*, while other, undesired signals that affect the system are called *disturbances*. Measurable signals that describe some property of the system are called *outputs*. Note that also disturbances can be measurable and that the classification of external signals as either inputs or disturbances is somewhat arbitrary. However, for a particular application it is usually easy to distinguish inputs and disturbances.

From a control engineering perspective, a system is some device whose behavior we would like to make more intelligent in some way. This can be done by designing a controller that measures the outputs from the system and then alters the input signals in order to achieve the desired behavior. From a control scientist's point of view, the only systems of interest are those with both input and output signals. In most parts of this thesis, only scalar systems, i.e., systems with one input and one output, are considered.

A mathematical description of a system is called a model. Whenever a system corresponds to a real-life object, it cannot be described exactly and any model of it will thus contain errors. Only in constructed examples, it is possible to give an exact description of the true system. However, any reference to the system always concerns the actual true system. Hence we can talk about model errors but not about system errors.

The common practice to approximate nonlinear systems using linear models can be done in many ways. For example, differentiation can be used to linearize a nonlinear system description locally, or some kind of linear equivalent of a nonlinear system can be derived for a particular input. In this thesis, we will investigate the latter of these two approaches. More specifically, we will study the behavior of linear model estimates obtained by system identification using input and output data from nonlinear systems. The system identification method that will be used here is the well-known *prediction-error method* (see Section 2.3), and we will only investigate its behavior when the number of measurements tends to infinity.

The prediction-error method can be used to compute estimates of some parameters  $\theta$ 

in a general linear model

$$y(t) = G(q, \theta)u(t) + H(q, \theta)e(t)$$

of a system with input u(t) and output y(t). Here, q denotes the shift operator, qu(t) = u(t+1),  $G(q, \theta)$  is the linear model from input to output and  $H(q, \theta)$  is a model of how the noise e(t) affects the output. Both these models are parameterized by the vector  $\theta$ . It will be assumed that both  $H^{-1}(q, \theta)G(q, \theta)$  and  $H^{-1}(q, \theta)$ , with  $H^{-1}(q, \theta) = 1/H(q, \theta)$ , are stable models (see Section 2.1). It can be shown (Ljung, 1978) that the predictionerror parameter estimates under rather general conditions will converge to the parameters that minimize a *mean-square error criterion*  $E((H^{-1}(q, \theta)(y(t) - G(q, \theta)u(t)))^2)$ , where E(x) denotes the expected value of the random value x.

In the special case when the noise model is equal to one, the mean-square error optimal model G(q) will here be called the *Output Error Linear Time-Invariant Second Order Equivalent* (OE-LTI-SOE) and it will be denoted  $G_{0,OE}(q)$ . The corresponding mean-square error optimal model for a general noise model will be called the *General Error Linear Time-Invariant Second Order Equivalent* (GE-LTI-SOE) and it will be denoted  $(G_{0,GE}(q), H_{0,GE}(q))$ . In the next section, some motivating examples that illustrate the properties of OE-LTI-SOEs and GE-LTI-SOEs will be presented.

## 1.2 Motivating Examples

Although the use of linear models of nonlinear systems is straightforward in some cases, it can sometimes give rise to rather nonintuitive phenomena. This is shown in the following examples.

— Example 1.1 —

Consider the simple static nonlinear system

$$y(t) = u(t)^3. (1.1)$$

Intuitively, the best linear approximation of this system would be a static linear system  $y(t) = c_0 u(t)$ , where  $c_0$  is some constant. However, this is not always the case. Let the input to the system (1.1) be

$$u(t) = e(t) + \frac{1}{2}e(t-1),$$

where e(t) is a sequence of independent random variables with uniform distribution over the interval [-1, 1]. In this case, it turns out that the OE-LTI-SOE of the system (1.1) is

$$G_{0,OE}(q) = \frac{0.85 + 0.575q^{-1}}{1 + 0.5q^{-1}}.$$

Hence, a static nonlinear system can have a nonstatic OE-LTI-SOE.

1

#### \_\_\_\_\_ Example 1.2 \_\_\_\_\_

Let the input signal to (1.1) be generated in a different way according to

$$u(t) = \frac{1}{2}e(t) + e(t-1), \qquad (1.2)$$

where e(t) is the same signal as in Example 1.1. In this way, this input will have the same spectral density  $\Phi_u(e^{i\omega})$  as the one in the previous example. However, the OE-LTI-SOE of the system (1.1) for the input (1.2) is

$$G_{0,OE}(q) = \frac{0.925 + 0.425q^{-1}}{1 + 0.5q^{-1}}$$

Hence, a nonlinear system can have different OE-LTI-SOEs for two input signals with equal spectral densities.

# \_\_\_\_\_ Example 1.3\_\_\_\_\_\_

Consider the static nonlinear system

$$y(t) = u(t)^2 - 3$$

with the input

$$u(t) = e(t) + e(t-1)^2 - 1,$$

where e(t) here is a white Gaussian process with zero mean and unit variance. The OE-LTI-SOE of this system is

$$G_{0,OE}(q) = \frac{8}{3} \approx 2.6667$$

while the GE-LTI-SOE is

$$G_{0,GE}(q) = \frac{\sqrt{4161} - 33}{12} \approx 2.6255,$$
  
$$H_{0,GE}(q) = 1 + \frac{65 - \sqrt{4161}}{8}q^{-1}.$$

As can be seen from these expressions,  $G_{0,OE}(q) \neq G_{0,GE}(q)$  despite the fact that the system operates in open loop.

Hence, the OE-LTI-SOE  $G_{0,OE}(q)$  of an open-loop nonlinear system can be different from  $G_{0,GE}(q)$  in the corresponding GE-LTI-SOE.

\_\_\_\_\_Example 1.4 \_\_\_\_\_

Consider the nonlinear system

$$y(t) = y_l(t) + 0.01y_n(t),$$
  
 $y_l(t) = u(t),$   
 $y_n(t) = u(t)^3.$ 



(a) The output y(t) (dashed) of the nonlinear system in Example 1.4 and the output  $y_l(t) = u(t)$  (solid) of the linear part of that system for a particular realization of the input signal.



(**b**) The frequency response of the OE-LTI-SOE (dashed) and of the linear part (solid) of the nonlinear system in Example 1.4

**Figure 1.1:** The frequency response of the OE-LTI-SOE can be far from the response of the linear part of the system also when the nonlinear contributions to the output are small.

The output from this system consists of a linear part,  $y_l(t) = G_l(q)u(t)$  with  $G_l(q) = 1$ , and a nonlinear part,  $0.01y_n(t) = 0.01u(t)^3$ . Let the input signal be

$$u(t) = (1 - 2cq^{-1} + c^2q^{-2})e(t),$$

where c = 0.99 and where e(t) is a white noise process with uniform distribution over the interval [-1, 1]. For this input, it is hard to distinguish the output y(t) of the nonlinear system from the output  $y_l(t)$  of the linear part of the system. This can be seen in Figure 1.1a for a particular realization of the input signal. However, the small differences between the output signals y(t) and  $y_l(t)$  will make the OE-LTI-SOE very different from the linear part  $G_l$  of the system. This difference can be seen in Figure 1.1b.

Hence, the distance between the OE-LTI-SOE and the linear part of the true system can be large also when the nonlinearities are small.

As can be seen in the previous examples, the OE-LTI-SOE of a nonlinear system is input dependent. Furthermore, there is no guarantee that the OE-LTI-SOE and the GE-LTI-SOE will be equal even for an open-loop nonlinear system. Neither will the OE-LTI-SOE always be close to the linear part of the system.

In particular, this last property can in some circumstances be undesirable, for example if the OE-LTI-SOE is supposed to be used as a basis for robust control design. Such a design puts restrictions on the control laws in order to guarantee the stability of the resulting true closed-loop system, despite the presence of model errors. A drawback with a model that is far from the linear part of an almost linear system is that the gain of the model errors might be unnecessarily large.

A robust control design based on a model with large model errors usually implies that the restrictions on the control laws will be rather hard. Hence, the use of an OE-LTI-SOE with large model errors can result in a rather poor control performance for the true system. It is thus interesting to understand under which circumstances the OE-LTI-SOE will be close to the linear part of the system when the nonlinearities are small. Furthermore, it would be interesting to have an identification method such that small nonlinearities can be ignored when a linear model is desired. A method with this option is discussed in this thesis.

Examples 1.1 and 1.2 show that the OE-LTI-SOE of a nonlinear system will be input dependent. One could argue that this input dependency is a problem and that an LTI approximation of a nonlinear system should not be derived for a particular input or class of inputs. However, as the following example illustrates, it is not realistic to believe that a linear model can be a good approximation of a nonlinear system for *all* inputs.

– Example 1.5 –

Consider the nonlinear system

$$y(t) = \begin{cases} 1, & u(t) > 1, \\ u(t), & |u(t)| \le 1, \\ -1, & u(t) < -1 \end{cases}$$

and assume that a linear approximation

$$\hat{y}(t) = b_0 u(t)$$

of this system is desired. Assume that we want this approximation to be optimal in the sense that

$$\sup_{u(t)\in\mathbb{R}}|y(t)-\hat{y}(t)|$$

is minimized. In this case, it is easy to see that the optimal model is  $b_0 = 0$ . Of course, this is not a very useful model.

Hence, a linear model of a nonlinear system should typically be derived and used only for a restricted class of inputs.

Examples 1.1-1.5 will be discussed in more detail later in this thesis (see Examples 4.2, 5.1, 4.3 and 8.1 and Chapter 10, respectively). Furthermore, some conditions and methods that prevent the behaviors shown in these examples will be presented.

## 1.3 Outline of the Thesis

Most of the results in this thesis concern systems with stationary stochastic input and output signals. Some background material about such signals and about linear and non-linear systems can be found in Chapter 2. This chapter contains also a brief description of system identification using the prediction-error method and an introduction to separable

processes. An overview of some existing methods for linearization of nonlinear systems can be found in Chapter 3.

The rest of this thesis is divided into two parts. The first part comprises Chapters 4 to 9 and concerns analysis of LTI-SOEs of nonlinear systems. The second part comprises Chapters 10 and 11 and concerns robustness issues for control design and system identification using linear models.

The linearization approach used in this thesis is based on minimization of the meansquare error. The LTI-SOEs obtained by this approach are described in Chapter 4 and some basic properties of these models are discussed in Chapter 5. There, it is also shown that a minimum phase filtered white noise input implies useful properties for the LTI-SOE of a nonlinear system.

Furthermore, it turns out that the class of separable inputs is especially useful for LTI approximations of nonlinear systems. Some results for these inputs are described in Chapter 6 while Gaussian inputs, which belong to the class of separable inputs, are discussed in Chapter 7. This chapter contains also some results about LTI-SOEs of generalized Wiener-Hammerstein systems. Furthermore, LTI approximations of almost linear systems are studied in Chapter 8 and the first part of the thesis is summarized with a discussion about different input signals in Chapter 9.

The second part of the thesis is more focused on methods. An approach to robust control using realistic model error models is described in Chapter 10. Furthermore, an identification method that sometimes can reduce the sensitivity of an estimated LTI model to small nonlinearities is discussed in Chapter 11.

Some final conclusions concerning the previously presented topics are given in Chapter 12.

## 1.4 Contributions

The main objective of this thesis is to explain some of the behavior of LTI-SOEs of nonlinear systems and to investigate some robustness issues concerning control and identification using linear models of nonlinear systems. For example, the phenomena shown in Examples 1.1-1.4 are discussed.

From a practical point of view, there are four contributions in this thesis that probably are more important than the others. The first one is the observation described in Sections 5.4 and 5.5 that minimum phase filtered white noise in many senses is a good choice of input signal for LTI approximations of nonlinear systems. Furthermore, the result in Lemma 6.1 that some random multisines are separable has direct practical implications and can be viewed as a theoretical motivation for an input signal that is already commonly used. The third contribution of practical interest is the result in Corollary 7.1 about generalized Wiener-Hammerstein systems with Gaussian inputs. This result implies that the linear parts of such a system will be factors in the OE-LTI-SOE of it. Finally, the mixed parametric and nonparametric identification method in Chapter 11 might be useful in some applications.

Some other results can also be viewed as main contributions. For example, the result about higher order separability in Theorem 6.3 is a generalization of a classic theoretical result. With the behavior of LTI-SOEs shown in Examples 1.1-1.4 in mind, also the result

in Theorem 8.1 about uniform convergence of the linear approximations when the size of the nonlinearities tends to zero can be viewed as a main contribution.

Most of the material of this thesis has previously been published. With exception for the discussion about LTI-SOEs for periodic inputs and the separability of random multisines, the results in Chapters 4 to 9 have been published previously in

M. Enqvist. Some results on linear models of nonlinear systems. Licentiate thesis no. 1046. Department of Electrical Engineering, Linköpings universitet, Linköping, Sweden, 2003.

The results about higher order separability and LTI-SOEs for Gaussian inputs in Section 6.3 and Chapter 7 can also be found in

M. Enqvist and L. Ljung. Linear approximations of nonlinear FIR systems for separable input processes. *Automatica*, 41(3):459–473, 2005.

Early versions of the results in Chapter 8 about approximations of almost linear systems can be found in

M. Enqvist and L. Ljung. Estimating nonlinear systems in a neighborhood of LTI-approximants. In *Proceedings of the 41st IEEE Conference on Decision and Control*, pages 1005–1010, Las Vegas, Nevada, December 2002.

The material in Chapter 7 about Gaussian inputs has previously been published also in

M. Enqvist and L. Ljung. Linear models of nonlinear FIR systems with Gaussian inputs. In *Proceedings of the 13th IFAC Symposium on System Identification*, pages 1910–1915, Rotterdam, The Netherlands, August 2003.

Some of the examples and results in Chapter 8 can also be found in

M. Enqvist and L. Ljung. LTI approximations of slightly nonlinear systems: Some intriguing examples. In *Proceedings of the 6th IFAC Symposium on Nonlinear Control Systems*, pages 639–644, Stuttgart, Germany, September 2004.

The approach to robust control in Chapter 10 has also been studied in

S. T. Glad, A. Helmersson, M. Enqvist, and L. Ljung. Controllers for amplitude limited model error models. In *Proceedings of the 16th IFAC World Congress*, Prague, Czech Republic, July 2005.

Some of the results about minimum phase filtered white noise inputs in Chapter 5 are described in

M. Enqvist. Benefits of the input minimum phase property for linearization of nonlinear systems. In *Proceedings of the International Symposium on Nonlinear Theory and its Applications*, pages 618–621, Bruges, Belgium, October 2005.

Most of the results about the mixed parametric and nonparametric method for system identification in Chapter 11 are published in

J. Roll, M. Enqvist, and L. Ljung. Consistent nonparametric estimation of NARX systems using convex optimization. In *Proceedings of the 44th IEEE Conference on Decision and Control and the European Control Conference*, Seville, Spain, December 2005a. (To appear).

The results about separability of random multisines in Section 6.2 can be found also in

M. Enqvist. Identification of Hammerstein systems using separable random multisines. Submitted to the 14th IFAC Symposium on System Identification, Newcastle, Australia, March 2006.

2

# **Preliminaries**

In this chapter, some background material about linear and nonlinear systems will be presented and the notation that will be used throughout this thesis will be introduced. Furthermore, a brief description of the basic ideas of system identification based on predictionerror methods and an introduction to separable processes will be given.

# 2.1 Linear Systems and Stochastic Processes

Linear time-invariant (LTI) dynamic systems and models are the foundation of control theory and system identification and are described in many textbooks (see, for example, Kailath, 1980; Rugh, 1996). Any discrete-time LTI system with input u(t) and output y(t) can be written as a convolution

$$y(t) = \sum_{k=-\infty}^{\infty} g(k)u(t-k).$$

The sequence  $(g(k))_{k=-\infty}^{\infty}$  is called the *impulse response* of the system. An LTI system can also be represented by a *transfer function* G(z), which is obtained by taking the z-transform of the impulse response, i.e.,

$$G(z) = \sum_{k=-\infty}^{\infty} g(k) z^{-k}.$$

Similarly, the function G(q), where q is the shift operator qu(t) = u(t+1), will be called the *transfer operator* of the system. A third way to represent a discrete-time LTI system is to write it as a *state-space description* or a *state equation* 

$$\begin{aligned} x(t+1) &= Ax(t) + Bu(t), \\ y(t) &= Cx(t) + Du(t), \end{aligned}$$

where x(t) is a state vector and where A, B, C and D are matrices.

Although a transfer function sometimes can be written more compactly as a rational function of z, it should always be thought of as a certain series expansion in order to avoid any ambiguities. These ambiguities can occur due to the fact that a rational function corresponds to different series expansions in different regions of convergence. However, the series expansion is unique if the region of convergence is specified (Brown and Churchill, 1996, Sec. 50). Sometimes, this specification will be done using the following terminology.

**Definition 2.1.** A sequence  $(m(k))_{k=-\infty}^{\infty}$  is causal if m(k) = 0 for all k < 0 and strictly causal if m(k) = 0 for all  $k \le 0$ . The sequence is anticausal if m(k) = 0 for all k > 0 and strictly anticausal if m(k) = 0 for all  $k \ge 0$ .

The notion of causality can be used also for LTI systems.

**Definition 2.2.** An LTI system is *(strictly) causal* if its impulse response is *(strictly) causal*. Similarly, an LTI system is *(strictly) anticausal* if its impulse response is *(strictly) anticausal*.

In some cases, we will need to extract the causal part of a noncausal system. This will be done using the notation

$$[G(z)]_{\text{causal}} = \left[\sum_{k=-\infty}^{\infty} g(k) z^{-k}\right]_{\text{causal}} = \sum_{k=0}^{\infty} g(k) z^{-k}.$$

Causality of an LTI system implies that the system output only depends on past and present values of the input signal. Hence, all real-life systems are causal. Another important property of LTI systems is stability. In this thesis, we will only use the type of stability called *bounded input bounded output stability*, which is defined as follows.

**Definition 2.3.** An LTI system with impulse response g(k) is stable if

$$\sum_{k=-\infty}^{\infty} |g(k)| < +\infty.$$

If a transfer function is said to be stable, it should always be viewed as coming from the series expansion, causal or noncausal, whose region of convergence contains the unit circle. On the other hand if a transfer function is said to be causal it should be viewed as coming from a, possibly unstable, causal series expansion.

Furthermore, an LTI system G(z) is said to be *static* if only g(0) is nonzero and nonstatic if there exists a  $k \in \mathbb{Z} \setminus \{0\}$  such that  $g(k) \neq 0$ . If g(k) is nonzero only for a finite number of k:s, the system is said to be a *finite impulse response* (FIR) system. An LTI system is said to be *monic* if g(0) = 1. In some cases, we will use the notation

$$G^{-1}(z) = \frac{1}{G(z)} = \sum_{k=0}^{\infty} \tilde{g}(k) z^{-k}$$

for the inverse system of a causal LTI system. As indicated above,  $G^{-1}(z)$  should always be viewed as a causal series expansion. An important notion for control theory, and also for the discussion later in this thesis, is the concept of *minimum phase* systems. **Definition 2.4.** An LTI system is minimum phase if both G(z) and  $G^{-1}(z) = 1/G(z)$  are stable and causal transfer functions.

The definitions that have been introduced so far have concerned LTI systems but do of course hold for LTI models and filters as well. The word filter will here be used as an alternative name for an LTI system whose main purpose is to change a signal in some way. The signals that will be discussed in this thesis are discrete-time stationary stochastic processes (see, for example, Gardner, 1986; Jazwinski, 1970).

Formally, a discrete-time stochastic process  $(u(t))_{t=-\infty}^{\infty}$  is an indexed sequence of random variables where the parameter t corresponds to time. The processes that will be studied in this thesis will be real and *stationary*. Stationarity of a process means that the simultaneous probability density function of any set of variables  $\{u(t + \tau), \tau \in D \subset \mathbb{Z}\}$ is independent of t. Furthermore, all processes will have zero mean, i.e., E(u(t)) = 0for all  $t \in \mathbb{Z}$ , and well-defined *covariance functions*  $R_u(\tau)$ . The covariance function of a process with zero mean is defined as

$$R_u(\tau) = \mathcal{E}(u(t)u(t-\tau)).$$

Furthermore, it will be assumed that the covariance function has a well-defined z-transform  $\Phi_u(z)$  whose region of convergence contains the unit circle. The function  $\Phi_u(z)$  can be written

$$\Phi_u(z) = \sum_{\tau = -\infty}^{\infty} R_u(\tau) z^{-\tau}$$

and it will, using the terminology in Kailath et al. (2000), be called the *z*-spectrum of the process. Properties like stability and causality that hold for LTI systems can be used also about z-spectra. Note that  $\Phi_u(z^{-1}) = \Phi_u(z)$  since  $R_u(-\tau) = R_u(\tau)$ . The real-valued function  $\Phi_u(e^{i\omega})$  of  $\omega \in [-\pi, \pi]$  that is obtained when  $z = e^{i\omega}$  will be called the spectral density function of the process.

If two processes  $(u(t))_{t=-\infty}^{\infty}$  and  $(y(t))_{t=-\infty}^{\infty}$  are considered, it will be assumed that they are jointly stationary and that the *cross-covariance function*  $R_{yu}(\tau)$  between these processes exists. The cross-covariance function is defined as

$$R_{yu}(\tau) = \mathcal{E}(y(t)u(t-\tau))$$

Furthermore, it will be assumed that also this function has a z-transform  $\Phi_{yu}(z)$  whose region of convergence contains the unit circle. The function  $\Phi_{uu}(z)$  can be written

$$\Phi_{yu}(z) = \sum_{\tau = -\infty}^{\infty} R_{yu}(\tau) z^{-\tau}$$

and will be called the *z*-cross-spectrum. Note that  $\Phi_{yu}(z^{-1}) = \Phi_{uy}(z)$  and that all *z*-spectra and *z*-cross-spectra should always be interpreted as the series expansion whose region of convergence contains the unit circle.

A very important class of processes is *white noise processes*, which have the property that all  $u(t), t \in \mathbb{Z}$ , are independent. Hence, for white processes only  $R_u(0)$  is nonzero. Using white processes as inputs to LTI filters, it is possible to construct processes with arbitrary z-spectra. This follows from the next lemma about LTI filtering of stationary stochastic processes. This lemma has been taken from Kailath et al. (2000, p. 195).



Figure 2.1: The general LTI model.

#### Lemma 2.1 (Filtering of Stationary Processes)

Let  $(y(t))_{t=-\infty}^{\infty}$  be the stationary process that is obtained by passing a stationary process  $(u(t))_{t=-\infty}^{\infty}$  with zero mean through a stable LTI system with transfer function H(z). Then the relations

$$\Phi_y(z) = H(z)\Phi_u(z)H(z^{-1}),$$
  
$$\Phi_{uu}(z) = H(z)\Phi_u(z)$$

hold. Furthermore, if  $(x(t))_{t=-\infty}^{\infty}$  is jointly stationary with  $(y(t))_{t=-\infty}^{\infty}$  and  $(u(t))_{t=-\infty}^{\infty}$  as just defined, then

$$\Phi_{xy}(z) = \Phi_{xu}(z)H(z^{-1})$$

Proof: See Kailath et al. (2000, pp. 195-197).

LTI models and stochastic processes will in this thesis be used to model arbitrary systems. Usually, it will be assumed that these systems contain some noise and hence we need models that include some kind of noise description. One model with this property is the following general LTI model of a system with input u(t) and output y(t),

$$y(t) = G(q)u(t) + H(q)e(t),$$
 (2.1)

where H(q) is a monic transfer operator that describes how the output depends on the white noise e(t). The structure of the model (2.1) is illustrated in Figure 2.1.

The LTI model (2.1) can be used to define the optimal predictor  $\hat{y}(t)$  of y(t) given past output values  $(y(t-k))_{k=1}^{\infty}$  and past and present input values  $(u(t-k))_{k=0}^{\infty}$  (see, for example, Ljung, 1999, Chap. 3). This predictor can be written as

$$\hat{y}(t) = H^{-1}(q)G(q)u(t) + (1 - H^{-1}(q))y(t).$$
 (2.2)

The predictor (2.2) is optimal in the sense that if (2.1) is an accurate description of the true system, it minimizes the *mean-square error*  $E((y(t) - \hat{y}(t))^2)$  and is equal to the conditional expectation of y(t) given past output and past and present input values (Ljung, 1999, Chap. 3). Predictors of this kind are used in the prediction-error method, which will be described in Section 2.3. First, however, we will give a brief overview of some types of nonlinear systems that will be discussed later in this thesis.

## 2.2 Nonlinear Systems

In some of the results that will be presented in this thesis, there will be no explicit assumptions on the true nonlinear system that is modeled. Hence, the system can often be viewed as a black box that for a given stationary input signal  $(u(t))_{t=-\infty}^{\infty}$  produces the output  $(y(t))_{t=-\infty}^{\infty}$ . However, in some other results, we will assume that the system belongs to certain classes of nonlinear systems and these classes will be defined here.

We will only consider nonlinear systems in discrete time in this thesis. Similarly to the LTI case, a nonlinear system will be said to be static if its output y(t) can be written as a function of u(t) only, i.e., if y(t) = f(u(t)), and the system is said to be nonstatic if y(t) depends on any other u(t-k),  $k \in \mathbb{Z} \setminus \{0\}$ . A class of systems that will be discussed later in this thesis is *nonlinear finite impulse response* (NFIR) systems. An NFIR system can for some  $M \in \mathbb{N}$  be written as

$$y(t) = f((u(t-k))_{k=0}^{M}).$$

Here, the compact notation  $f((u(t-k))_{k=0}^{M})$  simply means

$$f(u(t), u(t-1), \ldots, u(t-M)),$$

i.e., a function of a finite number of input components. An NFIR system is a special case of a *nonlinear autoregressive system with external input* (NARX system) (Sjöberg et al., 1995). Such a system can be written as

$$y(t) = f(\varphi(t)) + e(t),$$

where the vector  $\varphi(t)$  has signal components of u and y as elements and where e(t) is a white noise process. An NFIR system is also a special case of a *nonlinear output error* (NOE) system

$$y(t) = f((u(t-k))_{k=0}^{\infty}) + e(t),$$

where e(t) is white noise.

Two other system classes that will be discussed later are *Wiener* and *Hammerstein* systems. A Wiener system consists of an LTI model followed by a static nonlinearity, i.e.,

$$y(t) = f(v(t)),$$
  
$$v(t) = G(q)u(t),$$

while a Hammerstein system has these linear and nonlinear subsystems in the opposite order, i.e.,

$$y(t) = G(q)v(t),$$
  
$$v(t) = f(u(t)).$$

Actually, Wiener and Hammerstein systems can be viewed as special cases of a more general system class known as *Wiener-Hammerstein* systems. Such a system consists of a Wiener system followed by an LTI system.

Just like LTI systems, many nonlinear systems can be written also in state-space form

$$\begin{aligned} x(t+1) &= f(x(t), u(t)), \\ y(t) &= h(x(t), u(t)), \end{aligned}$$

where x(t) is a state vector.

A detailed description and characterization of many other types of nonlinear systems and models can be found in Pearson (1999) and Sjöberg et al. (1995). In the next section, we will describe some of the basic ideas in system identification.

## 2.3 System Identification

As was mentioned in the introduction to this thesis, *system identification* can be viewed as a synonym for mathematical modeling of dynamic systems using measurements of the input and output signals. Various identification methods can be found in the literature, but here we will only discuss one family of methods, namely *prediction-error methods*. We will discuss the general idea behind this method, some of its properties and also a special version of it designed for a class of input signals called *random multisines*.

## 2.3.1 Prediction-Error Methods

Prediction-error methods are based on the observation that predictors like (2.2) can be used to compare how well different LTI models can predict the output y(t). The main idea is to use some kind of measure of the distance between the predicted output and the true output and to minimize this distance by adjusting some parameters in the model. Typically, a prediction-error method works with a finite data set  $Z^N = (u(t), y(t))_{t=1}^N$  that contains simultaneous measurements of the input and output signals and a parameterized version of the general LTI model (2.1). This parameterized model can be written as

$$y(t,\theta) = G(q,\theta)u(t) + H(q,\theta)e(t), \qquad (2.3)$$

where  $\theta$  is a *d*-dimensional vector of parameters. For example,  $\theta$  can be the coefficients of the numerator and denominator polynomials of *G* and *H*, provided that these transfer functions are chosen as rational functions.

Different model structures can be obtained by imposing some restrictions on the rational functions G and H. For example, the *autoregressive with external input* (ARX) model structure is obtained by letting

$$G(q,\theta) = \frac{B(q,\theta)}{A(q,\theta)},$$
$$H(q,\theta) = \frac{1}{A(q,\theta)},$$

where A and B are polynomials. Similarly, the *output error* (OE) model structure is acquired if  $H(q, \theta) = 1$ . A family of LTI model structures is described in Ljung (1999, pp. 81-88).

If the model (2.3) would be a perfect description of the true system for some white noise process e(t), the mean-square error optimal predictor  $\hat{y}(t, \theta)$  of y(t) would be

$$\hat{y}(t,\theta) = H^{-1}(q,\theta)G(q,\theta)u(t) + (1 - H^{-1}(q,\theta))y(t).$$
(2.4)

When a model structure has been selected in the prediction-error method, the corresponding predictor (2.4) is used to compute  $\theta$ -dependent predictions  $\hat{y}(t,\theta)$  based on the data in  $Z^N$ . A parameter estimate  $\hat{\theta}_N$  can then be computed by minimizing a criterion  $V_N(\theta, Z^N)$ . For example, this criterion can be chosen to be quadratic such that

$$\hat{\theta}_N = \operatorname*{arg\,min}_{\theta \in D_M} V_N(\theta, Z^N) = \operatorname*{arg\,min}_{\theta \in D_M} \frac{1}{N} \sum_{t=1}^N (y(t) - \hat{y}(t, \theta))^2. \tag{2.5}$$

Here,  $\theta$  is restricted to some pre-specified set  $D_M \subset \mathbb{R}^d$ . Usually,  $D_M$  is the set of parameters that make the predictor (2.4) stable. In general, the minimization of  $V_N(\theta, Z^N)$  has to be performed using some kind of numerical method. A common choice is to use a *Gauss-Newton* or a *damped Gauss-Newton* method. These methods use the gradient and an approximation of the Hessian of  $V_N(\theta, Z^N)$  and have good convergence properties, especially in the vicinity of the optimum (see Ljung, 1999).

Detailed studies of the properties of the prediction-error estimate when the number of measurements tends to infinity have been made (Ljung, 1978, 1999). In Ljung (1978) it is shown that under rather weak conditions on the true system and on the input and output signals, the following convergence result holds with probability one.

$$\hat{\theta}_N \to \theta^* = \underset{\theta \in D_M}{\operatorname{arg\,min}} \operatorname{E}((y(t) - \hat{y}(t, \theta))^2), \quad \text{w.p.1 as } N \to \infty$$
(2.6)

With some abuse of notation, y(t) and  $\hat{y}(t,\theta)$  here denote the stochastic signals while they previously in this section have denoted realizations of these signals. An important necessary condition on the input and output signals for (2.6) to hold concerns the dependency between signal components over time. Intuitively, this condition requires that the remote past of the process should be forgotten at an exponential rate. This condition is satisfied for many random input signals of practical interest, e.g., most filtered white noise signals and random binary signals. For such inputs, the properties of a model estimated using (2.5) can often be understood by studying the model that minimizes the mean-square error in (2.6). However, there is at least one important class of input signals for which the result (2.6) is not applicable, namely random multisines. Usually, a modified version of the prediction-error method is used for these input signals. This will be discussed in the next section.

The convergence result (2.6) holds also for many nonlinear systems. Since (2.6) shows that the prediction-error estimate with probability one will converge to the mean-square error optimal estimate  $\theta^*$ , it is interesting to investigate what can be said about the LTI models that are defined by  $\theta^*$  when the true system is nonlinear. This is the main objective of this thesis.

A convergence result that is similar to (2.6) can be shown under less restrictive assumptions on the input signal if the system is linear (Ljung, 1999, Chap. 8). In this result, the expectation in (2.6) is replaced with both an average over time and an expectation, and the convergence of the parameter estimates is obtained also for *quasi-stationary* signals (Ljung, 1999, Chap. 2), e.g., pseudo-random binary signals and deterministic multisine signals. However, this result cannot be applied to nonlinear systems. Furthermore, it is not obvious that all quasi-stationary signals are suitable for estimation of LTI models of nonlinear systems since some averaging due to the randomness of the input signal might be beneficial in this type of identification problem. Intuitively, the reason why a random input signal might be useful is that it often will give a linear model that approximates the average behavior of the system rather than just the system behavior for one fixed input.

However, also when a nonlinear system is approximated by an LTI model using the prediction-error method and a realization of a stochastic process as input, there is a risk that the obtained model will be adjusted too much to the particular realization of the input used in the identification experiment. For example, consider an identification experiment where a realization of a stochastic process with zero mean is used to model the average behavior of a nonlinear system in an interval around zero. If the realization of the input is short, there is often a significant probability that, for example, all signal components will have equal signs. In this case, an estimated model will usually not be able to describe the desired system behavior accurately. However, if the dependency between two input components u(t) and  $u(t - \tau)$  decreases as  $|\tau|$  increases, the probability to get a realization where all components have equal signs tends to zero when the number of signal components in the realization goes to infinity. Hence, an estimated model will typically describe the average system behavior better if a large data set is used in the identification procedure. For many systems, such an input signal will guarantee also that (2.6) holds.

The method described here is not the only prediction-error method, but rather a commonly used member of a family of methods. The main differences between these methods are due to different choices of criterion in (2.5) and to the fact that a prefilter is used in some methods. It should also be noted that prediction-error methods can be used for other model structures than the ones based on (2.1). For example, both linear state-space models and general nonlinear black-box models can be used.

## 2.3.2 A Prediction-Error Method for Random Multisines

The use of periodic input signals in identification experiments is common in many applications and can be motivated in several ways (Pintelon and Schoukens, 2001; Ljung, 1999). For example, if the modeling is done using a frequency domain criterion, a periodic signal will remove the undesirable leakage effects that are usually present. Furthermore, it is easy to calculate good estimates of the noise level with such an input.

A discrete-time signal u(t) is periodic if there is a positive integer P such that

$$u(t+P) = u(t), \quad \forall t \in \mathbb{Z}.$$

Consider a *P*-periodic input, with  $P \in \mathbb{Z}_+$ , to a system with the output

$$y(t) = y_{nf}(t) + w(t),$$

where  $y_{nf}(t)$  is the noise-free output and w(t) is measurement noise. Assume that measurements from M periods have been collected and that all transient effects have disappeared such that  $y_{nf}(t)$  is P-periodic too. In this case, an average  $\bar{y}(t)$  of the output signal
over the periods can be calculated as

$$\bar{y}(t) = \frac{1}{M} \sum_{k=0}^{M-1} y(t+kP), \quad 1 \le t \le P.$$

In this way, a shorter signal with a higher signal-to-noise ratio is obtained and an estimate  $\hat{\lambda}_w$  of the variance of the measurement noise can be calculated as

$$\hat{\lambda}_w = \frac{1}{(M-1)P} \sum_{k=0}^{M-1} \sum_{t=1}^{P} \left( y(t+kP) - \bar{y}(t) \right)^2.$$

A disadvantage with a P-periodic signal is that it can only be persistently exciting of, at most, order P (Ljung, 1999, Chap. 13). Hence, models with arbitrarily many parameters cannot be uniquely determined if a periodic input has been used. Further properties of periodic excitation signals can be found in Pintelon and Schoukens (2001) and Ljung (1999).

A particular class of periodic signals that has turned out to be useful for identification of linear and nonlinear systems is *random multisines* (Pintelon and Schoukens, 2001).

**Definition 2.5.** A random multisine signal is a stationary stochastic process u(t) that can be written

$$u(t) = \sum_{k=1}^{Q} A_k \cos(\omega_k t + \psi_k),$$
(2.7)

where both  $A_k$  and  $\psi_k$  can be random variables and where all  $\omega_k$  are constants that satisfy  $|\omega_k| \leq \pi$ .

Here, the phases  $\psi_k$  will usually be independent random variables with uniform distribution on the interval  $[0, 2\pi)$  and the amplitudes  $A_k$  will usually be constants. Furthermore, we will only consider periodic random multisines such that the period P is an integer, i.e., such that all  $\omega_k$  can be written  $\omega_k = \pi p_k$  for some  $p_k \in \{x \in \mathbb{Q} \mid |x| \le 1\}$ .

A linear system can be identified from one realization of a random multisine, but for a nonlinear system it is important to use several realizations in order to get a model that is not too adjusted to the signal shape of one realization only. Since a random multisine under certain assumptions on  $\omega_k$  is periodic, the dependency between the signal components does not decrease over time. Hence, the convergence result (2.6) will not hold in general. This implies that for a nonlinear system, parameter estimates that give models that are good approximations of the mean-square error optimal model cannot be obtained by collecting many measurement from one single identification experiment. Instead, several experiments have to be performed.

With data sets from  $N_E$  experiments where different realizations of the input signal have been used and N = MP measurements in each data set, with  $M \in \mathbb{Z}_+$ , a model can be estimated by minimizing the cost function

$$V_{N_E,N}(\theta, Z_{N_E}^N) = \frac{1}{N_E} \sum_{s=1}^{N_E} \frac{1}{N} \sum_{t=0}^{N-1} (y_s(t) - G(q, \theta) u_s(t))^2$$
(2.8)

with respect to the parameters  $\theta$ . Here,  $u_s(t)$  and  $y_s(t)$  are the input and output signals from experiment s, respectively, and  $Z_{N_E}^N$  is the combined data set with measurements from all experiments. Intuitively, this cost function can be viewed as an approximation of the mean-square error  $E((y(t) - G(q, \theta)u(t))^2)$ , just like the cost function  $V_N$  in (2.5) can be viewed as an approximation of the mean-square error. However,  $V_{N_E,N}$  will typically approach the mean-square error when the number of experiments  $N_E$  tends to infinity while  $V_N$  approaches it when the number of measurements N in one experiment tends to infinity according to (2.6).

With a periodic input, it is very natural to consider the modeling problem in the frequency domain. Applying the *Discrete Fourier Transform* (DFT) to the input and output signals gives the transforms

$$U_{s,N}(n) = \sum_{t=0}^{N-1} u_s(t) e^{-i2\pi nt/N},$$
(2.9a)

$$Y_{s,N}(n) = \sum_{t=0}^{N-1} y_s(t) e^{-i2\pi nt/N}.$$
(2.9b)

Let  $\hat{y}_s(t,\theta)$  denote the output from the stable model  $G(q,\theta)$  for the input  $u_s(t)$  and assume that the input has been applied at  $t = -\infty$  such that all transients have disappeared at  $t \ge 0$ , i.e., that  $\hat{y}_s(t,\theta)$  is *P*-periodic in the interval  $0 \le t \le N-1$ . Furthermore, let  $\hat{Y}_{s,N}(n,\theta)$  denote the DFT of  $\hat{y}_s(t,\theta)$ , i.e.,

$$\hat{Y}_{s,N}(n,\theta) = \sum_{t=0}^{N-1} \hat{y}_s(t,\theta) e^{-i2\pi nt/N}.$$

The frequency response of the stable model  $G(z, \theta)$  is obtained for  $z = e^{i\omega}$ . In particular, since  $v(k) \triangleq e^{-i2\pi nk/N}$  is an N-periodic signal, it follows that

$$G(e^{i2\pi n/N}, \theta) = \sum_{k=0}^{\infty} g(k, \theta) e^{-i2\pi nk/N}$$
$$= \sum_{t=0}^{N-1} \underbrace{\left(\sum_{l=0}^{\infty} g(t+lN, \theta)\right)}_{\triangleq \tilde{g}_N(t, \theta)} e^{-i2\pi nt/N} \triangleq \tilde{G}_N(n, \theta).$$
(2.10)

Furthermore, since  $u_s(t)$  is a *P*-periodic signal and N = MP, with  $M \in \mathbb{Z}_+$ ,  $u_s(t)$  is also *N*-periodic. Hence,

$$\hat{y}_s(t,\theta) = G(q,\theta)u_s(t) = \sum_{k=0}^{N-1} \tilde{g}(k,\theta)u_s(t-k)$$

and this implies that

$$\hat{Y}_{s,N}(n,\theta) = \tilde{G}_N(n,\theta) U_{s,N}(n) = G(e^{i2\pi n/N},\theta) U_{s,N}(n),$$
(2.11)

where we have used (2.10) in the last equality.

Using Parseval's formula, the cost function can be rewritten as

$$\begin{aligned} V_{N_E,N}(\theta, Z_{N_E}^N) &= \frac{1}{N_E} \sum_{s=1}^{N_E} \frac{1}{N} \sum_{t=0}^{N-1} (y_s(t) - \underbrace{G(q, \theta) u_s(t)}_{=\hat{y}_s(t, \theta)})^2 \\ &= \frac{1}{N_E} \sum_{s=1}^{N_E} \frac{1}{N^2} \sum_{n=0}^{N-1} \left| Y_{s,N}(n) - \hat{Y}_{s,N}(n, \theta) \right|^2 \\ &= \frac{1}{N_E} \sum_{s=1}^{N_E} \frac{1}{N^2} \sum_{n=0}^{N-1} \left| Y_{s,N}(n) - G(e^{i2\pi n/N}, \theta) U_{s,N}(n) \right|^2. \end{aligned}$$

From this expression, it is obvious that two linear models will give the same value of the cost function if their frequency responses are equal at the frequencies where  $U_{s,N}(n)$  is nonzero. Assume that the input is a random multisine such that  $U_{s,N}(n)$  is nonzero at the frequencies where  $n \in \Omega \subset \{0, 1, \ldots, N-1\}$  and zero otherwise and consider a nonparametric frequency response model  $G_{np}(n) = G(e^{i2\pi n/N}, \theta)$ . In this case, minimizing  $V_{N_{E,N}}$  is equivalent to solving a least squares problem for each  $n \in \Omega$ . The resulting nonparametric estimate can be written

$$\hat{G}_{np}(n) = \frac{\sum_{s=1}^{N_E} Y_{s,N}(n) \overline{U_{s,N}(n)}}{\sum_{s=1}^{N_E} |U_{s,N}(n)|^2}, \quad n \in \Omega.$$
(2.12)

In particular, if  $|U_{s,N}(n)|$  are equal for all s, (2.12) can be simplified to

$$\hat{G}_{np}(n) = \frac{1}{N_E} \sum_{s=1}^{N_E} \frac{Y_{s,N}(n)}{U_{s,N}(n)}, \quad n \in \Omega.$$
(2.13)

For example, this expression can be used when the input is a random multisine where all amplitudes  $A_k$  are constants and all frequencies  $\omega_k$  are separate. More results about random multisines and frequency domain identification can be found in, for example, Pintelon and Schoukens (2001).

## 2.4 Separable Processes

Some of the results in this thesis concern processes that are separable in Nuttall's sense (Nuttall, 1958a), i.e., processes which satisfy the condition described in the following definition.

**Definition 2.6 (Separability).** A stationary stochastic process u(t) with E(u(t)) = 0 is *separable* (in Nuttall's sense) if

$$E(u(t - \tau)|u(t)) = a(\tau)u(t).$$
(2.14)

for some function  $a(\tau)$ .

In this section, some of the main results for separable processes will be presented. These results can all be found in Nuttall (1958a,b) but they are here rewritten with the notation used in this thesis. Note that the technical report Nuttall (1958a) is an almost identical copy of the thesis Nuttall (1958b). Here, the first of these works will be used as the main reference. Furthermore, note that some of the proofs in this section are slightly different from the corresponding ones in Nuttall (1958a).

It is easy to show that the function  $a(\tau)$  in (2.14) can be expressed using the covariance function of u(t).

#### Lemma 2.2

Consider a separable stationary stochastic process u(t) with E(u(t)) = 0. The function  $a(\tau)$  from (2.14) can then be written

$$a(\tau) = \frac{R_u(\tau)}{R_u(0)}.$$
 (2.15)

**Proof:** The result follows immediately from the fact that

$$R_{u}(\tau) = \mathcal{E}(u(t)u(t-\tau)) = \mathcal{E}(u(t)\mathcal{E}(u(t-\tau)|u(t))) = a(\tau)\mathcal{E}(u(t)^{2}) = a(\tau)R_{u}(0)$$

if u(t) is separable. Here, we have used the facts that

$$E(Y) = E(E(Y|X)), \qquad (2.16a)$$

$$E(g(X)Y|X) = g(X)E(Y|X)$$
(2.16b)

for two random variables X and Y (see, for example, Gut, 1995, Chap. 2).  $\Box$ 

Separability can be expressed also using characteristic functions. Hence, the following definition is useful.

**Definition 2.7.** Consider a stationary stochastic process u(t) with E(u(t)) = 0 and with first and second order characteristic functions

$$f_{u,1}(\xi_1) = \mathcal{E}(e^{i\xi_1 u(t)}), \tag{2.17a}$$

$$f_{u,2}(\xi_1,\xi_2,\tau) = \mathcal{E}(e^{i\xi_1 u(t) + i\xi_2 u(t-\tau)}),$$
(2.17b)

respectively. Then the *G*-function  $G_u(\xi_1, \tau)$  of this process is defined as

$$G_u(\xi_1, \tau) = \left. \frac{\partial f_{u,2}(\xi_1, \xi_2, \tau)}{\partial \xi_2} \right|_{\xi_2 = 0} = \mathcal{E}(iu(t-\tau)e^{i\xi_1 u(t)}).$$
(2.18)

In Nuttall (1958a), a number of separable signals are listed, e.g., Gaussian processes, random binary processes and several types of modulated processes. For example, a single sinusoid with random phase is separable according to the following lemma.

Lemma 2.3

A random sine process

$$u(t) = A\cos(\omega t + \psi), \qquad (2.19)$$

where  $\psi$  is a random variable with uniform distribution on the interval  $[0, 2\pi)$  and where A and  $\omega$  are constants, is a separable process. Furthermore, this process has the properties

$$R_u(\tau) = \frac{A^2}{2}\cos(\omega\tau), \qquad (2.20a)$$

$$f_{u,1}(\xi_1) = J_0(A\xi_1), \tag{2.20b}$$

where  $J_0$  is the zeroth order Bessel function.

Proof: Using basic properties of trigonometric functions, we have

$$u(t - \tau) = A\cos(\omega t - \omega\tau + \psi) = A\cos(\omega t + \psi)\cos(\omega\tau) + A\sin(\omega t + \psi)\sin(\omega\tau)$$
$$= u(t)\cos(\omega\tau) + A\sin(\omega t + \psi)\sin(\omega\tau).$$

Since  $A\sin(\omega t + \psi)$  equals  $\sqrt{A^2 - u(t)^2}$  or  $-\sqrt{A^2 - u(t)^2}$  with equal probabilities if u(t) is given,

$$\mathcal{E}(A\sin(\omega t + \psi)|u(t)) = 0.$$

Hence, it follows that

$$E(u(t-\tau)|u(t)) = \cos(\omega\tau)u(t)$$

i.e., u(t) is separable. Furthermore, Lemma 2.2 implies that the covariance function of u(t) is

$$R_u(\tau) = \cos(\omega\tau) \mathcal{E}(u(t)^2) = A^2 \cos(\omega\tau) \frac{1}{2\pi} \int_0^{2\pi} \cos(\omega t + \psi)^2 d\psi = \frac{A^2}{2} \cos(\omega\tau)$$

and the characteristic function is

$$f_{u,1}(\xi_1) = \mathcal{E}(e^{i\xi_1 u(t)}) = \frac{1}{2\pi} \int_0^{2\pi} e^{iA\xi_1 \cos(\omega t + \psi)} d\psi =$$
  
=  $\left\{ \tilde{\psi} = \omega t + \psi + \frac{\pi}{2} \right\} = \frac{1}{2\pi} \int_{\omega t + \pi/2}^{\omega t + 5\pi/2} e^{iA\xi_1 \cos(\tilde{\psi} - \pi/2)} d\tilde{\psi} =$   
=  $\frac{1}{2\pi} \int_0^{2\pi} e^{iA\xi_1 \sin(\tilde{\psi})} d\tilde{\psi} = J_0(A\xi_1).$ 

In the previous lemma, separability of a single sinusoid with random phase was proved directly using Definition 2.6. However, in many cases it is more convenient to show separability of a signal using characteristic functions. This is possible according to the following theorem.

#### Theorem 2.1

Consider a stationary stochastic process u(t) with E(u(t)) = 0. This process is separable if and only if

$$G_u(\xi_1, \tau) = a(\tau) f'_{u,1}(\xi_1), \qquad (2.21)$$

where  $a(\tau) = R_u(\tau)/R_u(0)$  and where  $f_{u,1}$  and  $G_u$  are defined in (2.17a) and (2.18), respectively.

**Proof:** IF: Assume that (2.21) holds. Then, using the definition of  $G_u$  in (2.18), it follows that

$$\mathbb{E}\Big(ie^{i\xi_1 u(t)}\Big(\mathbb{E}\big(u(t-\tau)|u(t)\big) - a(\tau)u(t)\Big)\Big) = G_u(\xi_1,\tau) - a(\tau)f'_{u,1}(\xi_1) = 0.$$

From the uniqueness of Fourier transforms it thus follows that (2.14) holds. Hence, u(t) is separable if (2.21) holds.

ONLY IF: Assume that u(t) is separable, i.e., that (2.14) holds. This implies that

$$G_u(\xi_1, \tau) = \mathcal{E}(iu(t-\tau)e^{i\xi_1u(t)}) = \mathcal{E}(ie^{i\xi_1u(t)}\mathcal{E}(u(t-\tau)|u(t)))$$
  
=  $a(\tau)\mathcal{E}(iu(t)e^{i\xi_1u(t)}) = a(\tau)f'_{u,1}(\xi_1),$ 

where we have used (2.14) in the second equality. Hence, (2.21) holds if u(t) is separable.

In the next theorem from Nuttall (1958a), it is shown that the sum of Q independent separable processes is separable if the characteristic functions satisfy a certain condition.

#### Theorem 2.2

Consider Q independent and separable stationary stochastic processes  $u_k(t)$  with

$$\mathbf{E}(u_k(t)) = 0$$

for  $k = 1, \ldots, Q$  and let

$$u(t) = \sum_{k=1}^{Q} u_k(t).$$
 (2.22)

Assume that the characteristic functions satisfy

$$f_{u_k,1}(\xi_1)^{1/\sigma_k^2} = f_{u_l,1}(\xi_1)^{1/\sigma_l^2}, \quad \forall k, l \in \{1, 2, \dots, Q\},$$
(2.23)

where  $\sigma_m^2 = R_{u_m}(0)$ . Then u(t) is separable.

**Proof:** Since the signals  $u_k(t)$  are independent, we have

$$f_{u,1}(\xi_1) = \prod_{k=1}^{Q} f_{u_k,1}(\xi_1) = f_{u_1,1}(\xi_1) \sum_{k=1}^{Q} \sigma_k^2 / \sigma_1^2, \qquad (2.24)$$

$$f_{u,2}(\xi_1,\xi_2,\tau) = \prod_{k=1}^Q f_{u_k,2}(\xi_1,\xi_2,\tau),$$
(2.25)

$$R_u(\tau) = \sum_{k=1}^{Q} R_{u_k}(\tau),$$
(2.26)

where the last equality in (2.24) follows from (2.23). Furthermore, (2.25) implies that

$$G_{u}(\xi_{1},\tau) = \frac{\partial f_{u,2}(\xi_{1},\xi_{2},\tau)}{\partial \xi_{2}} \Big|_{\xi_{2}=0} = \sum_{k=1}^{Q} G_{u_{k}}(\xi_{1},\tau) \prod_{l=1,l\neq k}^{Q} f_{u_{l},2}(\xi_{1},0,\tau)$$
$$= \sum_{k=1}^{Q} G_{u_{k}}(\xi_{1},\tau) \prod_{l=1,l\neq k}^{Q} f_{u_{l},1}(\xi_{1}), \qquad (2.27)$$

where we have used that  $f_{u_l,2}(\xi_1, 0, \tau) = f_{u_l,1}(\xi_1)$  in the last equality. From (2.23) it follows that

$$f'_{u_k,1}(\xi_1) = \frac{\sigma_k^2}{\sigma_1^2} f_{u_1,1}(\xi_1)^{\sigma_k^2/\sigma_1^2 - 1} f'_{u_1,1}(\xi_1).$$
(2.28)

Since all  $u_k(t)$  are separable, (2.21) holds and by inserting (2.28) we obtain

$$G_{u_k}(\xi_1,\tau) = \frac{R_{u_k}(\tau)}{\sigma_1^2} f_{u_1,1}(\xi_1)^{\sigma_k^2/\sigma_1^2 - 1} f'_{u_1,1}(\xi_1).$$
(2.29)

Inserting (2.23) and (2.29) in (2.27) gives

$$G_{u}(\xi_{1},\tau) = \sum_{k=1}^{Q} \frac{R_{u_{k}}(\tau)}{\sigma_{1}^{2}} f_{u_{1},1}(\xi_{1}) \sum_{l=1}^{Q} \sigma_{l}^{2} / \sigma_{1}^{2-1} f_{u_{1},1}'(\xi_{1})$$
$$= \frac{d}{d\xi_{1}} \left( f_{u_{1},1}(\xi_{1}) \sum_{l=1}^{Q} \sigma_{l}^{2} / \sigma_{1}^{2} \right) \frac{\sum_{k=1}^{Q} R_{u_{k}}(\tau)}{\sum_{l=1}^{Q} \sigma_{l}^{2}} = f_{u,1}'(\xi_{1}) \frac{R_{u}(\tau)}{R_{u}(0)},$$

where we have used (2.24) and (2.26) in the last equality. Hence, Theorem 2.1 gives that u(t) is separable.

In Nuttall (1958a), two different results for sums of separable processes are presented. Both these results concern sufficient conditions for the separability of the sum of a finite number of independent separable processes. The first of these conditions is that the individual correlation functions should be equal, while the second is the condition on the characteristic functions restated here in Theorem 2.2. Furthermore, it is shown in Nuttall (1958a) that the product of two independent separable processes with zero mean always will be separable.

The reason why separable processes are useful for identification experiments is that they are the most general class of input signals for which a certain *invariance property* holds.

**Definition 2.8.** Consider a stationary stochastic process u(t) with E(u(t)) = 0 and  $R_u(\tau) < \infty$  for all  $\tau \in \mathbb{Z}$  and a static nonlinearity y(t) = f(u(t)) such that E(y(t)) = 0 and  $R_{yu}(\tau) < \infty$  for all  $\tau \in \mathbb{Z}$ . The *invariance property* holds if

$$R_{yu}(\tau) = b_0 R_u(\tau), \quad \forall \tau \in \mathbb{Z},$$
(2.30)

for some constant  $b_0$ .

 $\square$ 

It is easy to show that the separability of u(t) is a sufficient condition for the invariance property (2.30) to hold. Consider a separable process u(t) with zero mean and a static nonlinearity such that y(t) has zero mean too. Then it follows that

$$R_{yu}(\tau) = E(f(u(t))u(t-\tau)) = E(E(f(u(t))u(t-\tau)|u(t)))$$
  
=  $E(f(u(t))E(u(t-\tau)|u(t))) = a(\tau)E(f(u(t))u(t)) = b_0R_u(\tau),$  (2.31)

where  $b_0 = E(f(u(t))u(t))/R_u(0)$  and where (2.15) has been used in the last equality.

In a certain sense, separability is also a necessary condition for (2.30) to hold. Consider an arbitrary stationary stochastic process u(t) with zero mean and let  $D_u$  be a class of Lebesgue integrable functions such that

$$D_u = \{f : \mathbb{R} \to \mathbb{R} \mid \mathrm{E}(f(u(t))) = 0, \mathrm{E}(f(u(t))^2) < \infty, R_{yu}(\tau) = \mathrm{E}(f(u(t))u(t-\tau)) \text{ exists } \forall \tau \in \mathbb{Z}\}.$$
(2.32)

The following result shows a certain equivalence between the invariance property and separability of the input signal. In Section 6.3.2, this result will be extended to a more general type of nonlinear systems and thus the proof of Theorem 2.3 is omitted here.

#### Theorem 2.3

Consider a stationary stochastic process u(t) with E(u(t)) = 0 and  $R_u(\tau) < \infty$  for all  $\tau \in \mathbb{Z}$ . The invariance property (2.30) holds for all  $f \in D_u$  if and only if u(t) is separable.

**Proof:** See Nuttall (1958a).

For a separable process, it is easy to show that the mean-square error optimal LTI model of a static nonlinearity is a constant. However, this important property will not be described in this section but later in Chapter 6 when the LTI approximations based on the mean-square error have been properly defined. Several other results about separable processes are also presented in Nuttall (1958a), but since these results are not used in this thesis, they are not restated here.

After this discussion about the properties of LTI systems and some nonlinear systems and the brief introductions to the prediction-error method, random multisines and separable processes, we are now ready to move on to the main part of this thesis. First, however, we will in the next chapter give an overview of some of the linearization approaches that can be found in the literature.

3

# Methods for Linearization of Nonlinear Systems

Since there are many different circumstances when LTI models of nonlinear systems are useful, various linearization approaches have been proposed. These approaches differ in many aspects, for example in the types of signals they are defined for, in their optimality properties and in which mathematical tools that are used. In this chapter, a brief overview of some of the linearization frameworks that can be found in the literature will be given. First, we will in the next section consider linearizations in a deterministic framework while we in the second part of this chapter will discuss LTI approximations for stochastic signals.

# 3.1 Deterministic Approaches

The most straightforward linearization approach is perhaps to use some kind of local linearization based on a truncated Taylor series expansion. For example, if the true system is an NFIR system (see Section 2.2)

$$y(t) = f((u(t-k))_{k=0}^{M}),$$

it can for small inputs be approximated with an LTI system

$$G_0(z) = \sum_{k=0}^M g_0(k) z^{-k},$$

where  $g_0(k) = f'_{u(t-k)}(0)$ . Of course, this approximation is only well-defined if f is differentiable at 0. Similar LTI approximations can be calculated also for nonlinear state-space systems (see, for example, Ljung and Glad, 1994, pp. 347-349).

The idea of deriving an LTI approximation by differentiation of a nonlinear system is used also in Mäkilä and Partington (2003). There, LTI approximations of some classes

of nonlinear systems with  $l^{\infty}$  input and output signals are studied. It is shown that the *Fréchet derivative* of a Wiener, an NFIR or a bi-gain system is an optimal approximation in the sense that the induced gain of the model error is minimized. Furthermore, some relations to controller design and identification are discussed. Induced gains of linear and nonlinear systems for  $l^p$  signals are studied also in Partington and Mäkilä (2002). In particular, this paper contains some results about the accuracy of LTI approximations of piecewise linear systems.

LTI models that are optimal in the sense that they minimize an average squared error condition are characterized in Mäkilä (2004a) using a deterministic quasi-stationary signal framework. Actually, LTI approximations can be viewed as a special case of a general approximation problem for nonlinear systems studied in Mäkilä (2003a,b). These papers concern also approximations in an absolute error framework. In Mäkilä and Partington (2004), the least-squares LTI approximation problem from Mäkilä (2004a) is studied for a more general class of input signals. Furthermore, this paper presents conditions that guarantee quasi-stationarity of the output signal for some classes of nonlinear systems. Some of these results have also been published in the thesis Mäkilä (2004b).

LTI approximations of NFIR systems are considered in Mäkilä (2005). There, the concept of *nearly linear systems* is introduced. A nonlinear system is nearly linear if there exists an LTI system such that the  $l^{\infty}$  norm of the difference between the outputs of the nonlinear and the linear systems is bounded for all input signals. An LTI system with this property is called an *LTI companion* of the nonlinear system. For a nearly linear NFIR system, it turns out that a controller stabilizes the system if and only if it stabilizes its LTI companion. The importance of a careful input design for least-squares LTI approximations of NFIR systems is also discussed in Mäkilä (2005).

A different approach is used in Horowitz (1993). There, the LTI equivalent P(s) of a continuous-time nonlinear system that has the output y(t) for a particular input u(t) is defined as the ratio

$$P(s) = \frac{Y(s)}{U(s)},$$

where Y(s) and U(s) are the Laplace transforms of the output and input signal, respectively. Furthermore, it is shown that a set of such LTI equivalents can be used for controller design.

LTI approximations for a class of deterministic signals are also discussed in Sastry (1999). There, the existence of an optimal LTI model is mentioned and it is related to the theory of *describing functions*. A describing function is a parameterized linear approximation of a nonlinearity derived for a sinusoidal input signal. Describing functions are often used for analysis of closed-loop nonlinear systems (see, for example, Atherton, 1982; Sastry, 1999). LTI approximations for deterministic periodic signals are also investigated in Evans and Rees (2000a,b).

The concept of LTI-SOEs used in this thesis is, for example, discussed in Ljung (2001). However, a deterministic quasi-stationary framework is used in this paper while we here will study LTI-SOEs for stochastic signals. Despite this, the fundamental approximation results in Ljung (2001) are completely analogous to the ones that will be presented in Chapter 4.

## 3.2 Stochastic Approaches

Since the work of Wiener (1949), there has been a great activity in estimation and filtering using random signals. We will here present some of the existing approaches to linearization using stochastic signals and we begin with linearizations of static nonlinearities.

#### 3.2.1 Results for Static Nonlinearities

Many problems concerning the interplay between stochastic signals and nonlinear systems are difficult to solve. However, some nonlinear systems can be written as combinations of LTI subsystems and static nonlinear functions. For example, LTI systems with input and/or output saturation turn out to be very common in applications.

It is usually easier to analyze a static nonlinearity than a dynamic one. Hence, it is no surprise that there has been a wide interest in understanding how a static nonlinearity can be linearized for a stochastic input signal. Many results in this area are directly or indirectly related to Bussgang's classic theorem about Gaussian signals (see Bussgang (1952) for the original report and, for example, Papoulis (1984) for a more recent reference).

In Bussgang (1952), it is shown that the cross-covariance function between the output and the input of a static nonlinear function is a scaled version of the input covariance function if the input is Gaussian, i.e., that the invariance property in Definition 2.8 holds in this case. This result is summarized in the following theorem.

#### Theorem 3.1 (Bussgang)

Let y(t) be the output from a differentiable static nonlinearity f with a Gaussian input u(t), i.e., y(t) = f(u(t)). Assume that the expectations E(y(t)) = E(u(t)) = 0, that the cross-covariance function  $R_{yu}(\tau)$  and the covariance function  $R_u(\tau)$  are well-defined for all  $\tau \in \mathbb{Z}$  and that E(f'(u(t))) exists. Then it holds that

$$R_{yu}(\tau) = b_0 R_u(\tau),$$

where  $b_0 = E(f'(u(t)))$ .

A direct implication of this result is that the mean-square error optimal LTI approximation of a static nonlinearity always will be a constant  $b_0 = E(f'(u(t)))$  when the input to the nonlinearity is Gaussian (cf. Section 6.1). It will be shown later in this thesis that this is not always true in the non-Gaussian case (see, for example, Example 4.2). The constant  $b_0$  is called *equivalent gain* in Booton (1954) an can be viewed as a describing function for a random input signal. Just like ordinary describing functions, it can be used to analyze nonlinear closed-loop systems (Atherton, 1982, Chap. 8). The relation between Bussgang's theorem and some other results about Gaussian processes is discussed in Gorman and Zaborszky (1968).

Furthermore, Bussgang's theorem has been generalized to other classes of signals than Gaussian in Barrett and Lampard (1955), Brown (1957) and Nuttall (1958a). Nuttall's generalization is to the class of separable processes defined in Section 2.4. This class of signals is especially interesting since it contains the other two, and since it gives a useful necessary and sufficient condition on the input signal for the invariance property in Definition 2.8 to hold for any static nonlinearity in a wide class of functions (cf. Theorem 2.3). As was mentioned in Section 2.4, a number of separable signals are listed in Nuttall (1958a), e.g., Gaussian processes, random binary processes and several types of modulated processes. In addition, it is shown in McGraw and Wagner (1968) that signals with *elliptically symmetric distributions* are separable and they have also characterized these signals further.

A result related to separable processes can be found in Balakrishnan (1960). There, it is shown that processes for which conditional expectations like  $E(u(t - \sigma)|u(t))$  are of a specified form have characteristic functions that satisfy certain equations.

#### 3.2.2 Results for Hammerstein and Wiener Systems

Separable processes in general, and Gaussian processes in particular, have turned out to be very useful for system identification of Wiener-Hammerstein systems (cf. Section 2.2). The reason for this is that the use of a separable input signal makes it possible to estimate the linear part of a Hammerstein system without compensating for the nonlinearity at the input. The corresponding result holds also for Wiener-Hammerstein systems with Gaussian inputs. Some results about identification of Wiener, Hammerstein or Wiener-Hammerstein systems using Gaussian signals can, for example, be found in Billings and Fakhouri (1978, 1982), Korenberg (1985), Hunter and Korenberg (1986) and Bendat (1998). Gaussian signals are not the only signals that have been studied in this context. For example, it is also possible to estimate the linear and nonlinear parts of a Wiener or Hammerstein system separately when the input is sinusoidal (Bai, 2003a,b). The case of a Hammerstein system with a pseudo-random binary input signal is studied in Bai (2004).

In this thesis, we will discuss linearization of Hammerstein systems in some detail (see Chapter 6). This system structure is common in many real-life applications and it is thus natural that identification of Hammerstein systems has been an active research field for quite some time. A brief overview of some of the existing methods can be found in Bai and Li (2004).

One popular method for identification of Hammerstein systems was originally proposed in Narendra and Gallman (1966) and has been studied in many papers (see, for example, Stoica, 1981; Bai and Li, 2004). In this method, a Hammerstein model with independent parameterizations of the linear and the nonlinear submodels is estimated by minimizing a quadratic cost function iteratively starting from an initial estimate, or guess, of the parameter values. In each step, the cost function is minimized with respect only to one of its arguments while using the previous value of the other one. Often, an LTI approximation of the system is first estimated without considering the nonlinearity at the input and the parameters from this approximate model are used as parameters of the linear submodel when the first estimate of the nonlinearity is computed in the next step. For example, the initial LTI model can be obtained by estimating an OE model using the prediction-error method (see Section 2.3.1).

The iterative approach guarantees that the cost function will be monotonically decreasing over the iterations. Furthermore, the convergence of the parameter estimates to the true parameter values can be shown in some special cases (Bai and Li, 2004). However, there are no such convergence results available for the general case. Hence, it is important that the initial LTI approximation is as good as possible in the sense that it resembles the true LTI subsystem as well as possible.

For a separable input signal, the iterative approach can be motivated by the fact that the

invariance property holds, since this property implies that the LTI model that minimizes the mean-square error  $E((y(t) - G(q)u(t))^2)$  will be equal to a scaled version of the LTI part of the system.

For nonseparable input signals, the estimation of the initial LTI model requires more attention. Some results concerning this problem for Wiener and Hammerstein systems with random multisine inputs can be found in Crama and Schoukens (2001, 2004). The approach has also been generalized to Hammerstein-Wiener and Wiener-Hammerstein systems in Crama and Schoukens (2004) and Crama and Schoukens (2005), respectively. All these papers concern random multisine inputs and use a nonparametric frequency response function estimate as the very first LTI model estimate. Some related material can also be found in Vandersteen et al. (1997) and Vandersteen and Schoukens (1999).

#### 3.2.3 Results for General Nonlinear Systems

Bussgang's theorem has been generalized to functions of several variables (Atalik and Utku, 1976; Scarano et al., 1993; Lutes and Sarkani, 1997, Chap. 9). This generalization can be used to characterize optimal linear approximations of NFIR systems, as will be shown later in Chapter 7. However, the generalized version of Bussgang's theorem has been used mainly for linear approximations of continuous-time nonlinear state-space systems in the field of stochastic mechanical vibrations. This linearization approach is there known under names such as equivalent linearization, statistical linearization and stochastic linearization and it can, for example, be used to compute an approximation of the variance of the output from a nonlinear system.

LTI approximations of nonlinear systems for stochastic signals are often studied in a mean-square error framework. In this case, the optimal LTI model of a nonlinear system with input u(t) and output y(t) can be defined as the stable model  $G_0(q)$  that minimizes the mean-square error  $E((y(t) - G(q)u(t))^2)$ . It is a well-known result that this optimal approximation can be written

$$G_0(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)}$$

and it is sometimes called the *equivalent linear model* (see, for example, Gardner, 1986, p. 382). The model  $G_0(z)$  can also be called the *noncausal Wiener filter*, since it in general will be a stable but noncausal LTI system. In this thesis, we will also call it the *noncausal LTI-SOE* (cf. Chapter 5).

If the LTI approximation is assumed to be causal, it will in general not be equal to the ratio between  $\Phi_{yu}(z)$  and  $\Phi_u(z)$ . Instead, it can be derived using techniques for causal Wiener filtering (see, for example, Kailath et al., 2000, Chap. 7). The mean-square error optimal stable and *causal* LTI approximation of a nonlinear system has been discussed in Schetzen (1980, p. 330) and is also the main topic of this thesis.

Optimal LTI approximations are also discussed in Pintelon and Schoukens (2001). There, the term *related dynamic system* is used for the mean-square error optimal LTI model and the part of the output signal that this model cannot explain is viewed as a *non-linear distortion*. This name is natural since the effects of unmodeled nonlinearities on nonparametric frequency response estimates can be similar to the effects of measurement noise. For some special classes of input signals, including random multisines, a number of interesting properties of related linear systems can be derived. For example, the

asymptotic behavior of the related dynamic system when the number of excited frequencies for a random multisine input signal tends to infinity has been studied. It turns out that such a signal in many ways is similar to a Gaussian signal when the number of frequency components is large.

Fundamental results about, for example, related dynamic systems, nonparametric frequency response estimates and random multisines can be found in Schoukens et al. (1998), Schoukens et al. (2001), Pintelon et al. (2001), Schoukens et al. (2002), Pintelon and Schoukens (2002), Pintelon et al. (2003) and Schoukens et al. (2005a). These results have been applied to various identification problems. In Schoukens et al. (2004a), the risk for instability in nonlinear closed-loop systems is investigated, while closed-loop identification is discussed in Schoukens et al. (2005b). An approach where approximate models consisting of an LTI part, a Wiener part, a Hammerstein part and a Wiener-Hammerstein part are estimated is described in Schoukens et al. (2003). Furthermore, some input design issues are discussed in Vanhoenacker et al. (2001) and Vanhoenacker and Schoukens (2003). Related results can also be found in Dobrowiecki and Schoukens (2001), Vandersteen et al. (2001) and Schoukens et al. (2004b).

In this chapter, an overview of some existing linearization approaches has been given. These approaches deal either with deterministic or stochastic signals and have rather different properties. In this thesis, we will consider only LTI approximations for stochastic signals and these approximations will be described in the next chapter.

# Part I LTI-SOEs

4

# The Notion of LTI Second Order Equivalents

The previous chapter contained an overview of the various linearization frameworks that can be found in the literature. All these frameworks have their benefits and drawbacks depending on which type of linearization is desired. In this thesis, the main objective is to understand the behavior of the prediction-error method when the measured input and output signals come from a nonlinear system.

Hence, with the discussion from Section 2.3 about the asymptotic properties of the prediction-error method in mind, it is here natural to study linear approximations of nonlinear systems that are optimal in the mean-square error sense. Such approximations will be called *LTI Second Order Equivalents* (LTI-SOEs) of the nonlinear systems. This chapter contains both detailed derivations of two types of LTI-SOEs and some interpretations of these approximations. First, some restrictions on the input and output signals will be imposed in order to make the LTI-SOEs well-defined.

# 4.1 Assumptions on the Input and Output Signals

Since the class of nonlinear systems literally contains all kinds of systems, it is too general to be studied as a whole. In many cases, explicit restrictions on the considered types of nonlinear systems are introduced in order to enable further analyses of the properties of these systems. Examples of explicit restrictions are that the nonlinear systems should have finite gain, finite memory or some kind of stability property.

Also in this thesis, the class of nonlinear systems has to be restricted. However, we will not impose any explicit restrictions but instead assume that the input and output signals of the nonlinear systems have certain properties. These signal assumptions are listed here and impose implicit restrictions on the class of nonlinear systems that will be studied in the sequel.

#### Assumption A1. Assume that

(i) The input u(t) is a real-valued stationary stochastic process with

$$\mathbf{E}(u(t)) = 0.$$

(ii) There exist K > 0 and  $\alpha$ ,  $0 < \alpha < 1$  such that the second order moment  $R_u(\tau) = E(u(t)u(t-\tau))$ , satisfies

$$|R_u(\tau)| < K\alpha^{|\tau|}, \quad \forall \tau \in \mathbb{Z}.$$

(iii) The z-spectrum  $\Phi_u(z)$  (i.e., the z-transform of  $R_u(\tau)$ ) has a unique canonical spectral factorization

$$\Phi_u(z) = L(z)r_u L(z^{-1}), \tag{4.1}$$

where L(z) and 1/L(z) are causal transfer functions that are analytic in the set  $\{z \in \mathbb{C} \mid |z| \ge 1\}, L(\infty) \triangleq \lim_{|z| \to \infty} L(z) = 1$  and  $r_u$  is a positive constant.

#### Assumption A2. Assume that

(i) The output y(t) is a real-valued stationary stochastic process with

$$\mathbf{E}(y(t)) = 0.$$

(ii) There exist K > 0 and  $\alpha$ ,  $0 < \alpha < 1$  such that the second order moments  $R_{yu}(\tau) = E(y(t)u(t-\tau))$  and  $R_y(\tau) = E(y(t)y(t-\tau))$  satisfy

$$\begin{split} |R_{yu}(\tau)| &< K\alpha^{|\tau|}, \quad \forall \tau \in \mathbb{Z}, \\ |R_y(\tau)| &< K\alpha^{|\tau|}, \quad \forall \tau \in \mathbb{Z}. \end{split}$$

In this thesis, Assumptions A1 and A2 are the standard assumptions on the input and output signals. However, some signals that do not satisfy these assumptions will be discussed too. For these signals, the following assumptions will be used.

#### Assumption A3 (Periodic input). Assume that

(i) The input u(t) and output y(t) are real-valued stationary stochastic processes with

$$\mathbf{E}(u(t)) = \mathbf{E}(y(t)) = 0.$$

(ii) Assume that u(t) is *P*-periodic for some  $P \in \mathbb{Z}_+$ , i.e., that

$$u(t+P) = u(t), \quad \forall t \in \mathbb{Z}.$$

(iii) Assume that the second order moments  $R_u(\tau) = E(u(t)u(t-\tau))$  and  $R_{yu}(\tau) = E(y(t)u(t-\tau))$  exist and are *P*-periodic.

In Assumptions A1(i), A2(i) and A3(i) it is required that both the input and the output signal have zero mean. In practice, this assumption does not exclude systems with input and output signals that vary around a nonzero set point from being analyzed using the results in this thesis. For such systems, it is always possible to define new input and output signals that describe the deviations from the set point by subtracting the corresponding means of the two signals. By their construction, these new signals will have zero mean and they will hence satisfy the zero mean assumption in this thesis.

Assumptions A1(ii) and A2(ii) imply that the second order moments of u(t) and y(t) are bounded and that the z-spectra  $\Phi_u(z)$  and  $\Phi_y(z)$  and the z-cross-spectrum  $\Phi_{yu}(z)$  converge absolutely and are analytic in the annulus  $\{z \in \mathbb{C} \mid \alpha < |z| < \frac{1}{\alpha}\}$ . It should be mentioned that the canonical spectral factorization in Assumption A1(iii) exists for all rational  $\Phi_u(z)$  without zeros on the unit circle (Kailath et al., 2000, pp. 198-199). For a rational  $\Phi_u(z)$ , the conditions on L(z) in Assumption A1(iii) mean just that it should be a monic minimum phase filter. However, we will not restrict ourselves only to rational spectra here. The following example shows that also a nonrational z-spectrum can have a canonical spectral factorization.

—— Example 4.1 –

The z-spectrum

$$\Phi_u(z) = e^{1/z + z}$$

can be factorized as  $\Phi_u(z) = L(z)r_uL(z^{-1})$  with  $L(z) = e^{1/z}$  and  $r_u = 1$ . The causal series expansions

$$L(z) = e^{1/z} = \sum_{k=0}^{\infty} \frac{1}{k!} z^{-k},$$
$$L^{-1}(z) = e^{-1/z} = \sum_{k=0}^{\infty} \frac{(-1)^k}{k!} z^{-k}$$

converge absolutely and are analytic for  $\{z \in \mathbb{C} \mid |z| > 0\}$  and  $L(\infty) = 1$ . Hence,  $L(z)r_uL(z^{-1})$  is the unique canonical spectral factorization of the nonrational z-spectrum  $\Phi_u(z)$ .

Assumptions A1 and A2 give sufficient conditions for the type of LTI-SOE that will be defined in Section 4.2 to be well-defined. However, for the second type of LTI-SOE that will be introduced in Section 4.4 we will need another assumption. Let  $\zeta(t) = (u(t) y(t-1))^T$ . Then it follows that

$$R_{\zeta}(\tau) = \begin{pmatrix} R_u(\tau) & R_{uy}(\tau+1) \\ R_{yu}(\tau-1) & R_y(\tau) \end{pmatrix},$$
(4.2a)

$$\Phi_{\zeta}(z) = \begin{pmatrix} \Phi_u(z) & z\Phi_{uy}(z) \\ z^{-1}\Phi_{yu}(z) & \Phi_y(z) \end{pmatrix}.$$
(4.2b)

Assumption A4. Assume that the signals u(t) and y(t) fulfill Assumptions A1 and A2 and that they also are such that  $\Phi_{\zeta}(z)$  in (4.2b) has a unique canonical spectral factorization

$$\Phi_{\zeta}(z) = T(z)Q_{\zeta}T^{T}(z^{-1}),$$



Figure 4.1: The output error model.

where T(z) and  $T^{-1}(z)$  are analytic in  $\{z \in \mathbb{C} \mid |z| \ge 1\}$ ,  $T(\infty) = I$  and  $Q_{\zeta}$  is a positive definite matrix, i.e.,  $Q_{\zeta} \succ 0$ . (Here,  $T^{-1}$  denotes the matrix inverse.)

Also the existence of a canonical factorization of a matrix-valued z-spectrum is guaranteed if, for example,  $\Phi_{\zeta}(z)$  is a rational matrix without unit circle zeros (Kailath et al., 2000, p. 205). Assumptions A1, A2 and A4 will be used in the derivations of LTI-SOEs in the next section and in Section 4.4 and will be the standard assumptions throughout this thesis.

# 4.2 The Output Error Model Type

In Section 2.1 it was mentioned that a general LTI model can be written as

$$y(t) = G(q)u(t) + H(q)e(t).$$

In general, both G and H in this model can be transfer functions of any order. Here, however, at first we will only consider models where H is fixed to 1, i.e., *output error* models (Ljung, 1999). If G is causal this implies that only the input components

$$u(t), u(t-1), u(t-2), \dots$$

are used to predict the output y(t) according to (2.2). The structure of an output error model is shown in Figure 4.1.

Using only output error models, the mean-square error optimal LTI approximation of a certain nonlinear system is simply the stable and causal LTI model  $G_{0,OE}$  that minimizes  $E((y(t) - G(q)u(t))^2)$ . This model is often called the Wiener filter for prediction of y(t) from  $(u(t-k))_{k=0}^{\infty}$  (Wiener, 1949). However, we will not use the term Wiener filter here, but instead call  $G_{0,OE}$  the Output Error LTI Second Order Equivalent (OE-LTI-SOE) of the nonlinear system. Hence, we have the following definition.

**Definition 4.1.** Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled. The *Output Error LTI Second Order Equivalent* (OE-LTI-SOE) of this system is the stable and causal LTI model  $G_{0,OE}(q)$  that minimizes the mean-square error  $E((y(t) - G(q)u(t))^2)$ , i.e.,

$$G_{0,OE}(q) = \operatorname*{arg\,min}_{G \in \mathcal{G}} \mathrm{E}((y(t) - G(q)u(t))^2),$$

where  $\mathcal{G}$  denotes the set of all stable and causal LTI models.

39

There are two main reasons for the change of name from the commonly used Wiener filter to OE-LTI-SOE. First, we want to avoid any ambiguities. Many different Wiener filters can be constructed for a given pair of input and output signals. Here, however, we are only interested in the Wiener filter that predicts y(t) from  $(u(t-k))_{k=0}^{\infty}$ .

The second reason for the change of name is that we want to emphasize that the OE-LTI-SOE is an equivalent to the nonlinear system in the sense that it can explain the causal part of the cross-covariance function  $R_{yu}(\tau)$  between the input and the output of the system. This observation, which is rather obvious for OE-LTI-SOEs (see Corollary 4.2), becomes more interesting if we study LTI models that contain a general error description, i.e., models with  $H \neq 1$ . LTI-SOEs for this case will be defined later in Section 4.4 and will be called *General Error LTI Second Order Equivalent* (GE-LTI-SOE). In Section 4.4, it will be shown that GE-LTI-SOEs can explain both the covariance function  $R_{yu}(\tau)$  and the cross-covariance function  $R_{yu}(\tau)$ . Hence, such a model is an equivalent to the non-linear system in the sense that it is impossible to distinguish it from the true system if only second order properties of the input, output and model residuals are considered. This will be discussed in more detail in Section 4.4 (see the comments to Corollary 4.4 on page 56).

It should be noted that we are not only interested in the filtering and prediction capabilities of the OE-LTI-SOE (and the GE-LTI-SOE), but also in the model itself. For example, we are not only interested in how good an estimate of y(t) the model can produce, but also in issues like how the model order and model coefficients depend on the nonlinear system and on the input signal.

It should immediately be pointed out that the OE-LTI-SOE of a nonlinear system depends on which input signal is used. Hence, we can only talk about the OE-LTI-SOE of a nonlinear system with respect to a particular input signal. The fact that the OE-LTI-SOE is input-dependent is natural if we view it as an example of undermodeling. Undermodeling occurs when a system is approximated by a model of lower complexity. An often studied example of undermodeling is when an LTI system is approximated with an LTI model of lower order than the true system. Actually, also in this case of linear undermodeling the mean-square error optimal approximation is input-dependent (Ljung, 1999, Sec. 8.5).

Another property of OE-LTI-SOEs that should be pointed out, besides their inputdependency, is that they are assumed to be initialized at  $t = -\infty$  such that all transients have died out. In Definition 4.1 it is assumed that the complete infinite sequence

$$(u(t-k))_{k=0}^{\infty}$$

of random variables is available for the computation of an estimate of y(t).

The following theorem is a direct consequence of classic Wiener filter theory, and the proof of the theorem is almost identical to the derivation of the scalar Wiener filter in Kailath et al. (2000, pp. 231-234). It is included here for the sake of completeness.

#### Theorem 4.1 (OE-LTI-SOEs)

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled. Then the OE-LTI-SOE  $G_{0,OE}$  of this system is

$$G_{0,OE}(z) = \frac{1}{r_u L(z)} \left[ \frac{\Phi_{yu}(z)}{L(z^{-1})} \right]_{\text{causal}},$$
(4.3)

where  $[\ldots]_{\text{causal}}$  denotes taking the causal part (see Section 2.1), and where L(z) is the canonical spectral factor of  $\Phi_u(z)$  from (4.1).

**Proof:** The criterion  $E((y(t) - G(q)u(t))^2)$  that  $G_{0,OE}$  should minimize is equivalent to the Wiener-Hopf condition

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} g_{0,OE}(k) R_u(\tau - k) = 0, \quad \tau \ge 0$$
(4.4)

or, alternatively, to

$$\Phi_{yu}(z) - G_{0,OE}(z)\Phi_u(z) = K_{OE}(z), \tag{4.5}$$

where  $K_{OE}(z)$  is a stable and strictly anticausal transfer function. The Wiener-Hopf criterion follows from the fact that for the optimal model, the error

$$y(t) - G_{0,OE}(q)u(t)$$

should be orthogonal to  $\{u(t-k)\}_{k=0}^{\infty}$  (using the inner-product  $\langle u, v \rangle = E(uv)$ ). Using the spectral factorization according to (4.1) and multiplying by  $L^{-1}(z^{-1})$  now gives

$$\tilde{K}_{OE}(z) \triangleq K_{OE}(z)L^{-1}(z^{-1}) = \Phi_{yu}(z)L^{-1}(z^{-1}) - G_{0,OE}(z)L(z)r_u,$$
(4.6)

where  $\tilde{K}_{OE}(z)$  is a stable and strictly anticausal transfer function due to the fact that it is a product of the stable and strictly anticausal transfer function  $K_{OE}(z)$  and the stable and anticausal transfer function  $L^{-1}(z^{-1})$ . For (4.6) to hold it is necessary that the second right hand term, which is a stable and causal transfer function, is equal to the causal part of the first, i.e.,

$$G_{0,OE}(z)L(z)r_u = \left[\frac{\Phi_{yu}(z)}{L(z^{-1})}\right]_{\text{causal}}$$
(4.7)

 $\square$ 

and (4.3) follows.

In general, the OE-LTI-SOE has to be calculated as in (4.3), which means that the canonical spectral factor L(z) of the input z-spectrum has to be obtained. However, in some cases this is not necessary and the OE-LTI-SOE can be calculated using a simplified expression. This is shown in the following corollary.

#### Corollary 4.1

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled, and assume that the ratio  $\Phi_{yu}(z)/\Phi_u(z)$  defines a stable and causal LTI system. Then

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)}.$$
(4.8)

**Proof:** Assume that

$$C(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)}$$

is a stable and causal transfer function. Then

$$\Phi_{yu}(z) = C(z)\Phi_u(z) = C(z)L(z)r_uL(z^{-1})$$

and (4.3) gives

$$G_{0,OE}(z) = \frac{1}{r_u L(z)} \left[ \frac{C(z)L(z)r_u L(z^{-1})}{L(z^{-1})} \right]_{\text{causal}} = C(z),$$

since  $C(z)L(z)r_u$  is a stable and causal transfer function.

The OE-LTI-SOE of a system will be called *regular* if (4.8) holds. Hence, we have the following definition.

**Definition 4.2.** An OE-LTI-SOE  $G_{0,OE}(z)$  is regular if it can be written

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)}.$$
(4.9)

The Wiener-Hopf condition (4.4) implies that the model residuals for the OE-LTI-SOE are uncorrelated with past and current inputs. This is stated more clearly in the following corollary.

#### **Corollary 4.2**

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled. Let the residuals be defined by

$$\eta_0(t) = y(t) - G_{0,OE}(q)u(t).$$
(4.10)

Then

$$\Phi_{\eta_0 u}(z) = \Phi_{yu}(z) - G_{0,OE}(z)\Phi_u(z) \quad is \ strictly \ anticausal$$
(4.11)

and

$$R_{\eta_0}(0) = R_y(0) - \frac{1}{2\pi} \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega})|^2 \Phi_u(e^{i\omega}) \, d\omega.$$
(4.12)

**Proof:** The result (4.11) follows directly from (4.5). Furthermore, we can write

$$\begin{split} \Phi_{\eta_0}(z) &= \Phi_y(z) - \Phi_{yu}(z)G_{0,OE}(z^{-1}) - G_{0,OE}(z)\Phi_{uy}(z) \\ &+ G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1}) \\ &= \Phi_y(z) - (\Phi_{yu}(z) - G_{0,OE}(z)\Phi_u(z))G_{0,OE}(z^{-1}) \\ &- G_{0,OE}(z) \left(\Phi_{uy}(z) - \Phi_u(z)G_{0,OE}(z^{-1})\right) - G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1}). \end{split}$$

Since  $G_{0,OE}(z^{-1})$  is anticausal and since  $\Phi_{yu}(z) - G_{0,OE}(z)\Phi_u(z)$  by (4.11) is strictly anticausal, it follows that

$$\int_{-\pi}^{\pi} \left( \Phi_{yu}(e^{i\omega}) - G_{0,OE}(e^{i\omega}) \Phi_u(e^{i\omega}) \right) G_{0,OE}(e^{-i\omega}) \, d\omega = 0$$

and that

$$\int_{-\pi}^{n} G_{0,OE}(e^{i\omega}) \left( \Phi_{uy}(e^{i\omega}) - \Phi_{u}(e^{i\omega})G_{0,OE}(e^{-i\omega}) \right) d\omega = 0.$$

Because

$$R_{\eta_0}(0) = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\eta_0}(e^{i\omega}) \, d\omega$$

we get

$$R_{\eta_0}(0) = \underbrace{\frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_y(e^{i\omega}) \, d\omega}_{=R_y(0)} - \frac{1}{2\pi} \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega})|^2 \Phi_u(e^{i\omega}) \, d\omega.$$

Corollary 4.2 shows that the OE-LTI-SOE really is the best output error model of the true system as the remaining residuals  $\eta_0(t)$  are uncorrelated with past and present input signal components.

An example of the OE-LTI-SOE of a simple nonlinear system can be found below. Note that the OE-LTI-SOE is nonstatic in this example although the nonlinear system is static.

Consider the static nonlinear system

$$y(t) = u(t)^3 (4.13)$$

with the input

$$u(t) = e(t) + \frac{1}{2}e(t-1),$$

where e(t) is a sequence of independent random variables with uniform distribution over the interval [-1, 1]. Hence,  $E(e(t)^2) = \frac{1}{3}$  and  $E(e(t)^4) = \frac{1}{5}$  and, using the fact that e(t)and e(t-1) are independent, we get

$$\begin{aligned} R_{yu}(0) &= \mathcal{E}(u(t)^4) = \mathcal{E}(e(t)^4) + \frac{6}{4} \mathcal{E}(e(t)^2 e(t-1)^2) + \frac{1}{16} \mathcal{E}(e(t-1)^4) \\ &= \frac{1}{5} + \frac{6}{4} \cdot \frac{1}{9} + \frac{1}{16} \cdot \frac{1}{5} = \frac{91}{240}, \\ R_{yu}(1) &= \mathcal{E}(u(t)^3 u(t-1)) \\ &= \mathcal{E}(\left(e(t)^3 + \frac{3}{2}e(t)^2 e(t-1) + \frac{3}{4}e(t)e(t-1)^2 + \frac{1}{8}e(t-1)^3\right) \cdot \\ &\quad \cdot \left(e(t-1) + \frac{1}{2}e(t-2)\right)) = \frac{3}{2} \mathcal{E}(e(t)^2 e(t-1)^2) + \frac{1}{8} \mathcal{E}(e(t-1)^4) \\ &= \frac{3}{2} \cdot \frac{1}{9} + \frac{1}{8} \cdot \frac{1}{5} = \frac{46}{240}, \end{aligned}$$

$$\begin{aligned} R_{yu}(-1) &= \mathcal{E}(u(t)^{3}u(t+1)) \\ &= \mathcal{E}(\left(e(t)^{3} + \frac{3}{2}e(t)^{2}e(t-1) + \frac{3}{4}e(t)e(t-1)^{2} + \frac{1}{8}e(t-1)^{3}\right) \cdot \\ &\cdot \left(e(t+1) + \frac{1}{2}e(t)\right)) = \frac{1}{2}\mathcal{E}(e(t)^{4}) + \frac{3}{8}\mathcal{E}(e(t)^{2}e(t-1)^{2}) \\ &= \frac{1}{2} \cdot \frac{1}{5} + \frac{3}{8} \cdot \frac{1}{9} = \frac{34}{240}, \\ R_{yu}(\tau) &= 0 \quad \forall \tau \in \mathbb{Z} \setminus \{-1, 0, 1\}. \end{aligned}$$

This gives

$$\Phi_{yu}(z) = \frac{1}{240} (34z + 91 + 46z^{-1}) = \frac{1}{240} \left(1 + \frac{1}{2}z\right) \left(68 + 46z^{-1}\right).$$

Furthermore, Lemma 2.1 gives

$$\Phi_u(z) = \left(1 + \frac{1}{2}z^{-1}\right) \cdot \frac{1}{3} \cdot \left(1 + \frac{1}{2}z\right) = \frac{1}{12}\left(2z + 5 + 2z^{-1}\right)$$

and hence

$$\frac{\Phi_{yu}(z)}{\Phi_u(z)} = \frac{1}{40} \cdot \frac{34 + 23z^{-1}}{1 + \frac{1}{2}z^{-1}}$$

Since the ratio  $\Phi_{yu}(z)/\Phi_u(z)$  here is stable and causal, the OE-LTI-SOE of the system (4.13) for this input is

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)} = \frac{1}{40} \cdot \frac{34 + 23z^{-1}}{1 + \frac{1}{2}z^{-1}} = \frac{0.85 + 0.575z^{-1}}{1 + 0.5z^{-1}}.$$

Note that although the nonlinear system is static the OE-LTI-SOE is not.

The fact that a static nonlinear system can have a nonstatic OE-LTI-SOE is a first indication that LTI approximations of nonlinear systems in a mean-square error framework are not as straightforward as one might expect. More examples of this will be presented later in this thesis. In Chapters 6 and 7 we will also discuss classes of input signals which guarantee that the output of the OE-LTI-SOE will depend on the same number of input signal components as the nonlinear system.

For most systems, the order of the OE-LTI-SOE is unknown. In practice, this implies that several output error models have to be estimated and that a validation procedure has to be used in order to find the best model. Naturally, there is no guarantee that the correct order of the OE-LTI-SOE will be found. As a matter of fact, the OE-LTI-SOE can sometimes be an infinite order model. Hence, it is interesting to characterize in what sense an output error model with lower order than the OE-LTI-SOE approximates the OE-LTI-SOE.

This is a relevant question also when the true system is an LTI system. In that case, it can be shown that a low order model will approximate the true system mainly for frequencies where  $\Phi_u(e^{i\omega})$  is large (Ljung, 1999, p. 266). As a matter of fact, this result holds

also when the true system is nonlinear. In this case, a low order output error model will approximate the OE-LTI-SOE instead of the true system as well as possible for frequencies where  $\Phi_u(e^{i\omega})$  is large according to the following theorem. This theorem is basically a special case of Theorem 4.1 in Ljung (2001) and the proof is very similar to the outlined proof in Problem 8G.5 in Ljung (1999).

#### Theorem 4.2

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled. Let  $G_{0,OE}$  be the corresponding OE-LTI-SOE according to Theorem 4.1. Suppose that a parameterized stable and causal output error model  $G(q, \theta)$  is fitted to the signals u and y according to

$$\hat{\theta} = \arg\min_{\theta} \mathcal{E}(\eta(t,\theta)^2), \qquad (4.14)$$

where

$$\eta(t,\theta) = y(t) - G(q,\theta)u(t). \tag{4.15}$$

Then it follows that

$$\hat{\theta} = \arg\min_{\theta} \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G(e^{i\omega},\theta)|^2 \Phi_u(e^{i\omega}) \, d\omega.$$
(4.16)

**Proof:** The z-spectrum of  $\eta(t, \theta)$  is

$$\begin{split} \Phi_{\eta}(z,\theta) &= \begin{pmatrix} -G(z,\theta) & 1 \end{pmatrix} \begin{pmatrix} \Phi_{u}(z) & \Phi_{uy}(z) \\ \Phi_{yu}(z) & \Phi_{y}(z) \end{pmatrix} \begin{pmatrix} -G(z^{-1},\theta) \\ 1 \end{pmatrix} \\ &= \Phi_{y}(z) - G(z,\theta)\Phi_{uy}(z) - G(z^{-1},\theta)\Phi_{yu}(z) + G(z,\theta)\Phi_{u}(z)G(z^{-1},\theta) \\ &= \begin{pmatrix} G(z,\theta) - \frac{\Phi_{yu}(z)}{\Phi_{u}(z)} \end{pmatrix} \Phi_{u}(z) \begin{pmatrix} G(z^{-1},\theta) - \frac{\Phi_{yu}(z^{-1})}{\Phi_{u}(z^{-1})} \end{pmatrix} \\ &- \frac{\Phi_{yu}(z)\Phi_{yu}(z^{-1})}{\Phi_{u}(z)} + \Phi_{y}(z). \end{split}$$

Let

$$A_{0} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \left( \Phi_{y}(e^{i\omega}) - \frac{|\Phi_{yu}(e^{i\omega})|^{2}}{\Phi_{u}(e^{i\omega})} \right) d\omega,$$
$$B_{0} = \frac{1}{2\pi} \int_{-\pi}^{\pi} \frac{|\Phi_{\eta_{0}u}(e^{i\omega})|^{2}}{\Phi_{u}(e^{i\omega})} d\omega,$$

where  $\eta_0$  is the residual signal defined in (4.10). Parseval's relation gives

$$E(\eta(t,\theta)^2) = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\eta}(e^{i\omega},\theta) \, d\omega$$

$$\begin{split} &= \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| \frac{\Phi_{yu}(e^{i\omega})}{\Phi_u(e^{i\omega})} - G(e^{i\omega},\theta) \right|^2 \Phi_u(e^{i\omega}) \, d\omega + A_0 \\ &= \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| G_{0,OE}(e^{i\omega}) + \frac{\Phi_{\eta_0 u}(e^{i\omega})}{\Phi_u(e^{i\omega})} - G(e^{i\omega},\theta) \right|^2 \Phi_u(e^{i\omega}) \, d\omega + A_0 \\ &= \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| G_{0,OE}(e^{i\omega}) - G(e^{i\omega},\theta) \right|^2 \Phi_u(e^{i\omega}) \, d\omega \\ &+ \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\eta_0 u}(e^{i\omega}) (G_{0,OE}(e^{-i\omega}) - G(e^{-i\omega},\theta)) \, d\omega \\ &+ \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\eta_0 u}(e^{-i\omega}) (G_{0,OE}(e^{i\omega}) - G(e^{i\omega},\theta)) \, d\omega + A_0 + B_0, \end{split}$$

where we have used (4.11) in the third equality. Since  $\Phi_{\eta_0 u}(z)$  by (4.11) is strictly anticausal and since  $G_{0,OE}(z)$  and  $G(z, \theta)$  both are causal, a term-by-term integration shows that

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\eta_0 u}(e^{i\omega}) (G_{0,OE}(e^{-i\omega}) - G(e^{-i\omega},\theta)) \, d\omega = 0,$$
  
$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\eta_0 u}(e^{-i\omega}) (G_{0,OE}(e^{i\omega}) - G(e^{i\omega},\theta)) \, d\omega = 0.$$

Thus

$$\mathbf{E}(\eta(t,\theta)^2) = \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| G_{0,OE}(e^{i\omega}) - G(e^{i\omega},\theta) \right|^2 \Phi_u(e^{i\omega}) \, d\omega + A_0 + B_0$$

and (4.16) follows as  $A_0$  and  $B_0$  are independent of  $\theta$ .

Theorem 4.2 shows that a low order output error model approximation of an OE-LTI-SOE results in the same kind of approximation as a low order approximation of an LTI system. More specifically, (4.16) shows that if  $\Phi_u(e^{i\omega})$  is large in a certain frequency region, the parameter vector  $\theta$  will be chosen such that

$$|G_{0,OE}(e^{i\omega}) - G(e^{i\omega},\theta)|$$

is small in that frequency region.

However, it is important to remember that there is a major difference between the linear and the nonlinear cases. If the true system is an LTI system, it is always desirable to approximate it as well as possible, at least for some frequencies. On the other hand, if the system is nonlinear, there is no guarantee that the OE-LTI-SOE is a good model of the

system for any other input signals than the one that it was defined for. Actually, it might be a bad model also for this signal.

If, for example, a second order output error model is estimated and the input power is focused in a certain frequency region, the model will in general approximate a *different* OE-LTI-SOE than if, for example, a white input signal had been used. Actually, input signals with equal  $\Phi_u(z)$  but different distributions will in general give rise to different OE-LTI-SOEs of a nonlinear system. A simple example of this will be shown later in Section 5.4. These observations make it much harder to design the input such that it is suitable for low order LTI approximations when the system is nonlinear.

# 4.3 The Output Error Model Type for Periodic Inputs

For nonlinear systems with periodic inputs satisfying Assumption A3, OE-LTI-SOEs can be defined using the following alternative definition.

**Definition 4.3.** Consider a nonlinear system with a *P*-periodic stochastic input u(t) and an output y(t) such that Assumption A3 is fulfilled. An OE-LTI-SOE of this system is a stable and causal LTI model  $G_{0,OE}(q)$  that minimizes the mean-square error  $E((y(t) - G(q)u(t))^2)$ , i.e.,

$$G_{0,OE}(q) = \operatorname*{arg\,min}_{G \in \mathcal{G}} \mathrm{E}((y(t) - G(q)u(t))^2),$$

where  $\mathcal{G}$  denotes the set of all stable and causal LTI models. Let  $\mathcal{G}_{0,OE}$  denote the set of all OE-LTI-SOEs for this particular pair of input and output signals, i.e.,

$$\mathcal{G}_{0,OE} = \{G_{0,OE}(q)\}.$$

Note that  $\mathcal{G}_{0,OE}$  always will contain more than one model. For example, consider a system with an OE-LTI-SOE  $G_{0,OE,0}(q)$  for a particular *P*-periodic input u(t). Then the models  $G_{0,OE,k}(q) = (1 + q^{-kP})G_{0,OE,0}(q)/2$ ,  $k \in \mathbb{N}$ , are OE-LTI-SOEs too, since they will produce the same stationary output as  $G_{0,OE,0}(q)$ . Obviously, the impulse responses from these OE-LTI-SOEs are quite different, but this does not matter here since the transient response of a model is not considered in the definition of the OE-LTI-SOE.

The fact that the OE-LTI-SOE of a nonlinear system with a periodic input is not unique does not imply that there are not any uniquely determined features in these models. These unique features are most easy to describe in the frequency domain. Hence, we will now rewrite the mean-square error in the frequency domain in a similar way as the cost function was rewritten in Section 2.3.2. Consider a positive integer M and a P-periodic input signal and an output such that Assumption A3 is satisfied. Applying the Discrete Fourier Transform (DFT) to the input and output signal gives

$$U_N(n) = \sum_{t=0}^{N-1} u(t)e^{-i2\pi nt/N},$$
(4.17a)

$$Y_N(n) = \sum_{t=0}^{N-1} y(t) e^{-i2\pi n t/N},$$
(4.17b)

where N = MP. Note that we here consider stochastic processes, i.e.,  $U_N(n)$  and  $Y_N(n)$  are random variables. Let  $\hat{y}_G(t)$  denote the output from the stable model G(q) for the input u(t) and assume that the input has been applied at  $t = -\infty$  such that all transients are gone at  $t \ge 0$ , i.e., that  $\hat{y}_G(t)$  is *P*-periodic in the interval  $0 \le t \le N - 1$ . Furthermore, let  $\hat{Y}_{G,N}(n)$  denote the DFT of  $\hat{y}_G(t)$ , i.e.,

$$\hat{Y}_{G,N}(n) = \sum_{t=0}^{N-1} \hat{y}_G(t) e^{-i2\pi nt/N}.$$

Using the same reasoning as in (2.10), we obtain that

$$G(e^{i2\pi n/N}) = \tilde{G}_N(n), \qquad (4.18)$$

where  $\tilde{G}_N(n)$  is the DFT of

$$\tilde{g}_N(t) = \sum_{l=0}^{\infty} g(t+lN), \quad 0 \le t \le N-1.$$

Since u(t) is a *P*-periodic signal and N = MP, with  $M \in \mathbb{Z}_+$ , u(t) is also *N*-periodic. Hence,

$$\hat{y}_G(t) = G(q)u(t) = \sum_{k=0}^{N-1} \tilde{g}(k)u(t-k)$$

and this implies that

$$\hat{Y}_{G,N}(n) = \tilde{G}_N(n)U_N(n) = G(e^{i2\pi n/N})U_N(n),$$
(4.19)

where we have used (4.18) in the last equality.

Since both the input and the output are stationary stochastic processes, the meansquare error can be rewritten as

$$E((y(t) - G(q)u(t))^{2}) = E(\frac{1}{N} \sum_{t=0}^{N-1} (y(t) - G(q)u(t))^{2})$$
  
=  $E(\frac{1}{N^{2}} \sum_{n=0}^{N-1} |Y_{N}(n) - \hat{Y}_{G,N}(n)|^{2})$   
=  $E(\frac{1}{N^{2}} \sum_{n=0}^{N-1} |Y_{N}(n) - G(e^{i2\pi n/N})U_{N}(n)|^{2}),$  (4.20)

where we have used Parseval's formula in the second equality and (4.19) in the third. Using this expression for the mean-square error, it is easy to show how the frequency response of the OE-LTI-SOE can be calculated.

#### Theorem 4.3 (OE-LTI-SOEs for Periodic Inputs)

Consider a nonlinear system with a P-periodic input u(t) and output y(t) such that Assumption A3 is fulfilled and a positive integer M. Then an OE-LTI-SOE  $G_{0,OE}$  of this system has a frequency response with the property

$$G_{0,OE}(e^{i2\pi n/N}) = \frac{\mathrm{E}(Y_N(n)U_N(n))}{\mathrm{E}(|U_N(n)|^2)}, \quad n \in \Omega,$$
(4.21)

where  $\Omega \subset \{0, 1, ..., N-1\} \triangleq \mathbb{Z}_N$  is the set of integers for which  $E(|U_N(n)|^2) \neq 0$ and where N = MP.

**Proof:** Schwarz inequality for random variables (Råde and Westergren, 1995, p. 407) can be used to show that  $E(|U_N(n)|^2) = 0$  implies that

$$E(Y_N(n)\overline{U_N(n)}) = E(U_N(n)\overline{Y_N(n)}) = 0.$$

Let  $\Omega_c$  denote the set of integers in  $\mathbb{Z}_N$  that does not belong to  $\Omega$ , i.e., the complement of  $\Omega$  in  $\mathbb{Z}_N$ . The mean-square error (4.20) can be rewritten as

$$E((y(t) - G(q)u(t))^{2}) = E(\frac{1}{N^{2}} \sum_{n=0}^{N-1} \left| Y_{N}(n) - G(e^{i2\pi n/N})U_{N}(n) \right|^{2})$$
  
=  $E(\frac{1}{N^{2}} \sum_{n \in \Omega} \left| Y_{N}(n) - G(e^{i2\pi n/N})U_{N}(n) \right|^{2}) + E(\frac{1}{N^{2}} \sum_{n \in \Omega_{c}} |Y_{N}(n)|^{2}).$ 

The condition that  $G_{0,OE}$  should minimize this criterion is equivalent to that  $G_{0,OE}$  should satisfy the Wiener-Hopf condition

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} g_{0,OE}(k) R_u(\tau - k) = 0, \quad \tau \ge 0$$
(4.22)

or, alternatively, that

$$E(Y_N(n)\overline{U_N(n)}) - G_{0,OE}(e^{i2\pi n/N})E(|U_N(n)|^2) = 0, \quad n \in \Omega.$$
(4.23)

This equivalence follows from a similar orthogonality argument as the Wiener-Hopf condition followed from in the proof of Theorem 4.1. The expression for the OE-LTI-SOE in (4.21) follows readily from (4.23).

Note that since  $R_{yu}(\tau)$  and  $R_u(\tau)$  are *P*-periodic, the Wiener-Hopf criterion (4.22) is satisfied if

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} g_{0,OE}(k) R_u(\tau - k) = 0, \quad 0 \le \tau \le P - 1.$$

Actually, the periodicity implies that (4.22) is equivalent to the criterion

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} g_{0,OE}(k) R_u(\tau - k) = 0, \quad \forall \tau \in \mathbb{Z}.$$
 (4.24)

Theorem 4.3 shows that the frequency response of an OE-LTI-SOE of a nonlinear system with a periodic input signal is uniquely determined only at the excited frequencies  $\omega_n = 2\pi n/N$ ,  $n \in \Omega$ . The frequency response of the OE-LTI-SOE is arbitrary at all other frequencies. Of course, this is not a surprising observation but merely a version of a classic result from system identification literature (Ljung, 1999, Chap. 8).

However, Theorem 4.3 gives a theoretical motivation to why a nonparametric frequency response estimate can be particularly useful when the input is periodic. For example, if  $N_E$  experiments have been performed with different realizations of the input signal, the nonparametric frequency response estimate

$$\hat{G}_{np}(n) = \frac{\sum_{s=1}^{N_E} Y_{s,N}(n) \overline{U_{s,N}(n)}}{\sum_{s=1}^{N_E} |U_{s,N}(n)|^2} = \frac{\frac{1}{N_E} \sum_{s=1}^{N_E} Y_{s,N}(n) \overline{U_{s,N}(n)}}{\frac{1}{N_E} \sum_{s=1}^{N_E} |U_{s,N}(n)|^2}$$
(4.25)

from (2.12) can be calculated at the excited frequencies. If  $N_E$  is large, this estimate will typically be a good approximation of the expression for the OE-LTI-SOE in (4.21). Hence, by calculating a nonparametric frequency response estimate based on data from several experiments, approximations of the properties that are common to all OE-LTI-SOEs of the system can be obtained.

In the previous section, it was shown that the model residuals for the OE-LTI-SOE are uncorrelated with past and current inputs but in general not with future input components. However, for a periodic input, the residuals will actually be uncorrelated with all input components. This observation is summarized in the following corollary.

#### Corollary 4.3

Consider a nonlinear system with a P-periodic input u(t) and output y(t) such that Assumption A3 is fulfilled, an OE-LTI-SOE  $G_{0,OE}$  of this system and a positive integer M. Let the residuals be defined by

$$\eta_0(t) = y(t) - G_{0,OE}(q)u(t) \tag{4.26}$$

Then

$$E(H_{0,N}(n)\overline{U_N(n)}) = E(Y_N(n)\overline{U_N(n)}) - G_{0,OE}(e^{i2\pi n/N})E(|U_N(n)|^2)$$
  
= 0,  $\forall n \in \Omega$ , (4.27a)

$$R_{\eta_0 u}(\tau) = R_{yu}(\tau) - G_{0,OE}(q)R_u(\tau) = 0, \qquad (4.27b)$$

where  $H_{0,N}(n)$  is the DFT of  $\eta_0$ , N = MP and  $\Omega$  defines the excited frequencies. Furthermore, the variance of the residuals can be calculated as

$$R_{\eta_0}(0) = R_y(0) - \frac{1}{N^2} \sum_{n=0}^{N-1} |G_{0,OE}(e^{i2\pi n/N})|^2 \mathbf{E}(|U_N(n)|^2).$$
(4.28)

**Proof:** The results (4.27a) and (4.27b) follow directly from (4.23) and (4.24), respectively. The expression for the variance of  $\eta_0$  can be shown using (4.21). Using Parseval's identity, we get

$$R_{\eta_0}(0) = \mathbf{E}\left(\frac{1}{N}\sum_{t=0}^{N-1} (y(t) - G_{0,OE}(q)u(t))^2\right)$$
$$= \mathbf{E}\left(\frac{1}{N^2}\sum_{n=0}^{N-1} |Y_N(n) - G_{0,OE}(e^{i2\pi n/N})U_N(n)|^2\right)$$

$$= \mathbf{E}\left(\frac{1}{N^2}\sum_{n=0}^{N-1} (|Y_N(n)|^2 - Y_N(n)\overline{U_N(n)G_{0,OE}(e^{i2\pi n/N})} - G_{0,OE}(e^{i2\pi n/N})U_N(n)\overline{Y_N(n)} + |G_{0,OE}(e^{i2\pi n/N})U_N(n)|^2)\right)$$
$$= \mathbf{E}\left(\frac{1}{N^2}\sum_{n=0}^{N-1} |Y_N(n)|^2\right) - \frac{1}{N^2}\sum_{n=0}^{N-1} |G_{0,OE}(e^{i2\pi n/N})|^2 \mathbf{E}(|U_N(n)|^2),$$

where we have used (4.21) and the fact that  $E(|U_N(n)|^2) = 0$  implies that

$$E(Y_N(n)\overline{U_N(n)}) = E(U_N(n)\overline{Y_N(n)}) = 0$$

in the last equality. Since

$$R_y(0) = \mathcal{E}\left(\frac{1}{N}\sum_{t=0}^{N-1} y(t)^2\right) = \mathcal{E}\left(\frac{1}{N^2}\sum_{n=0}^{N-1} |Y_N(n)|^2\right),$$

(4.28) has been shown.

Although the results (4.27) in the previous corollary are rather obvious, they show an important difference between OE-LTI-SOEs for nonperiodic inputs and OE-LTI-SOEs for periodic inputs. However, it will be shown later that many nonperiodic signals also give OE-LTI-SOEs with residual signals that are uncorrelated with the input signal.

If a parametric model is calculated by minimizing the variance of the residuals, it will approximate the frequency response of the OE-LTI-SOE according to the following theorem.

#### Theorem 4.4

Consider a nonlinear system with a P-periodic input u(t) and output y(t) such that Assumption A3 is fulfilled, an OE-LTI-SOE  $G_{0,OE}$  of this system and a positive integer M. Suppose that a parameterized stable and causal output error model  $G(q, \theta)$  is fitted to the signals u and y according to

$$\hat{\theta} = \operatorname*{arg\,min}_{\theta} \mathrm{E}(\eta(t,\theta)^2), \tag{4.29}$$

where

$$\eta(t,\theta) = y(t) - G(q,\theta)u(t) \tag{4.30}$$

Then it follows that

$$\hat{\theta} = \arg\min_{\theta} \sum_{n=0}^{N-1} |G_{0,OE}(e^{i2\pi n/N}) - G(e^{i2\pi n/N}, \theta)|^2 \mathbf{E}(|U_N(n)|^2).$$
(4.31)

Proof: The variance of the model residual can be written

$$\begin{split} \mathbf{E}(\eta(t,\theta)^2) &= \mathbf{E}(\frac{1}{N}\sum_{t=0}^{N-1}(y(t) - G(q,\theta)u(t))^2) \\ &= \mathbf{E}(\frac{1}{N^2}\sum_{n=0}^{N-1}|Y_N(n) - G(e^{i2\pi n/N},\theta)U_N(n)|^2) \\ &= \mathbf{E}(\frac{1}{N^2}\sum_{n=0}^{N-1}|H_{0,N}(n) + (G_{0,OE}(e^{i2\pi n/N}) - G(e^{i2\pi n/N},\theta))U_N(n)|^2) \\ &= \frac{1}{N^2}\sum_{n=0}^{N-1}\mathbf{E}(|H_{0,N}(n)|^2) \\ &+ \frac{1}{N^2}\sum_{n=0}^{N-1}|G_{0,OE}(e^{i2\pi n/N}) - G(e^{i2\pi n/N},\theta)|^2\mathbf{E}(|U_N(n)|^2), \end{split}$$

where  $H_{0,N}(n)$  is the DFT of  $\eta_0(t)$  in (4.26) and where we have used (4.27a) in the last equality. The result (4.31) follows since

$$\frac{1}{N^2} \sum_{n=0}^{N-1} \mathcal{E}(|H_{0,N}(n)|^2)$$

is independent of  $\theta$ .

Theorem 4.4 shows that if a parametric model is calculated by minimizing the variance of the model residuals, this model will try to approximate the frequency response of the OE-LTI-SOE at the excited frequencies. Note that there is no guarantee that  $\hat{\theta}$  will be unique. If the order of the parametric model is higher than the lowest order OE-LTI-SOE, there will be many optimal values for  $\hat{\theta}$ . However, in this sense, there is no difference between the results for nonperiodic and periodic inputs. No matter the type of input, if the order of the parametric model is higher than for the OE-LTI-SOE, the optimal parametric model will not be unique.

More properties of OE-LTI-SOEs will be presented in Chapter 5. First, we will turn our attention to LTI-SOEs that contain a noise model.

## 4.4 The General Error Model Type

Consider once again the general LTI model (2.1) from Section 2.1. For this model, the optimal predictor can be written as in (2.2). Predictors with this structure are used in the prediction-error method to compute parameter dependent predictions of the output signal of the system (cf. Section 2.3). The predicted output is compared with the measured output and the parameters are selected such that a quadratic criterion is minimized.

With this in mind, it is natural to define the best general LTI model in the mean-square error sense as the LTI model whose predictor (2.2) minimizes the mean-square error. The optimal LTI model according to this definition will here be called the *General Error LTI* 

 $\square$ 

Second Order Equivalent (GE-LTI-SOE). It is described more clearly in the following definition.

**Definition 4.4.** Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A4 are fulfilled. The General Error LTI Second Order Equivalent (GE-LTI-SOE) of this system is the pair of transfer operators  $(G_{0,GE}(q), H_{0,GE}(q))$  that are formed as

$$G_{0,GE}(q) = (1 - q^{-1} \tilde{W}_y(q))^{-1} \tilde{W}_u(q),$$
  

$$H_{0,GE}(q) = (1 - q^{-1} \tilde{W}_y(q))^{-1}.$$

Here,  $\tilde{W}_u(q)$  and  $\tilde{W}_y(q)$  are the stable and causal LTI filters that minimize the meansquare error  $E((y(t) - W_u(q)u(t) - W_y(q)y(t-1))^2)$ , i.e.,

$$(\tilde{W}_u(q), \ \tilde{W}_y(q)) = \underset{W_u, W_y \in \mathcal{G}}{\operatorname{arg min}} \operatorname{E}((y(t) - W_u(q)u(t) - W_y(q)y(t-1))^2),$$

where  $\mathcal{G}$  denotes the set of all stable and causal LTI models.

GE-LTI-SOEs are, just like OE-LTI-SOEs, input-dependent and hence it is only possible to talk about the GE-LTI-SOE of a nonlinear system with respect to a particular input signal. Another similarity between GE-LTI-SOEs and OE-LTI-SOEs is that  $\tilde{W}_u(q)$  and  $\tilde{W}_u(q)$  are assumed to be initialized at  $t = -\infty$ .

In the next theorem, expressions for the GE-LTI-SOE similar to the ones in Ljung (2001) will be derived. The only differences between these two versions of GE-LTI-SOE expressions are that the GE-LTI-SOE here is allowed to contain a direct term from the input and that it is expressed explicitly using components of the canonical spectral factor T(z) from Assumption A4. The proof of the theorem is very similar to the proof of Theorem 4.1 and hence to the proof of the scalar Wiener filter in Kailath et al. (2000).

#### Theorem 4.5 (GE-LTI-SOEs)

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A4 are fulfilled. Then the GE-LTI-SOE  $(G_{0,GE}(z), H_{0,GE}(z))$  of this system is

$$G_{0,GE}(z) = \frac{zT_{21}(z)}{T_{11}(z)},$$
(4.32a)

$$H_{0,GE}(z) = \frac{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)}{T_{11}(z)},$$
(4.32b)

where  $T_{11}(z)$ ,  $T_{12}(z)$ ,  $T_{21}(z)$  and  $T_{22}(z)$  are elements of the canonical spectral factor of the z-spectrum for  $\zeta(t) = (u(t), y(t-1))^T$ , i.e.,

$$\Phi_{\zeta}(z) = \begin{pmatrix} \Phi_u(z) & z\Phi_{uy}(z) \\ z^{-1}\Phi_{yu}(z) & \Phi_y(z) \end{pmatrix} = T(z)Q_{\zeta}T^T(z^{-1}),$$
(4.33a)

$$T(z) = \begin{pmatrix} T_{11}(z) & T_{12}(z) \\ T_{21}(z) & T_{22}(z) \end{pmatrix}.$$
(4.33b)

**Proof:** The GE-LTI-SOE is defined by means of the Wiener filter  $(\tilde{W}_u(q), \tilde{W}_y(q))$  that predicts y(t) from  $(y(t-k))_{k=1}^{\infty}$  and  $(u(t-k))_{k=0}^{\infty}$  using

$$\hat{y}(t) = \tilde{W}_u(q)u(t) + \tilde{W}_y(q)y(t-1).$$
(4.34)

The filter components  $\tilde{W}_u(q)$  and  $\tilde{W}_y(q)$  are defined as the stable and causal LTI filters that minimize  $E((y(t) - W_u(q)u(t) - W_y(q)y(t-1))^2)$  or, equivalently, the filters that satisfy the Wiener-Hopf conditions

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} \tilde{w}_y(k) R_{yu}(\tau - k - 1) - \sum_{k=0}^{\infty} \tilde{w}_u(k) R_u(\tau - k) = 0, \quad \tau \ge 0,$$
(4.35a)

$$R_y(\tau+1) - \sum_{k=0}^{\infty} \tilde{w}_y(k) R_y(\tau-k) - \sum_{k=0}^{\infty} \tilde{w}_u(k) R_{uy}(\tau+1-k) = 0, \quad \tau \ge 0.$$
(4.35b)

Using the z-transform, these conditions can be rewritten as

$$\Phi_{yu}(z) - \tilde{W}_y(z)z^{-1}\Phi_{yu}(z) - \tilde{W}_u(z)\Phi_u(z) = K_1(z),$$
(4.36a)

$$z\Phi_y(z) - \dot{W}_y(z)\Phi_y(z) - \dot{W}_u(z)z\Phi_{uy}(z) = K_2(z),$$
(4.36b)

where  $K_1(z)$  and  $K_2(z)$  are stable and strictly anticausal transfer functions. Equations (4.36) and (4.2b) give

$$\begin{pmatrix} K_1(z) & K_2(z) \end{pmatrix} = \underbrace{\left( \Phi_{yu}(z) & z \Phi_y(z) \right)}_{=(0 \ z) \Phi_\zeta(z)} - \begin{pmatrix} \tilde{W}_u(z) & \tilde{W}_y(z) \end{pmatrix} \Phi_\zeta(z).$$
(4.37)

Using the spectral factorization of  $\Phi_{\zeta}(z)$  and multiplying by  $T^{-T}(z^{-1})$  now gives

$$\tilde{K}(z) \triangleq \begin{pmatrix} K_1(z) & K_2(z) \end{pmatrix} T^{-T}(z^{-1}) = \begin{pmatrix} 0 & z \end{pmatrix} T(z)Q_{\zeta} - \begin{pmatrix} \tilde{W}_u(z) & \tilde{W}_y(z) \end{pmatrix} T(z)Q_{\zeta},$$
(4.38)

where K(z) is a stable and strictly anticausal transfer matrix, due to the fact that it is a product of the stable and strictly anticausal transfer matrix  $(K_1(z) K_2(z))$  and the stable and anticausal transfer matrix  $T^{-T}(z^{-1})$ . For (4.38) to hold it is necessary that the second right hand term, which is a stable and causal transfer matrix, is equal to the causal part of the first, i.e.,

$$\begin{split} & \left(\tilde{W}_{u}(z) \quad \tilde{W}_{y}(z)\right) T(z)Q_{\zeta} = \left[ \begin{pmatrix} 0 & z \end{pmatrix} T(z)Q_{\zeta} \right]_{\text{causal}} \\ &= \left[ \left( zT_{21}(z)Q_{\zeta 11} + zT_{22}(z)Q_{\zeta 21} \quad zT_{21}(z)Q_{\zeta 12} + zT_{22}(z)Q_{\zeta 22} \right) \right]_{\text{causal}} \\ &= \left( zT_{21}(z)Q_{\zeta 11} + z(T_{22}(z) - 1)Q_{\zeta 21} \quad zT_{21}(z)Q_{\zeta 12} + z(T_{22}(z) - 1)Q_{\zeta 22} \right) \\ &= \left( \begin{pmatrix} 0 & z \end{pmatrix} T(z) - \begin{pmatrix} 0 & z \end{pmatrix} \right) Q_{\zeta}, \end{split}$$

where the third equality follows since  $T_{21}(z)$  is a stable and strictly causal transfer function while  $T_{22}(z)$  is a monic stable and causal transfer function. This gives

$$\begin{pmatrix} \tilde{W}_u(z) & \tilde{W}_y(z) \end{pmatrix} = \left( \begin{pmatrix} 0 & z \end{pmatrix} T(z) - \begin{pmatrix} 0 & z \end{pmatrix} \right) T^{-1}(z) = \left( \frac{zT_{21}(z)}{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)} & z \left( 1 - \frac{T_{11}(z)}{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)} \right) \right).$$
(4.39)

Let

$$\varepsilon_0(t) = y(t) - \hat{y}(t) = (1 - q^{-1} \tilde{W}_y(q)) y(t) - \tilde{W}_u(q) u(t)$$
(4.40)

and rewrite this expression in analogy with (2.1) and (2.2) as

$$y(t) = (1 - q^{-1}\tilde{W}_y(q))^{-1}\tilde{W}_u(q)u(t) + (1 - q^{-1}\tilde{W}_y(q))^{-1}\varepsilon_0(t).$$
(4.41)

With

$$G_{0,GE}(q) = (1 - q^{-1} \tilde{W}_y(q))^{-1} \tilde{W}_u(q),$$
  

$$H_{0,GE}(q) = (1 - q^{-1} \tilde{W}_y(q))^{-1},$$

(4.41) can be written as

$$y(t) = G_{0,GE}(q)u(t) + H_{0,GE}(q)\varepsilon_0(t).$$
(4.42)

Hence, using (4.39) the GE-LTI-SOE of the nonlinear system turns out to be

$$G_{0,GE}(z) = \frac{zT_{21}(z)}{T_{11}(z)},$$
  

$$H_{0,GE}(z) = \frac{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)}{T_{11}(z)}.$$

From (4.32) we see that the calculation of a GE-LTI-SOE requires knowledge of the canonical spectral factorization of the matrix  $\Phi_{\zeta}(z)$ . Although there exists a number of methods for making this factorization, it is definitely more complicated to perform than the canonical factorization of the input z-spectrum that is required for the calculation of the OE-LTI-SOE in Theorem 4.1. In Section 5.6, we will describe some classes of input signals that simplify the calculation of GE-LTI-SOEs significantly.

It should be noted that the factor  $\frac{1}{T_{11}(z)}$  in (4.32) should as usual be interpreted as a causal series expansion. This is always possible because  $T_{11}(z)$  is analytic on and outside the unit circle and  $T_{11}(\infty) = 1$ . Let  $H_{0,GE}^{-1}$  denote the transfer function of the inverse model of  $H_{0,GE}$ , i.e.,

$$H_{0,GE}^{-1}(z) = \frac{1}{H_{0,GE}(z)}.$$
(4.44)

By the construction of the GE-LTI-SOE,  $H_{0,GE}^{-1}(z)G_{0,GE}(z)$  and  $H_{0,GE}^{-1}(z)$  will be stable transfer functions. However, neither  $G_{0,GE}(z)$  nor  $H_{0,GE}(z)$  needs to be stable since  $T_{11}(z)$  might have zeros outside the unit circle. The only thing that is guaranteed is that all unstable poles to  $G_{0,GE}(z)$  are also poles to  $H_{0,GE}(z)$  and vice versa.

In the Section 4.2, it was shown that the OE-LTI-SOE of a nonlinear system with a nonperiodic input can explain all correlations between the output and past and present input signal components (see Corollary 4.2). As the inclusion of a noise model in the GE-LTI-SOE makes this model more flexible than the OE-LTI-SOE, one might expect that the GE-LTI-SOE should be able to explain more correlations than the OE-LTI-SOE. In the following theorem it will be shown that this is also the case.
#### **Corollary 4.4**

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A4 are fulfilled. Let  $(G_{0,GE}, H_{0,GE})$  be the corresponding GE-LTI-SOE according to Theorem 4.5 and let  $\varepsilon_0(t)$  be defined by (4.40). Then it holds that

$$\Phi_{\varepsilon_0 u}(z) = H_{0,GE}^{-1}(z)(\Phi_{yu}(z) - G_{0,GE}(z)\Phi_u(z))$$
  
=  $zQ_{\zeta_{21}}T_{11}(z^{-1}) + zQ_{\zeta_{22}}T_{12}(z^{-1}),$  (4.45a)

$$\Phi_{\varepsilon_0 y}(z) = H_{0,GE}^{-1}(z)(\Phi_y(z) - G_{0,GE}(z)\Phi_{uy}(z))$$
  
=  $Q_{\zeta 21}T_{21}(z^{-1}) + Q_{\zeta 22}T_{22}(z^{-1}),$  (4.45b)

$$\Phi_{\varepsilon_0}(z) = H_{0,GE}^{-1}(z)(\Phi_{y\varepsilon_0}(z) - G_{0,GE}(z)\Phi_{u\varepsilon_0}(z)) = Q_{\zeta 22} \triangleq \lambda_0.$$
(4.45c)

Furthermore, an alternative way to describe the relations between the z-spectra of u, y and  $\varepsilon_0$  by the GE-LTI-SOE is

$$\Phi_{yu}(z) = G_{0,GE}(z)\Phi_u(z) + H_{0,GE}(z)\Phi_{\varepsilon_0 u}(z),$$
(4.46a)

$$\Phi_{y\varepsilon_0}(z) = G_{0,GE}(z)\Phi_{u\varepsilon_0}(z) + H_{0,GE}(z)\lambda_0,$$
(4.46b)

$$\Phi_y(z) = \begin{pmatrix} G_{0,GE}(z) & H_{0,GE}(z) \end{pmatrix} \begin{pmatrix} \Phi_u(z) & \Phi_{u\varepsilon_0}(z) \\ \Phi_{\varepsilon_0 u}(z) & \lambda_0 \end{pmatrix} \begin{pmatrix} G_{0,GE}(z^{-1}) \\ H_{0,GE}(z^{-1}) \end{pmatrix}.$$
(4.46c)

**Proof:** Note that (4.40) can be written

$$\varepsilon_0(t) = H_{0,GE}^{-1}(q)(y(t) - G_{0,GE}(q)u(t)).$$
(4.47)

This gives

$$\Phi_{\varepsilon_0 u}(z) = H_{0,GE}^{-1}(z)(\Phi_{yu}(z) - G_{0,GE}(z)\Phi_u(z)),$$
  
$$\Phi_{\varepsilon_0 y}(z) = H_{0,GE}^{-1}(z)(\Phi_y(z) - G_{0,GE}(z)\Phi_{uy}(z)).$$

Together with (4.32) and (4.33a) these expressions can be rewritten as

$$\begin{aligned} & \left(z^{-1}\Phi_{\varepsilon_{0}u}(z) \quad \Phi_{\varepsilon_{0}y}(z)\right) \\ &= H_{0,GE}^{-1}(z) \left(z^{-1}\Phi_{yu}(z) - z^{-1}G_{0,GE}(z)\Phi_{u}(z) \quad \Phi_{y}(z) - G_{0,GE}(z)\Phi_{uy}(z)\right) \\ &= \frac{T_{11}(z)}{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)} \left(-\frac{T_{21}(z)}{T_{11}(z)} \quad 1\right) \Phi_{\zeta}(z) = \begin{pmatrix} 0 & 1 \end{pmatrix} T^{-1}(z)\Phi_{\zeta}(z) \\ &= \begin{pmatrix} 0 & 1 \end{pmatrix} Q_{\zeta}T^{T}(z^{-1}) = \begin{pmatrix} Q_{\zeta 21} & Q_{\zeta 22} \end{pmatrix} T^{T}(z^{-1}) \\ &= \begin{pmatrix} Q_{\zeta 21}T_{11}(z^{-1}) + Q_{\zeta 22}T_{12}(z^{-1}) & Q_{\zeta 21}T_{21}(z^{-1}) + Q_{\zeta 22}T_{22}(z^{-1}) \end{pmatrix}. \end{aligned}$$

Hence, (4.45a) and (4.45b) have been shown. Furthermore, (4.47) gives, using (4.32),

(4.45a) and (4.45b), the following expression

$$\begin{split} \Phi_{\varepsilon_0}(z) &= H_{0,GE}^{-1}(z) (\Phi_{y\varepsilon_0}(z) - G_{0,GE}(z) \Phi_{u\varepsilon_0}(z)) \\ &= \frac{T_{11}(z)}{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)} (Q_{\zeta 21}T_{21}(z) + Q_{\zeta 22}T_{22}(z) \\ &- \frac{zT_{21}(z)}{T_{11}(z)} z^{-1} \left(Q_{\zeta 21}T_{11}(z) + Q_{\zeta 22}T_{12}(z)\right) \right) \\ &= \frac{T_{11}(z)}{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)} \left(Q_{\zeta 22}T_{22}(z) - Q_{\zeta 22}\frac{T_{12}(z)T_{21}(z)}{T_{11}(z)}\right) \\ &= Q_{\zeta 22} \end{split}$$

and (4.45c) follows.

The expressions (4.45a) and (4.45c) can be rewritten as (4.46a) and (4.46b), respectively. Finally, (4.46c) follows if (4.46a) and (4.46b) are inserted in (4.45b).  $\Box$ 

The main result of Corollary 4.4 is that  $\Phi_{\varepsilon_0 u}(z)$  and  $\Phi_{\varepsilon_0 y}(z)$  are strictly anticausal and anticausal, respectively, and that  $\Phi_{\varepsilon_0}(z)$  is a constant. Hence, (4.45) illustrates that the GE-LTI-SOE really is the best possible LTI model of the nonlinear system as the remaining residuals  $\varepsilon_0(t)$  are uncorrelated with past outputs, past and present inputs and with residuals at all other time instants.

In addition, Corollary 4.4 explains why the name *LTI Second Order Equivalent* is natural. The alternative version of (4.45) in (4.46) emphasizes the filtering capabilities of  $G_{0,GE}$  and  $H_{0,GE}$ . As a matter of fact, (4.46) shows that the GE-LTI-SOE is impossible to distinguish from the true nonlinear system only by looking at second order properties of y, u and  $\varepsilon_0$ . The LTI system  $(G_{0,GE}(q), H_{0,GE}(q))$  is thus equivalent to the nonlinear system for the input in question if only second order properties are considered, hence the name GE-LTI-SOE. The additional *General Error* in the name GE-LTI-SOE is added in order to distinguish this type of LTI model from the previously described output error model type, which does not include a noise model.

A fundamental observation about LTI-SOEs is that the OE-LTI-SOE  $G_{0,OE}$  and the GE-LTI-SOE  $G_{0,GE}$  are not always equal. This is shown in Example 4.3. The particular system and input signal used in this example have been taken from Forssell and Ljung (2000), where it is used to show that  $\Phi_{yu}(z)/\Phi_u(z)$  can be noncausal. Here, however, we will also derive the GE-LTI-SOE of this system.

— Example 4.3 -

Consider the static nonlinear system

$$y(t) = u(t)^2 - 3 \tag{4.48}$$

with the input

$$u(t) = e(t) + e(t-1)^2 - 1$$

where e(t) here is a white Gaussian process with zero mean and unit variance. Straightforward calculations (see Appendix A), which are similar to the ones in Example 4.2,

give

$$\begin{split} \Phi_u(z) &= 3, \\ \Phi_{yu}(z) &= 2z + 8, \\ \Phi_y(z) &= 8z + 66 + 8z^{-1}. \end{split}$$

This gives

$$\begin{split} \Phi_{\zeta}(z) &= \begin{pmatrix} 3 & 2+8z \\ 2+8z^{-1} & 8z+66+8z^{-1} \end{pmatrix} = T(z)Q_{\zeta}T^{T}(z^{-1}), \\ T(z) &= \begin{pmatrix} 1 & 0 \\ \frac{\sqrt{4161-33}}{12}z^{-1} & 1+\frac{65-\sqrt{4161}}{8}z^{-1} \end{pmatrix}, \\ Q_{\zeta} &= \begin{pmatrix} 3 & 2 \\ 2 & 23+\frac{\sqrt{4161}}{3} \end{pmatrix}. \end{split}$$

The spectral factor T(z) has been computed by a diagonalization of  $\Phi_{\zeta}(z)$  followed by a factorization of the derived diagonal matrix and an adjustment in order to achieve  $T(\infty) = I$ . The complete derivation of T(z) can be found in Appendix A. From the derived transform expressions, the OE-LTI-SOE of the system (4.48) for this input is found to be

$$G_{0,OE}(z) = \frac{8}{3} \approx 2.6667,$$

while the GE-LTI-SOE is

$$G_{0,GE}(z) = \frac{\sqrt{4161} - 33}{12} \approx 2.6255,$$
  
$$H_{0,GE}(z) = 1 + \frac{65 - \sqrt{4161}}{8} z^{-1}.$$

Note that  $G_{0,OE}(z) \neq G_{0,GE}(z)$  despite the fact that the system operates in open loop.

The fact that  $G_{0,OE}(z) \neq G_{0,GE}(z)$  for some open-loop nonlinear systems and inputs is the reason why a matrix-valued spectral factorization in general has to be performed when the GE-LTI-SOE is calculated. If  $G_{0,OE}(z)$  and  $G_{0,GE}(z)$  always would have been equal, the GE-LTI-SOE could have been calculated using only scalar spectral factorizations. In Section 5.6 it will be shown that such a simplified calculation of the GE-LTI-SOE is possible for some classes of input signals.

Just like in the case of OE-LTI-SOEs, it is in practice often hard to know the correct order of the GE-LTI-SOE. As a matter of fact, it might actually be infinite dimensional. Hence, it is also here interesting to understand in what sense a low order model can approximate the GE-LTI-SOE of a nonlinear system. Also for GE-LTI-SOEs, this approximation is similar to a low order approximation of an LTI system, something which is shown in the next theorem. Apart from the fact that a GE-LTI-SOE is here allowed to contain a direct term from the input, this theorem is identical to Theorem 4.1 in Ljung (2001). The proof is, just like for Theorem 4.2, similar to the outlined proof in Problem 8G.5 in Ljung (1999).

#### Theorem 4.6

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A4 are fulfilled. Let  $(G_{0,GE}, H_{0,GE})$  be the corresponding GE-LTI-SOE according to Theorem 4.5, let  $\varepsilon_0(t)$  be defined by (4.40) and assume that both  $G_{0,GE}$  and  $H_{0,GE}$ are stable. Consider a parameterized stable and causal model  $(G(q, \theta), H(q, \theta))$ , where  $H(q, \theta)$  is monic and where  $H^{-1}(q, \theta)$  is stable and causal. Suppose that this model is fit to the signals u and y according to

$$\hat{\theta} = \operatorname*{arg\,min}_{\theta} \mathrm{E}(\varepsilon(t,\theta)^2), \tag{4.49}$$

where

$$\varepsilon(t,\theta) = y(t) - H^{-1}(q,\theta)G(q,\theta)u(t) - (1 - H^{-1}(q,\theta))y(t)$$
  
=  $H^{-1}(q,\theta)(y(t) - G(q,\theta)u(t)).$  (4.50)

Then it follows that

$$\hat{\theta} = \arg\min_{\theta} \int_{-\pi}^{\pi} \Delta_{GH} (e^{i\omega}, \theta)^T \begin{pmatrix} \Phi_u(e^{i\omega}) & \Phi_{u\varepsilon_0}(e^{i\omega}) \\ \Phi_{\varepsilon_0 u}(e^{i\omega}) & \lambda_0 \end{pmatrix} \Delta_{GH} (e^{-i\omega}, \theta) \, d\omega, \quad (4.51)$$

where

$$\Delta_{GH}(z,\theta) = \frac{1}{H(z,\theta)} \begin{pmatrix} G_{0,GE}(z) - G(z,\theta) \\ H_{0,GE}(z) - H(z,\theta) \end{pmatrix}.$$
(4.52)

**Proof:** Equation (4.42) can be used to rewrite (4.50) as

$$\varepsilon(t,\theta) = H^{-1}(q,\theta)(y(t) - G(q,\theta)u(t))$$
  
=  $H^{-1}(q,\theta)(G_{0,GE}(q)u(t) + H_{0,GE}(q)\varepsilon_0(t) - G(q,\theta)u(t))$   
=  $H^{-1}(q,\theta)((G_{0,GE}(q) - G(q,\theta))u(t) + (H_{0,GE}(q) - H(q,\theta))\varepsilon_0(t))$   
+  $\varepsilon_0(t) = (\Delta_{GH}(q,\theta)^T + (0 \quad 1)) \begin{pmatrix} u(t) \\ \varepsilon_0(t) \end{pmatrix}.$ 

This gives

$$\begin{split} \Phi_{\varepsilon}(z,\theta) &= \left(\Delta_{GH}(z,\theta)^{T} + \begin{pmatrix} 0 & 1 \end{pmatrix}\right) \begin{pmatrix} \Phi_{u}(z) & \Phi_{u\varepsilon_{0}}(z) \\ \Phi_{\varepsilon_{0}u}(z) & \lambda_{0} \end{pmatrix} \left(\Delta_{GH}(z^{-1},\theta) + \begin{pmatrix} 0 \\ 1 \end{pmatrix}\right) \\ &= \Delta_{GH}(z,\theta)^{T} \begin{pmatrix} \Phi_{u}(z) & \Phi_{u\varepsilon_{0}}(z) \\ \Phi_{\varepsilon_{0}u}(z) & \lambda_{0} \end{pmatrix} \Delta_{GH}(z^{-1},\theta) \\ &+ \Delta_{GH}(z,\theta)^{T} \begin{pmatrix} \Phi_{u\varepsilon_{0}}(z) \\ \lambda_{0} \end{pmatrix} + \left(\Phi_{\varepsilon_{0}u}(z) & \lambda_{0}\right) \Delta_{GH}(z^{-1},\theta) + \lambda_{0}. \end{split}$$

Parseval's relation gives

$$\begin{split} \mathbf{E}(\varepsilon(t,\theta)^2) &= \frac{1}{2\pi} \int_{-\pi}^{\pi} \Phi_{\varepsilon}(e^{i\omega},\theta) d\omega \\ &= \frac{1}{2\pi} \int_{-\pi}^{\pi} \Delta_{GH}(e^{i\omega},\theta)^T \begin{pmatrix} \Phi_u(e^{i\omega}) & \Phi_{u\varepsilon_0}(e^{i\omega}) \\ \Phi_{\varepsilon_0u}(e^{i\omega}) & \lambda_0 \end{pmatrix} \Delta_{GH}(e^{-i\omega},\theta) d\omega \\ &+ \frac{1}{2\pi} \int_{-\pi}^{\pi} \Delta_{GH}(e^{i\omega},\theta)^T \begin{pmatrix} \Phi_{u\varepsilon_0}(e^{i\omega}) \\ \lambda_0 \end{pmatrix} d\omega \\ &+ \frac{1}{2\pi} \int_{-\pi}^{\pi} \left( \Phi_{\varepsilon_0u}(e^{i\omega}) & \lambda_0 \right) \Delta_{GH}(e^{-i\omega},\theta) d\omega + \lambda_0. \end{split}$$

The transfer function  $H^{-1}(z,\theta)(G_{0,GE}(z)-G(z,\theta))$  is causal and, since both  $H_{0,GE}(z)$  and  $H(z,\theta)$  are monic, the transfer function

$$H^{-1}(z,\theta)(H_{0,GE}(z) - H(z,\theta))$$

is strictly causal. Since  $\Phi_{u\varepsilon_0}(z)$  is strictly causal (cf. (4.45a)), this implies that a term-byterm integration gives

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \Delta_{GH}(e^{i\omega}, \theta)^T \begin{pmatrix} \Phi_{u\varepsilon_0}(e^{i\omega}) \\ \lambda_0 \end{pmatrix} d\omega = 0,$$
$$\frac{1}{2\pi} \int_{-\pi}^{\pi} \left( \Phi_{\varepsilon_0 u}(e^{i\omega}) \quad \lambda_0 \right) \Delta_{GH}(e^{-i\omega}, \theta) d\omega = 0.$$

Hence,

$$\mathbf{E}(\varepsilon(t,\theta)^2) = \frac{1}{2\pi} \int_{-\pi}^{\pi} \Delta_{GH}(e^{i\omega},\theta)^T \begin{pmatrix} \Phi_u(e^{i\omega}) & \Phi_{u\varepsilon_0}(e^{i\omega}) \\ \Phi_{\varepsilon_0 u}(e^{i\omega}) & \lambda_0 \end{pmatrix} \Delta_{GH}(e^{-i\omega},\theta) \, d\omega + \lambda_0$$

and (4.51) follows since  $\lambda_0$  is independent of  $\theta$ .

The previous theorem shows that the GE-LTI-SOE can be approximated by an LTI model of lower order than the GE-LTI-SOE according to (4.51). Just like for OE-LTI-SOEs, it is here important to remember that the GE-LTI-SOE is in general input-dependent. Hence, different input signals will make a parameterized LTI model close to different GE-LTI-SOEs.

Theorem 4.6 can be simplified if the true system actually is an open-loop LTI system. In this case, there is no correlation between the residuals and the input signal, i.e.,  $\Phi_{u\varepsilon_0}(z)$  is identically equal to zero (see Ljung, 1999, p. 265). However, for an open-loop nonlinear system this is not true in general.

Consider once again the system and input signal in Example 4.3. Using (4.45a) and the expressions for T(z) and  $Q_{\zeta}$  from Appendix A it is easy to verify that  $\Phi_{u\varepsilon_0}(z) = 2z^{-1}$  in this case, despite the fact that the system in Example 4.3 is an open-loop system.

Without any prior knowledge about the structure of this system and using an intuitive reasoning based on the linear case, a nonzero strictly causal  $\Phi_{u\varepsilon_0}(z)$  might have been taken as an indication that the system actually was a closed-loop system. Hence, it is a justified question whether such a closed-loop interpretation of a GE-LTI-SOE can always be made. This is the topic of the next section.

## 4.5 Interpretations of the GE-LTI-SOE

The fact that  $\Phi_{u\varepsilon_0}(z)$  in general is nonzero in (4.46) has some implications on the LTI interpretation. If we would like to interpret all second order correlations between u, y and  $\varepsilon_0$  as results of linear filter connections we have to allow the overall linear system to include some kind of feedback or feedforward connections. This is natural as we have not imposed any restrictions on the true system that exclude the possibility that it in fact is a linear closed-loop system.

It turns out that when the GE-LTI-SOE is well-defined according to Theorem 4.5 it is always possible to compose a stable linear closed-loop system that fulfills (4.46) and that also explains the correlation between u(t) and past  $\varepsilon_0(t-k)$ , i.e., it explains why  $\Phi_{u\varepsilon_0}(z) \neq 0$ . Consider the closed-loop system in Figure 4.2. This system is described by

$$u(t) = r_{0}(t) - F_{0}(q)N_{0}(q)\tilde{H}_{0}(q)\varepsilon_{0}(t) - F_{0}(q)N_{0}(q)\tilde{G}_{0}(q)u(t)$$

$$\Rightarrow u(t) = \frac{1}{1 + F_{0}(q)N_{0}(q)\tilde{G}_{0}(q)}r_{0}(t) + \frac{-F_{0}(q)N_{0}(q)\tilde{H}_{0}(q)}{1 + F_{0}(q)N_{0}(q)\tilde{G}_{0}(q)}\varepsilon_{0}(t), \quad (4.53a)$$

$$y(t) = N_{0}(q)\tilde{H}_{0}(q)\varepsilon_{0}(t) + N_{0}(q)\tilde{G}_{0}(q)r_{0}(t) - N_{0}(q)\tilde{G}_{0}(q)F_{0}(q)y(t)$$

$$\Rightarrow y(t) = \frac{N_{0}(q)\tilde{G}_{0}(q)}{1 + F_{0}(q)N_{0}(q)\tilde{G}_{0}(q)}r_{0}(t) + \frac{N_{0}(q)\tilde{H}_{0}(q)}{1 + F_{0}(q)N_{0}(q)\tilde{G}_{0}(q)}\varepsilon_{0}(t). \quad (4.53b)$$

Let  $N_0(z)$  be a causal, possibly unstable, LTI system such that both  $G_{0,GE}(z)$  and  $H_{0,GE}(z)$  can be factorized as

$$G_{0,GE}(z) = \tilde{G}_0(z)N_0(z), \qquad (4.54a)$$

$$H_{0,GE}(z) = \hat{H}_0(z)N_0(z), \tag{4.54b}$$

where  $\tilde{G}_0$  and  $\tilde{H}_0$  are causal and stable LTI systems. Furthermore, let

$$F_0(z) = -\frac{\Phi_{u\varepsilon_0}(z)}{\Phi_{y\varepsilon_0}(z)} \tag{4.55}$$

and, as usual, interpret the factor  $\frac{1}{\Phi_{y\varepsilon_0}(z)}$  in this expression as a causal series expansion. This is always possible because  $\Phi_{y\varepsilon_0}(z) = Q_{\zeta^{21}}T_{21}(z) + Q_{\zeta^{22}}T_{22}(z)$  is analytic on and outside the unit circle and  $\Phi_{y\varepsilon_0}(\infty) = Q_{\zeta^{22}} > 0$  (see (4.45b)). Since  $\Phi_{u\varepsilon_0}(z)$  is strictly



Figure 4.2: The GE-LTI-SOE can be interpreted as being part of a linear closed-loop system.

causal (see (4.45a)), also  $F_0(z)$  will be strictly causal. Note, however, that  $F_0(z)$  might be unstable.

The interpretation of the GE-LTI-SOE as a part of a linear closed-loop system is meaningless if this system is unstable. Hence, it is crucial to check whether the transfer functions in (4.53) are stable when the definitions (4.54) and (4.55) are used. First, (4.46b), (4.54a) and (4.55) give

$$1 + F_0(z) \underbrace{N_0(z)\tilde{G}_0(z)}_{=G_{0,GE}(z)} = \frac{\Phi_{y\varepsilon_0}(z) - G_{0,GE}(z)\Phi_{u\varepsilon_0}(z)}{\Phi_{y\varepsilon_0}(z)} = \frac{H_{0,GE}(z)\lambda_0}{\Phi_{y\varepsilon_0}(z)}.$$

Thus

$$\frac{1}{1 + F_0(z)N_0(z)\tilde{G}_0(z)} = \frac{\Phi_{y\varepsilon_0}(z)}{H_{0,GE}(z)\lambda_0}$$
(4.56)

is stable since  $H_{0,GE}^{-1}(z)$  is stable and  $\Phi_{y\varepsilon_0}(z)$  by its construction (4.45b) is analytic and hence absolutely convergent on the unit circle. Since  $H_{0,GE}^{-1}(z)G_{0,GE}(z)$  is stable as well, the stability property of  $\Phi_{y\varepsilon_0}(z)$  also imply that

$$\frac{N_0(z)\hat{G}_0(z)}{1+F_0(z)N_0(z)\tilde{G}_0(z)} = \frac{G_{0,GE}(z)\Phi_{y\varepsilon_0}(z)}{H_{0,GE}(z)\lambda_0},$$
$$\frac{N_0(z)\tilde{H}_0(z)}{1+F_0(z)N_0(z)\tilde{G}_0(z)} = \frac{\Phi_{y\varepsilon_0}(z)}{\lambda_0}$$

are stable. Finally,

$$\frac{-F_0(z)N_0(z)\hat{H}_0(z)}{1+F_0(z)N_0(z)\tilde{G}_0(z)} = \frac{\Phi_{u\varepsilon_0}(z)}{\lambda_0}$$
(4.57)

is stable since  $\Phi_{u\varepsilon_0}(z)$  by (4.45a) is analytic and hence absolutely convergent on the unit

circle. Hence, all transfer functions in (4.53) are stable, and since

$$\frac{F_0(z)}{1 + F_0(z)N_0(z)\tilde{G}_0(z)} = -\frac{\Phi_{u\varepsilon_0}(z)}{H_{0,GE}(z)\lambda_0}$$

is also stable, the closed-loop system is internally stable (Glad and Ljung, 2000, Def. 6.1).

As indicated in Figure 4.2, a reference signal  $r_0$  might have to be included as an external input signal to the closed-loop model in order to explain the part of u that does not originate from  $\varepsilon_0$ . Unlike  $G_{0,GE}$  and  $H_{0,GE}$ , which are defined such that (4.42) holds,  $F_0$  can only explain the correlation between u and  $\varepsilon_0$  and not the complete signal u. If (4.54) and (4.55) are inserted in (4.53a) and the result is used to express  $\Phi_{u\varepsilon_0}(z)$  in  $\Phi_{r_0\varepsilon_0}(z)$  and  $\lambda_0$  we get

$$\Phi_{u\varepsilon_{0}}(z) = \frac{1}{1 + F_{0}(z)N_{0}(z)\tilde{G}_{0}(z)} \Phi_{r_{0}\varepsilon_{0}}(z) + \frac{-F_{0}(z)N_{0}(z)\tilde{H}_{0}(z)}{1 + F_{0}(z)N_{0}(z)\tilde{G}_{0}(z)}\lambda_{0}$$
  

$$\Rightarrow \Phi_{u\varepsilon_{0}}(z) = \frac{\Phi_{y\varepsilon_{0}}(z)}{H_{0,GE}(z)\lambda_{0}} \Phi_{r_{0}\varepsilon_{0}}(z) + \frac{\Phi_{u\varepsilon_{0}}(z)}{\lambda_{0}}\lambda_{0}$$
  

$$\Rightarrow \Phi_{r_{0}\varepsilon_{0}}(z) = 0, \qquad (4.58)$$

where (4.56) and (4.57) have been used to rewrite the closed-loop transfer functions. Equation (4.58) shows that  $r_0$  is uncorrelated with  $\varepsilon_0$ , which is natural if we want a closed-loop interpretation.

Since the linear closed-loop system in Figure 4.2 is internally stable when  $F_0$  is defined as in (4.55), it can always be used to show how the signals u and y could have been generated from an LTI model. More specifically, it is impossible to disprove that the linear closed-loop system has not generated the signals merely by looking at second order properties of u, y and  $\varepsilon_0$ . Hence, the closed-loop model in Figure 4.2 with the definitions in (4.54) and (4.55) might be called the complete second order equivalent LTI description of the true system. Of course, since the GE-LTI-SOE is input-dependent, this description is too.

Actually, it is sometimes possible to directly judge the quality of the GE-LTI-SOE if some additional information about the system is available. For example, assume that the input signal u has been generated in such a way that there can be no correlation between u(t) and any system disturbances. If the GE-LTI-SOE of the true system for this u indeed results in a nonzero  $\Phi_{u\varepsilon_0}(z)$ , then the GE-LTI-SOE cannot be a correct description of the true system.

The interpretation of the GE-LTI-SOE can be somewhat simplified if both  $G_{0,GE}$  and  $H_{0,GE}$  turn out to be stable. If this is the case,  $N_0$  can be set to 1 in (4.54). This makes it possible to draw a simplified version of the closed-loop model from Figure 4.2. This simplified closed-loop model is shown in Figure 4.3.

However, the closed-loop model in Figure 4.3 is not the only interpretation of a stable GE-LTI-SOE. An alternative explanation of the cross-correlation between u and  $\varepsilon_0$  can be given by the feedforward model in Figure 4.4. In this model, let

$$F_{f0}(z) = \frac{\Phi_{u\varepsilon_0}(z)}{\lambda_0},\tag{4.59a}$$

$$r_{f0}(t) = u(t) - F_{f0}(q)\varepsilon_0(t).$$
 (4.59b)



**Figure 4.3:** The GE-LTI-SOE can be interpreted as being part of a simplified linear closed-loop system if both  $G_{0,GE}$  and  $H_{0,GE}$  are stable.



**Figure 4.4:** The interpretation of the GE-LTI-SOE as a linear feedforward structure when  $G_{0,GE}$  and  $H_{0,GE}$  are stable.

With this definition, the transfer function  $F_{f0}(z)$  is stable since  $\Phi_{u\varepsilon_0}(z)$  by (4.45a) is absolutely convergent on the unit circle. Furthermore, it follows that

$$\Phi_{r_{f0}\varepsilon_0}(z) = \Phi_{u\varepsilon_0}(z) - F_{f0}(z)\lambda_0 = 0 \tag{4.60}$$

and  $r_{f0}$  and  $\varepsilon_0$  are thus uncorrelated.

The conclusion that can be drawn from the discussion in this section is that a GE-LTI-SOE always can be interpreted as being a part of an internally stable feedback or, in the case of a stable GE-LTI-SOE, feedforward system. Hence, by looking only at second order properties, it is impossible to disprove that any data set, with input and output measurements that fulfill Assumptions A1, A2 and A4, might have been generated by this closed-loop system. However, in some cases additional prior knowledge about the structure of the nonlinear system is available, and this knowledge can influence the interpretation of the LTI-SOE. In the next section, additional knowledge about the noise in the system will be discussed.



Figure 4.5: In some cases, the noise is assumed to be output additive.

## 4.6 Assumptions on the Noise

In the derivation of the OE-LTI-SOE and GE-LTI-SOE in the previous sections, no assumptions were made about the structure of the nonlinear system. Although structural assumptions are not necessary for the existence of the OE-LTI-SOE and the GE-LTI-SOE, it is hard to draw any conclusions about the properties and usefulness of these second order equivalents without any further information about the nonlinear system.

One important structural property of a system is how the noise enters. For most of the results in this thesis we will need the following assumption that says that the noise is additive and uncorrelated with the input and the noise-free output.

**Assumption A5.** Assume that the output y(t) can be written

$$y(t) = y_{nf}(t) + w(t), (4.61)$$

where  $y_{nf}$  is the noise-free response of the nonlinear system and not influenced by any other external signals than u, and where w is a noise term which is uncorrelated with u and  $y_{nf}$ .

In most cases, this assumption does not hold for a closed-loop system. This is due to the fact that the measured output that is fed back to the controller in such a system usually contains some noise. The input is thus in general correlated with the noise when it comes from a closed-loop system. Hence, Assumption A5 can be viewed as an assumption that the system has the structure shown in Figure 4.5.

However, the fact that Assumption A5 is needed for most of the results in this thesis is not the only reason why LTI approximations for systems in closed loop are not studied much here. Most of the results later in this thesis hold only for input signals with special distributions, e.g., Gaussian inputs or LTI filtered white noise inputs. For an open-loop system, this limitation is often not a problem since the input signal in many cases can be designed rather freely by the user. On the other hand, for a nonlinear closed-loop system, it is usually very hard to guarantee that the distribution of the input signal belongs to a certain class. Hence, most parts of the remaining chapters in this thesis deal with open-loop systems.

As has been mentioned previously, the fact that a certain system operates in open loop can be used to disprove that the GE-LTI-SOE represents the true system. This can be done because it may be necessary to view the GE-LTI-SOE as being a part of a closedloop system in order to explain a nonzero  $\Phi_{u \in 0}(z)$  (cf. Example 4.3).

This concludes our introductory discussion about the notion of LTI Second Order Equivalents. In this chapter, it has been shown that the OE-LTI-SOE and GE-LTI-SOE of

a nonlinear system are well-defined for rather general classes of input and output signals. Furthermore, properties and interpretations of these LTI approximations that hold for all signals in these classes have been discussed. In the Chapters 5 to 8, we will consider more restricted and specialized classes of input signals for which further properties of the OE-LTI-SOEs and GE-LTI-SOEs can be shown.

5

# **Basic Properties of LTI-SOEs**

In Example 4.2 in the previous chapter, it was shown that the OE-LTI-SOE of a static nonlinear system can be nonstatic. This observation can be viewed as an indication that some caution is needed when conclusions are drawn about the behavior of OE-LTI-SOEs. A behavior that intuitively seems correct, for example that the OE-LTI-SOE of a static nonlinear system should be static, can actually be erroneous.

In this chapter, we will make some rather straightforward assumptions about the input signals and investigate what these assumptions imply. More specifically, we will assume that the input signal is symmetrically distributed and show that this implies that the OE-LTI-SOE only depends on the odd part of the system. We will show that if a nonlinear system can be written as a linear combination of several subsystems with the same input, its OE-LTI-SOE can be written as a linear combination too. We will also consider input signals generated by filtering white noise through a minimum phase filter. It turns out that for such an input signal, spectral and residual analysis can be used for validation just like in the LTI case. Furthermore, minimum phase filtered white noise can be useful if an LTI system is identified in closed-loop and a nonlinear controller is used. This signal type guarantees also that the  $G_{0,OE}(z)$  and  $G_{0,GE}(z)$  will be equal. This will be discussed at the end of this chapter. First, however, we will investigate how output additive noise affects the OE-LTI-SOE.

## 5.1 Additive Noise

Noise can affect a nonlinear system in many ways. For example, the noise can be added to or multiplied with the input before it enters the actual system, or affect the output of the system. Although a large variation of noise dependency can be found in applications, we will here mainly study one type of noise, namely output additive noise that is uncorrelated with the input and the noise-free output as in Assumption A5. Actually,

the following lemma shows that this type of noise does not affect the OE-LTI-SOE of a nonlinear system.

#### Lemma 5.1

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A5 are fulfilled. This means that the additive noise w(t) in (4.61) is uncorrelated with the input and the noise-free output  $y_{nf}$ . Then the OE-LTI-SOE is not influenced by w(t).

**Proof:** Since the noise w by Assumption A5 is uncorrelated with u and  $y_{nf}$ , we have for any stable LTI model G(q) that

$$E((y(t) - G(q)u(t))^2) = E((y_{nf}(t) + w(t) - G(q)u(t))^2)$$
  
=  $E(y_{nf}(t) - G(q)u(t)^2) + E(w(t)^2).$  (5.1)

Hence, the criterion in Definition 4.1 can be rewritten

$$G_{0,OE}(q) = \underset{G \in \mathcal{G}}{\arg\min} E((y(t) - G(q)u(t))^2) = \underset{G \in \mathcal{G}}{\arg\min} E((y_{nf}(t) - G(q)u(t))^2)$$

and this shows that the OE-LTI-SOE is not influenced by the additive noise.

The result of the previous lemma is the rather convenient fact that the OE-LTI-SOE is independent of the noise if Assumption A5 holds. Although this assumption might be considered rather restrictive, it will be used frequently in this thesis. Some results about OE-LTI-SOEs where Assumption A5 is not required will be presented later in this chapter (see Sections 5.4 and 5.5). First, however, we will draw some conclusions about the different influence odd and even nonlinearities have on the OE-LTI-SOE.

## 5.2 Even and Odd Nonlinearities

Assume that Assumption A5 holds for a certain system and that the noise-free output  $y_{nf}$  can be written as  $y_{nf}(t) = f((u(t-k))_{k=0}^M)$  for some nonnegative integer M. This means that the system is an NFIR system (see Section 2.2), and in this case, we can divide the real-valued function f in an even part  $f_e$  and an odd part  $f_o$ 

$$f((u(t-k))_{k=0}^{M}) = \underbrace{\frac{f((u(t-k))_{k=0}^{M}) + f(-(u(t-k))_{k=0}^{M})}{2}}_{=f_{e}((u(t-k))_{k=0}^{M})} + \underbrace{\frac{f((u(t-k))_{k=0}^{M}) - f(-(u(t-k))_{k=0}^{M})}{2}}_{=f_{e}((u(t-k))_{k=0}^{M})}$$
(5.2)

such that

$$f_e(-(u(t-k))_{k=0}^M) = f_e((u(t-k))_{k=0}^M),$$
  
$$f_o(-(u(t-k))_{k=0}^M) = -f_o((u(t-k))_{k=0}^M).$$

If all simultaneous probability density functions for the process u are even functions, the OE-LTI-SOE of an NFIR system will only depend on the odd part  $f_o$  of the system. Hence, we have the following lemma.

#### Lemma 5.2

Consider an NFIR system with input u(t) and output y(t) such that Assumptions A1, A2 and A5 are fulfilled. The noise-free output  $y_{nf}$  from (4.61) can then, for some nonnegative integer M, be written

$$y_{nf}(t) = f((u(t-k))_{k=0}^{M}) = f_e((u(t-k))_{k=0}^{M}) + f_o((u(t-k))_{k=0}^{M}),$$
(5.3)

where  $f_e$  and  $f_o$  are even and odd functions, respectively. Assume that all simultaneous probability density functions for the process u are even functions. Then the OE-LTI-SOE depends only on the odd part  $f_o$  of the system, i.e.,

$$G_{0,OE}(q) = \underset{G \in \mathcal{G}}{\arg\min} \operatorname{E}(\left(f_o((u(t-k))_{k=0}^M) - G(q)u(t)\right)^2).$$
(5.4)

Proof: From Lemma 5.1 we have

$$G_{0,OE}(q) = \operatorname*{arg\,min}_{G \in \mathcal{G}} \mathrm{E}((y_{nf}(t) - G(q)u(t))^2).$$

Using (5.3), this mean-square error criterion can be rewritten

$$\begin{split} & \mathcal{E}((y_{nf}(t) - G(q)u(t))^2) = \mathcal{E}(\left(f_o((u(t-k))_{k=0}^M) - G(q)u(t)\right)^2) \\ & + \mathcal{E}(\left(f_e((u(t-k))_{k=0}^M)\right)^2) + 2\mathcal{E}(f_o((u(t-k))_{k=0}^M)f_e((u(t-k))_{k=0}^M)) \\ & - 2\sum_{j=0}^{\infty} g(j)\mathcal{E}(u(t-j)f_e((u(t-k))_{k=0}^M)). \end{split}$$

The cross-terms in this expansion are equal to zero since

$$f_o((u(t-k))_{k=0}^M)f_e((u(t-k))_{k=0}^M)$$

and

$$u(t-j)f_e((u(t-k))_{k=0}^M),$$

 $j \in \mathbb{N}$ , are odd functions of  $(u(t-k))_{k=0}^{\max(M,j)}$  and since all probability density functions of u are even functions. Hence

$$E((y_{nf}(t) - G(q)u(t))^{2}) = E((f_{o}((u(t-k))_{k=0}^{M}) - G(q)u(t))^{2}) + E((f_{e}((u(t-k))_{k=0}^{M}))^{2})$$
(5.5)

and (5.4) follows.

The fact that the OE-LTI-SOE is independent of even nonlinearities when a symmetrically distributed input signal is used implies that if two nonlinear systems only differ by even nonlinearities, they will have the same OE-LTI-SOE. However, the variance of the model residuals will of course be larger if a system contains large even nonlinearities. The influence of the even and odd parts of a system on the variance and spectral density of the model residuals is shown by the following lemma.

#### Lemma 5.3

Consider an NFIR system with input u(t) and output y(t) such that Assumptions A1, A2 and A5 are fulfilled. The noise-free output  $y_{nf}(t)$  from (4.61) can then, for some nonnegative integer M, be written as in (5.3). Assume that all simultaneous probability density functions for the process u are even functions. Then the z-spectrum of the OE-LTI-SOE residuals  $\eta_0(t)$  can be written

$$\Phi_{\eta_0}(z) = \Phi_{d_o}(z) + \Phi_{y_e}(z) + \Phi_w(z), \tag{5.6}$$

where  $d_o(t) = f_o((u(t-k))_{k=0}^M) - G_{0,OE}(q)u(t)$  and  $y_e(t) = f_e((u(t-k))_{k=0}^M)$ .

**Proof:** Assumption A5 gives that

$$R_{\eta_0}(\tau) = \mathcal{E}((y_{nf}(t) - G_{0,OE}(q)u(t))(y_{nf}(t-\tau) - G_{0,OE}(q)u(t-\tau))) + R_w(\tau), \quad \forall \tau \in \mathbb{Z}$$

and using (5.3), we get

$$\begin{aligned} R_{\eta_0}(\tau) &= \mathrm{E}((d_o(t) + y_e(t))(d_o(t - \tau) + y_e(t - \tau))) + R_w(\tau) \\ &= R_{d_o}(\tau) + R_{d_o y_e}(\tau) + R_{y_e d_o}(\tau) + R_{y_e}(\tau) + R_w(\tau), \quad \forall \tau \in \mathbb{Z}. \end{aligned}$$

Since

$$R_{d_o y_e}(\tau) = \mathcal{E}(f_o((u(t-k))_{k=0}^M) f_e((u(t-k-\tau))_{k=0}^M)) - \sum_{j=0}^{\infty} g_{0,OE}(j) \mathcal{E}(u(t-j) f_e((u(t-k-\tau))_{k=0}^M))$$

contains only expectations of odd functions and since all probability density functions of u are even, we get  $R_{d_o y_e}(\tau) = R_{y_e d_o}(\tau) = 0$  for all  $\tau \in \mathbb{Z}$ . Hence,

$$R_{\eta_0}(\tau) = R_{d_o}(\tau) + R_{y_e}(\tau) + R_w(\tau), \quad \forall \tau \in \mathbb{Z}$$
(5.7)

and (5.6) follows.

In particular, (5.7) shows that the variance of the OE-LTI-SOE residuals  $\eta_0(t)$  can be written

$$R_{\eta_0}(0) = \mathcal{E}(\eta_0(t)^2) = \mathcal{E}(\left(f_o((u(t-k))_{k=0}^M) - G(q)u(t)\right)^2) + \mathcal{E}(\left(f_e((u(t-k))_{k=0}^M)\right)^2) + \mathcal{E}(w(t)^2).$$
(5.8)

This expression is valid when Assumption A5 holds and when the input signal has even probability density functions, and it shows that there are three conceptually different contributions to  $R_{\eta_0}(0)$ . The first term in (5.8) is the variance of the unmodeled part of the odd nonlinearities, while the second and third terms are the variance of the even part of the system and of the noise, respectively.

Usually, it is not obvious how the input signal should be designed in order to minimize the variance of the residuals. In Section 5.4, we will consider inputs that have been generated by filtering white noise through a minimum phase filter. Later in Section 5.5.1, it will be shown that these inputs reduce the variance of the residuals. First, we will study OE-LTI-SOEs for nonlinear systems with parallel subsystems.

## 5.3 Sums of Nonlinear Systems

If the output from a nonlinear system can be written as the sum of the outputs from several nonlinear subsystems with the same input, the OE-LTI-SOE of the complete system will be the sum of the OE-LTI-SOEs of the corresponding subsystems according to the following theorem.

#### Theorem 5.1

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A5 are satisfied. Assume also that the system can be written as a sum of several nonlinear subsystems, i.e., as

$$y(t) = \sum_{i=1}^{M} y_i(t) + w(t),$$
(5.9a)

$$y_i(t) = f_i((u(t-k))_{k=0}^{\infty})$$
 (5.9b)

and that each nonlinearity  $f_i$  has an OE-LTI-SOE  $G_{0,OE,i}(z)$ . Then the OE-LTI-SOE of the complete nonlinear system (from u to y) is

$$G_{0,OE}(z) = \sum_{i=1}^{M} G_{0,OE,i}(z).$$
(5.10)

**Proof:** The z-cross-spectrum can be written as

$$\Phi_{yu}(z) = \sum_{i=1}^{M} \Phi_{y_i u}(z).$$

Hence, the OE-LTI-SOE of the complete system can be written

$$\begin{aligned} G_{0,OE}(z) &= \frac{1}{r_u L(z)} \left[ \frac{\sum_{i=1}^{M} \Phi_{y_i u}(z)}{L(z^{-1})} \right]_{\text{causal}} = \sum_{i=1}^{M} \frac{1}{r_u L(z)} \left[ \frac{\Phi_{y_i u}(z)}{L(z^{-1})} \right]_{\text{causal}} \\ &= \sum_{i=1}^{M} G_{0,OE,i}(z), \end{aligned}$$

where we have used (4.3) and the fact that the causality operator is linear.

Theorem 5.1 implies that OE-LTI-SOEs of parallel connected nonlinear systems can be calculated by studying the subsystems. Hence, if a nonlinear system can be written as a linear combination of a number of nonlinear subsystems with the same input, the OE-LTI-SOE of this complete system can be written as the same linear combination of the OE-LTI-SOEs of the subsystems. The OE-LTI-SOE can thus be viewed as a linear operator on the set of all nonlinear systems.

In the next section, we will consider OE-LTI-SOEs for inputs generated by filtering white noise through a minimum phase filter. It turns out that these inputs imply some useful properties of the OE-LTI-SOEs.

 $\square$ 

## 5.4 Minimum Phase Input Filters

A common way to generate a signal u such that its spectral density is equal to some predefined function is to filter white noise e through an LTI filter L(z). Since, by Lemma 2.1, the result of this procedure will be a signal with spectral density

$$\Phi_u(e^{i\omega}) = |L(e^{i\omega})|^2 R_e(0),$$

it is often convenient to consider only  $|L(e^{i\omega})|$  when the filter L(z) is designed and let the phase  $\arg(L(e^{i\omega}))$  become whatever it becomes. For example, this works well if u is going to be used as input to an LTI system in a linear identification experiment.

However, if the signal u is to be used for an LTI approximation of a nonlinear system, the phase of the prefilter is crucial for the behavior of this approximation. In the following theorem, it will be shown that in this case, it is beneficial to generate the input signal by filtering white noise through an LTI filter which has the minimum phase property (see Definition 2.4).

#### Theorem 5.2

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled. Assume that the input signal has been generated by filtering white, possibly non-Gaussian, noise e(t) through a minimum phase filter  $L_m(z)$ . Assume also that any other external signals that affect the output are independent of u. Then the OE-LTI-SOE is regular (see Definition 4.2) and can be written

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)} = \frac{\Phi_{ye}(z)}{L_m(z)R_e(0)}.$$
(5.11)

**Proof:** The canonical spectral factorization of  $\Phi_u(z)$  (cf. (4.1)) is

$$\begin{split} L(z) &= \frac{L_m(z)}{l_m(0)}, \\ r_u &= l_m(0)^2 R_e(0). \end{split}$$

Using (4.3) from Theorem 4.1, this gives

$$\begin{split} G_{0,OE}(z) &= \frac{l_m(0)}{l_m(0)^2 R_e(0) L_m(z)} \left[ \frac{l_m(0) \Phi_{yu}(z)}{L_m(z^{-1})} \right]_{\text{causal}} \\ &= \frac{1}{R_e(0) L_m(z)} \left[ \frac{\Phi_{yu}(z)}{L_m(z^{-1})} \right]_{\text{causal}} \\ &= \frac{1}{R_e(0) L_m(z)} \left[ \frac{\Phi_{ye}(z) L_m(z^{-1})}{L_m(z^{-1})} \right]_{\text{causal}} \\ &= \frac{1}{R_e(0) L_m(z)} \left[ \Phi_{ye}(z) \right]_{\text{causal}}, \end{split}$$

where we have used Lemma 2.1 in the third equality. The nonlinear system is, by our standard assumption, causal and u is independent of all other external signals that affect

the output. This, together with the fact that e is a white noise process, implies that y(t) is independent of  $e(t - \tau)$  for all  $\tau < 0$ . Hence  $R_{ye}(\tau) = 0$  for all  $\tau < 0$  and

$$\Phi_{ye}(z) = \sum_{\tau=0}^{\infty} R_{ye}(\tau) z^{-\tau}.$$

Since the series  $\Phi_{ye}(z)$  contains no positive powers of z, taking the causal part does not remove anything. Hence, we have

$$G_{0,OE}(z) = \frac{1}{R_e(0)L_m(z)} \left[\Phi_{ye}(z)\right]_{\text{causal}} = \frac{\Phi_{ye}(z)}{L_m(z)R_e(0)} = \frac{\Phi_{yu}(z)}{\Phi_u(z)}$$

and we have shown (5.11).

The assumption in Theorem 5.2 that any other external signals that affect the output should be independent of the input u implies that this theorem usually cannot be applied if the system is a closed-loop LTI system. However, for open-loop systems, the conditions in Theorem 5.2 are not very restrictive.

Actually, the reason why the OE-LTI-SOE in Example 4.2 on page 42 was regular is that the input signal in that example was generated by filtering white noise through a minimum phase filter. It is interesting to see what happens with the OE-LTI-SOE if the input filter in Example 4.2 is replaced by a non-minimum phase filter giving the same  $\Phi_u(z)$ . This is done in the following example.

Example 5.1 –

Consider the static nonlinear system

$$y(t) = u(t)^3$$
 (5.12)

with the input

$$u(t) = \frac{1}{2}e(t) + e(t-1),$$

where e(t) is a sequence of independent random variables with uniform distribution over the interval [-1, 1]. For the moment, let  $\tilde{R}_{yu}(\tau)$  denote the cross-covariance function in Example 4.2. Then

$$\begin{aligned} R_{yu}(0) &= \mathcal{E}(u(t)^4) = \mathcal{E}\left(\left(\frac{1}{2}e(t) + e(t-1)\right)^4\right) = \mathcal{E}\left(\left(e(t) + \frac{1}{2}e(t-1)\right)^4\right) \\ &= \tilde{R}_{yu}(0) = \frac{91}{240}, \\ R_{yu}(1) &= \mathcal{E}(u(t)^3u(t-1)) \\ &= \mathcal{E}\left(\left(\frac{1}{2}e(t) + e(t-1)\right)^3 \cdot \left(\frac{1}{2}e(t-1) + e(t-2)\right)\right) \\ &= \mathcal{E}\left(\left(e(t) + \frac{1}{2}e(t-1)\right)^3 \cdot \left(e(t+1) + \frac{1}{2}e(t)\right)\right) \\ &= \tilde{R}_{yu}(-1) = \frac{34}{240}, \end{aligned}$$

$$\begin{aligned} R_{yu}(-1) &= \mathcal{E}(u(t)^{3}u(t+1)) \\ &= \mathcal{E}(\left(\frac{1}{2}e(t) + e(t-1)\right)^{3} \cdot \left(\frac{1}{2}e(t+1) + e(t)\right)) \\ &= \mathcal{E}(\left(e(t) + \frac{1}{2}e(t-1)\right)^{3} \cdot \left(e(t-1) + \frac{1}{2}e(t-2)\right)) \\ &= \tilde{R}_{yu}(1) = \frac{46}{240}, \\ R_{yu}(\tau) &= \tilde{R}_{yu}(\tau) = 0, \quad \forall \tau \in \mathbb{Z} \setminus \{-1, 0, 1\}. \end{aligned}$$

This gives

$$\Phi_{yu}(z) = \frac{1}{240} (46z + 91 + 34z^{-1}) = \frac{1}{240} \left(1 + \frac{1}{2}z^{-1}\right) (68 + 46z).$$

Furthermore, Lemma 2.1 gives

$$\Phi_u(z) = \left(\frac{1}{2} + z^{-1}\right) \cdot \frac{1}{3} \cdot \left(\frac{1}{2} + z\right) = \frac{1}{12} \left(2z + 5 + 2z^{-1}\right)$$

The canonical spectral factor of  $\Phi_u(z)$  is  $L(z) = 1 + \frac{1}{2}z^{-1}$  and  $r_u = \frac{1}{3}$ . Theorem 4.1 gives

$$\begin{split} G_{0,OE}(z) &= \frac{1}{r_u L(z)} \left[ \frac{\Phi_{yu}(z)}{L(z^{-1})} \right]_{\text{causal}} = \frac{3}{1 + \frac{1}{2}z^{-1}} \left[ \frac{46z + 91 + 34z^{-1}}{240(1 + \frac{1}{2}z)} \right]_{\text{causal}} \\ &= \frac{1}{80} \cdot \frac{1}{1 + \frac{1}{2}z^{-1}} \left[ \frac{9z}{1 + \frac{1}{2}z} + 74 + 34z^{-1} \right]_{\text{causal}} \\ &= \frac{1}{80} \cdot \frac{74 + 34z^{-1}}{1 + \frac{1}{2}z^{-1}} = \frac{1}{40} \cdot \frac{37 + 17z^{-1}}{1 + \frac{1}{2}z^{-1}} = \frac{0.925 + 0.425z^{-1}}{1 + 0.5z^{-1}}. \end{split}$$

Here, just like in Example 4.2, the OE-LTI-SOE of the static nonlinear system  $y(t) = u(t)^3$  is nonstatic. However, the OE-LTI-SOE in Example 4.2 is not equal to the OE-LTI-SOE here, since the two input signals have different distributions.

The input signals in Examples 4.2 and 5.1 are similar in the sense that they have equal z-spectra. Furthermore, the probability density function for one input signal component, (or the amplitude distribution of every single u(t)), is the same in both examples. Since the nonlinear system is static, this also implies that the probability density functions of a single y(t) are equal in these examples.

Despite these similarities between the input signals in Example 4.2 and 5.1, these inputs generate different OE-LTI-SOEs because the simultaneous probability density functions of u(t) and u(t-1) are different in the two examples. As we have not calculated  $R_{\eta_0}(0)$  in these examples, it is not obvious which OE-LTI-SOE that is most successful in approximating the true system. However, in the next section we will show that it is always better to use an input generated by a minimum phase filter than a non-minimum phase filter.

## 5.5 Properties for Regular OE-LTI-SOEs

The fact that the OE-LTI-SOE is regular for, for example, input signals generated by filtering white noise through a minimum phase filter is convenient, since it makes it possible to calculate the OE-LTI-SOE without spectral factorization of the input z-spectrum. In addition, OE-LTI-SOEs of this kind exhibit a number of interesting properties.

#### 5.5.1 Optimality Properties

The perhaps most obvious property that holds for a regular OE-LTI-SOE concerns the residuals  $\eta_0(t)$ . In the following lemma it will be shown that for such an OE-LTI-SOE the residuals will be uncorrelated with *all* input signal components (cf. Corollary 4.2).

#### Lemma 5.4

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1 and A2 are fulfilled. Assume that the OE-LTI-SOE is regular, i.e., that it can be written as

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)},$$
(5.13)

and let

$$\eta_0(t) = y(t) - G_{0,OE}(q)u(t).$$
(5.14)

Then it follows that

$$\Phi_{\eta_0 u}(z) = \Phi_{yu}(z) - G_{0,OE}(z)\Phi_u(z) = 0, \qquad (5.15a)$$

$$\Phi_{\eta_0}(z) = \Phi_y(z) - G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1}).$$
(5.15b)

**Proof:** The expression for  $\Phi_{\eta_0 u}(z)$  in (5.15a) follows directly from (5.13) and (5.14). Furthermore, using Lemma 2.1, (5.13) and (5.14) also give

$$\Phi_{\eta_0}(z) = \Phi_y(z) - G_{0,OE}(z)\Phi_{uy}(z) - \Phi_{yu}(z)G_{0,OE}(z^{-1}) + G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1}) = \Phi_y(z) - G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1})$$

and hence (5.15b) has been shown.

The fact that (5.15a) holds when  $G_{0,OE}(z)$  is regular shows that the OE-LTI-SOE in this case really is the best noncausal LTI model since the model residuals are uncorrelated with all input signal components. This is no surprise since it can be shown that the ratio  $\Phi_{yu}(z)/\Phi_u(z)$  is always the mean-square error optimal noncausal LTI model, i.e., the noncausal LTI-SOE (see Section 3.2.3). If this ratio is causal, it is of course equal to the OE-LTI-SOE.

Intuitively, it seems that it should always be a good idea to use input signals for which the OE-LTI-SOE is equal to the noncausal LTI-SOE. As a matter of fact, input signals for which the OE-LTI-SOE of a nonlinear system is regular exhibit the following optimality property.

#### Theorem 5.3

Consider a nonlinear system with input  $u_1(t)$  and output  $y_1(t)$  such that Assumptions A1 and A2 are fulfilled. Let  $G_{0,OE,1}(z)$  denote the OE-LTI-SOE of the nonlinear system with respect to  $u_1$  and assume that it is regular, i.e., that it can be written as

$$G_{0,OE,1}(z) = \frac{\Phi_{y_1u_1}(z)}{\Phi_{u_1}(z)}.$$

Furthermore, let  $\eta_{0,1}(t) = y_1(t) - G_{0,OE,1}(q)u_1(t)$ .

Consider also another input signal  $u_2(t)$  to the same nonlinear system. Assume that this signal generates the output  $y_2(t)$  and that  $u_2(t)$  and  $y_2(t)$  satisfy Assumptions A1 and A2. Let  $G_{0,OE,2}(z)$  denote the OE-LTI-SOE of the nonlinear system with respect to  $u_2$  and let  $\eta_{0,2}(t) = y_2(t) - G_{0,OE,2}(q)u_2(t)$ . Assume that

$$\begin{split} \Phi_{u_2}(e^{i\omega}) &= \Phi_{u_1}(e^{i\omega}), \quad \forall \omega \in [-\pi, \pi], \\ |\Phi_{y_2u_2}(e^{i\omega})| &= |\Phi_{y_1u_1}(e^{i\omega})|, \quad \forall \omega \in [-\pi, \pi], \\ R_{y_2}(0) &= R_{y_1}(0). \end{split}$$

Then the model residual variance for the OE-LTI-SOE corresponding to  $u_2$  cannot be smaller than it is for the one corresponding to  $u_1$ , i.e.,

$$R_{\eta_{0,2}}(0) \ge R_{\eta_{0,1}}(0). \tag{5.16}$$

**Proof:** From (4.12) we have for any OE-LTI-SOE that

$$R_{\eta_0}(0) = R_y(0) - \frac{1}{2\pi} \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega})|^2 \Phi_u(e^{i\omega}) \, d\omega.$$

For any input signal, the noncausal LTI-SOE is always  $\frac{\Phi_{yu}(z)}{\Phi_u(z)}$ . It is easy to verify that (4.12) holds also for the noncausal LTI-SOE if  $G_{0,OE}(e^{i\omega})$  is replaced by  $\frac{\Phi_{yu}(e^{i\omega})}{\Phi_u(e^{i\omega})}$ . As the stable and causal LTI systems are a subset of the stable and noncausal, it follows that the OE-LTI-SOE will always have a minimum mean-square error that is greater than or equal to the minimum mean-square error that is obtained for the noncausal LTI-SOE. Hence,

$$R_{\eta_{0,2}}(0) \ge R_{y_2}(0) - \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| \frac{\Phi_{y_2 u_2}(e^{i\omega})}{\Phi_{u_2}(e^{i\omega})} \right|^2 \Phi_{u_2}(e^{i\omega}) d\omega$$
$$= R_{y_1}(0) - \frac{1}{2\pi} \int_{-\pi}^{\pi} \left| \frac{\Phi_{y_1 u_1}(e^{i\omega})}{\Phi_{u_1}(e^{i\omega})} \right|^2 \Phi_{u_1}(e^{i\omega}) d\omega = R_{\eta_{0,1}}(0)$$

since  $G_{0,OE,1}(z) = \frac{\Phi_{y_1u_1}(z)}{\Phi_{u_1}(z)}$ .

Theorem 5.3 shows that, for example, a given minimum phase generated input signal is optimal over a set of other inputs in the sense that it minimizes the variance of the

model residuals. Usually, it is not easy to describe this set of input signals and in some cases it might actually be empty. However, the input signals in Examples 4.2 and 5.1 fulfill the assumptions in Theorem 5.3. Hence, it follows from (5.16) that the variance of the model residual in Example 4.2 is less than or equal to the corresponding variance in Example 5.1. Actually, it can be shown that the variance in the first example is strictly less than the variance in the second one.

It should be noted that there is no guarantee that a regular OE-LTI-SOE will be a useful model of a nonlinear system. The only thing that is guaranteed is that it will be easier to see if an estimated model is close to the OE-LTI-SOE since some useful validation methods will work in this case. This will be discussed more in the next section.

#### 5.5.2 Spectral and Residual Analysis

A common way to validate an estimated model of an open-loop LTI system is to compare the frequency response of the model with a nonparametric frequency response estimate obtained by spectral analysis. If these frequency responses are similar this indicates that the order of the parametric model is sufficiently high and that the numerical computation of the estimate has been successful. In Ljung (1999, Sec. 6.4), it is shown that the spectral analysis frequency response estimate  $\hat{G}_N(e^{i\omega_0})$  based on N measurements can be written

$$\hat{G}_N(e^{i\omega_0}) = rac{\hat{\Phi}_{yu}^N(e^{i\omega_0})}{\hat{\Phi}_u^N(e^{i\omega_0})},$$

where  $\hat{\Phi}_{u}^{N}(e^{i\omega_{0}})$  and  $\hat{\Phi}_{yu}^{N}(e^{i\omega_{0}})$  are estimates of the spectral and cross-spectral densities that can be written as smoothed periodograms.

If an LTI model is estimated for an open-loop nonlinear system, it might be tempting to use spectral analysis as a validation method also in this case. However, the spectral analysis frequency response estimate can be quite different from the frequency response of the OE-LTI-SOE and is thus in general useless for validation purposes. Only when the OE-LTI-SOE is regular, spectral analysis can be used as a validation method.

#### — Example 5.2 -

Consider once again the system and input signals from Examples 4.2 and 5.1. In these examples, it was shown that the OE-LTI-SOEs of this system are

$$G_{0,OE,1}(z) = \frac{0.85 + 0.575z^{-1}}{1 + 0.5z^{-1}}$$

and

$$G_{0,OE,2}(z) = \frac{0.925 + 0.425z^{-1}}{1 + 0.5z^{-1}},$$

respectively. Here, two data sets with 10 000 noise-free input and output measurements have been generated. The first of these data sets was generated with a realization of the minimum phase filtered signal from Example 4.2 as input while the second data set was generated with a realization of the non-minimum phase filtered signal from Example 5.1 as input.



(a) The OE-LTI-SOE (dashed) and the spectral analysis estimate (solid) for an input generated by a minimum phase filter.

(b) The OE-LTI-SOE (dashed) and the spectral analysis estimate (solid) for an input generated by a non-minimum phase filter.

**Figure 5.1:** A nonparametric frequency response estimate will be a good approximation of the OE-LTI-SOE only when  $G_{0,OE}(z)$  is regular.

Nonparametric frequency response estimates have been computed from these data sets using spectral analysis with a Hamming window of lag size 30. These estimates are shown in Figure 5.1 together with the frequency responses of the corresponding OE-LTI-SOEs. The MATLAB code that has been used to generate this figure is available in Appendix B.1.

In Figure 5.1 it can be seen that there is a close match between the OE-LTI-SOE and the nonparametric frequency response estimate when the input has been generated by a minimum phase filter. However, when the input has been generated by a non-minimum phase filter, the OE-LTI-SOE is quite different from the nonparametric estimate.

The conclusion that can be drawn from Example 5.2 is that for LTI approximations of nonlinear systems, spectral analysis can be used as a validation method only when an input signal that guarantees that  $G_{0,OE}(z)$  is regular has been used. An additional property of such input signals is that they make the result of another validation method, residual analysis, easier to interpret. Residual analysis can be used to check if there is remaining correlation between the input and the residuals for a certain model. Such remaining correlation indicates that the model order might be too low. If an accurate model of an open-loop LTI system has been found, the residuals will be uncorrelated with the input signal.

However, if the OE-LTI-SOE of a nonlinear system has been estimated, it will according to (4.11) in general only have residuals that are uncorrelated with past and present input signal components. If it is not known that the true system is nonlinear, the remaining correlation between the residuals and future input components might be taken as an indication that the system actually is a closed-loop system. However, if an input that guarantees that the OE-LTI-SOE is regular has been used, this cannot happen since the residuals then by (5.15a) will be uncorrelated with all input components. In the next section, this property will be used for another purpose.



Figure 5.2: A nonlinear closed-loop system.

#### 5.5.3 Closed-loop Identification

One application where LTI approximations of nonlinear systems are useful is closedloop identification of LTI systems which operate under nonlinear feedback. Consider the closed loop system in Figure 5.2. This system consists of an unknown LTI plant  $G_0$  and a nonlinear controller  $\mathcal{R}$ , and its output can be described as

$$y(t) = G_0(q)u(t) + w(t),$$
(5.17)

where u and w are correlated due to the feedback loop. The reference signal r is here assumed to be independent of the process noise w.

Suppose that a model of  $G_0$  is desired and that measurements of r, u and y are available. In this case, the most natural way to estimate such a model is often to use the *direct* prediction-error method (Ljung, 1999, Sec. 13.5). Provided that the model structure is flexible enough to contain the true system, including the true noise description, this approach defines a consistent estimator of  $G_0$ .

However, if the model of  $G_0$  is to be used for controller design, there might be one drawback with the direct approach. When an approximate model is used for controller design, it is usually appropriate that it is as accurate as possible in a frequency interval around the desired crossover frequency. In the open-loop case, this accuracy can be increased by the use of a frequency weighting. Unfortunately, in the closed-loop case, such a frequency weighting cannot be used in the direct approach without making the estimator of  $G_0$  biased.

A solution to this problem when the complete closed-loop system is linear is given by the two-step method (Van den Hof and Schrama, 1993) or, when a nonlinear controller is present, a version of this method called the projection method (Forssell and Ljung, 2000). Both these methods can be viewed as ways to translate the closed-loop identification problem to an open-loop problem where frequency weighting can be used. The main idea used in the two-step and projection methods is to first estimate an LTI model S(z)from r to u and then to use this model to construct a simulated input signal  $\hat{u}(t) =$ S(q)r(t). Finally, the identification of  $G_0$  is performed using measurements of y and of the simulated input  $\hat{u}$  instead of the true input u.

The reason why it is beneficial to use  $\hat{u}$  instead of u in the identification of  $G_0$  is that this enables the use of frequency weighting since  $\hat{u}$  and the noise will be uncorrelated, just like for an open-loop system. This is due to the fact that if an output error model

$$u(t) = S(q)r(t) + \eta(t)$$

is used to describe the mapping from r to u and if the identification of S(z) is successful, the result will, in the case of linear feedback, be that

$$S(z) = \frac{\Phi_{ur}(z)}{\Phi_r(z)}.$$
(5.18)

Let  $\tilde{u}(t) = u(t) - \hat{u}(t)$ . Then (5.18) implies that

$$\Phi_{\hat{u}\hat{u}}(z) = \Phi_{u\hat{u}}(z) - \Phi_{\hat{u}}(z) = \Phi_{ur}(z)S(z^{-1}) - S(z)\Phi_r(z)S(z^{-1}) = 0$$

and (5.17) can thus be rewritten as

$$y(t) = G_0(q)\hat{u}(t) + \tilde{w}(t), \tag{5.19}$$

where  $\tilde{w}(t) = w(t) + G_0(q)\tilde{u}(t)$  is uncorrelated with  $\hat{u}$  since both w and  $\tilde{u}$  are uncorrelated with  $\hat{u}$ . Hence, the system (5.19) can be viewed as an open-loop system and frequency weighting can thus be used.

For a linear closed-loop system, S(z) is simply the ordinary sensitivity function, which of course is causal. However, when a nonlinear controller is present in the closedloop system, the mapping from r to u is nonlinear. In that case, the ratio  $\frac{\Phi_{ur}(z)}{\Phi_{r}(z)}$  might be noncausal if we want a stable interpretation of it. This is also pointed out in Forssell and Ljung (2000). The main difference between the projection method and the two-step method is that in the former, a noncausal FIR model is used to model the mapping from r to u, while in the latter only a causal S(z) is used. Hence, the projection method is applicable to closed-loop systems with nonlinear controllers while the two-step method in general is not.

However, from the discussion in Section 5.4 we know that if, for example, the reference signal has been generated by filtering white noise through a minimum phase filter, then  $\frac{\Phi_{ur}(z)}{\Phi_r(z)}$  will be causal since the nonlinear mapping from r to u is causal and r and w are independent. This means that if a minimum phase generated input signal is used, there is no need to use a noncausal S(z), and hence the two-step method can be applied instead of the projection method.

One advantage of the two-step method is that S(z) can be a rational function. In the projection method in Forssell and Ljung (2000), S(z) is a noncausal FIR model, which means that the true sensitivity function usually cannot be modeled exactly even when the controller is linear. With this observation in mind, it seems that the two-step method is at least an as good alternative for closed-loop identification of an LTI system with nonlinear feedback as the projection method, provided that the reference signal has been designed such that  $\frac{\Phi_{ur}(z)}{\Phi_r(z)}$  will be stable and causal independently of the structure of the controller. One example of a class of such reference signals is minimum phase filtered white noise.

Finally, it should be emphasized that the discussion here is based on properties of OE-LTI-SOEs and that it hence is valid mostly for large data sets. As a matter of fact, the use of a noncausal S(z) in the projection method might be useful for smaller data sets also in cases where S(z) asymptotically will be causal (Forssell and Ljung, 2000).

### 5.6 LTI-SOEs with a General Error Model

In the previous section, it was shown that regular OE-LTI-SOEs exhibit a number of interesting properties. It is thus a legitimate question whether these input signals also generate GE-LTI-SOEs with special properties.

Hence, we will now shift focus and discuss GE-LTI-SOEs of nonlinear systems with input signals such that the OE-LTI-SOE is regular. In Example 4.3, it was shown that in general  $G_{0,OE}(z)$  and  $G_{0,GE}(z)$  are not equal even for open-loop nonlinear systems. However, when the OE-LTI-SOE is regular, it follows that  $G_{0,GE}(z)$  will be equal to  $G_{0,OE}(z)$ . This is shown in the following theorem.

#### Theorem 5.4

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A4 are fulfilled. Assume that the input signal is such that the OE-LTI-SOE is regular, i.e., that it can be written as

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)}.$$
(5.20)

Assume also that

$$\Phi_{\eta_0}(z) = \Phi_y(z) - G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1})$$

from (5.15b) has a canonical spectral factorization

$$\Phi_{\eta_0}(z) = L_{\eta_0}(z) r_{\eta_0} L_{\eta_0}(z^{-1})$$
(5.21)

with  $r_{\eta_0} > 0$ . Then the GE-LTI-SOE is

$$G_{0,GE}(z) = G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)},$$
(5.22a)

$$H_{0,GE}(z) = L_{\eta_0}(z). \tag{5.22b}$$

Proof: Let

$$T(z) = \begin{pmatrix} L(z) & 0\\ z^{-1}G_{0,OE}(z)L(z) & L_{\eta_0}(z) \end{pmatrix},$$
$$Q_{\zeta} = \begin{pmatrix} r_u & 0\\ 0 & r_{\eta_0} \end{pmatrix},$$

where L(z) and  $r_u$  are factors in the canonical spectral factorization of  $\Phi_u(z)$  according to (4.1). Then

$$T(z)Q_{\zeta}T^{T}(z^{-1}) = \begin{pmatrix} L(z) & 0\\ z^{-1}G_{0,OE}(z)L(z) & L_{\eta_{0}}(z) \end{pmatrix} \begin{pmatrix} r_{u} & 0\\ 0 & r_{\eta_{0}} \end{pmatrix}$$
$$\cdot \begin{pmatrix} L(z^{-1}) & zG_{0,OE}(z^{-1})L(z^{-1})\\ 0 & L_{\eta_{0}}(z^{-1}) \end{pmatrix} = \begin{pmatrix} \Phi_{u}(z) & z\Phi_{uy}(z)\\ z^{-1}\Phi_{yu}(z) & \Phi_{y}(z) \end{pmatrix} = \Phi_{\zeta}(z),$$

where we have used (5.20) in the second equality. Since T(z) and

$$T^{-1}(z) = \begin{pmatrix} \frac{1}{L(z)} & 0\\ -\frac{z^{-1}G_{0,OE}(z)}{L_{\eta_0}(z)} & \frac{1}{L_{\eta_0}(z)} \end{pmatrix}$$

both are analytic in  $\{z \in \mathbb{C} \mid |z| \ge 1\}$ ,  $T(\infty) = I$  and  $Q_{\zeta} \succ 0$ , we have found the canonical spectral factorization of  $\Phi_{\zeta}(z)$ , and from (4.32) in Theorem 4.5 we obtain

$$G_{0,GE}(z) = \frac{zT_{21}(z)}{T_{11}(z)} = G_{0,OE}(z),$$
  

$$H_{0,GE}(z) = \frac{T_{11}(z)T_{22}(z) - T_{12}(z)T_{21}(z)}{T_{11}(z)} = L_{\eta_0}(z).$$

Theorem 5.4 shows that  $G_{0,OE}(z)$  and  $G_{0,GE}(z)$  will be equal if the OE-LTI-SOE is regular and that  $H_{0,GE}(z)$  in this case will be equal to the canonical spectral factor of  $\Phi_{\eta_0}(z)$ . Hence, the GE-LTI-SOE will be stable when the OE-LTI-SOE is regular and a canonical spectral factorization of a matrix-valued z-spectrum will not be needed for the calculation of the GE-LTI-SOE. Furthermore, Theorem 5.4 can be used to describe how even and odd nonlinearities will affect the GE-LTI-SOE.

Consider an NFIR system with an input signal that has even probability density functions and is such that Lemmas 5.2 and 5.3 and Theorem 5.4 can be applied. In this case,  $G_{0,GE}(z) = G_{0,OE}(z)$  will, by Lemma 5.2, depend only on the odd nonlinearities. Furthermore, (5.6) shows that

$$\Phi_{\eta_0}(z) = \Phi_{d_o}(z) + \Phi_{y_e}(z) + \Phi_w(z).$$

Hence, (5.22b) implies that there will be three different contributions to  $H_{0,GE}(z)$ . The first,  $\Phi_{d_o}(z)$ , is the z-spectrum of the unmodeled odd nonlinear part of the system output, while  $\Phi_{y_e}(z)$  and  $\Phi_w(z)$  are the z-spectra of the contributions to the output from the even part and the noise, respectively. Furthermore, Theorem 5.4 also gives the following corollary.

#### Corollary 5.1

Consider a nonlinear system with input u(t) and output y(t) such that Assumptions A1, A2 and A4 are satisfied. Assume that the input signal is such that the OE-LTI-SOE is regular. Assume also that

$$\Phi_{\eta_0}(z) = \Phi_y(z) - G_{0,OE}(z)\Phi_u(z)G_{0,OE}(z^{-1})$$

from (5.15b) has a canonical spectral factorization

$$\Phi_{\eta_0}(z) = L_{\eta_0}(z)r_{\eta_0}L_{\eta_0}(z^{-1})$$

with  $r_{\eta_0} > 0$ . Then it follows that  $\varepsilon_0(t)$  (see (4.47)) has the following properties

$$\Phi_{\varepsilon_0 u}(z) = 0, \tag{5.23a}$$

$$\Phi_{\varepsilon_0}(z) = r_{\eta_0}.\tag{5.23b}$$

**Proof:** The fact that  $G_{0,GE}(z) = G_{0,OE}(z)$  according to Theorem 5.4 implies that (4.47) can be rewritten as

$$\varepsilon_0(t) = H_{0,GE}^{-1}(q)(y(t) - G_{0,GE}(q)u(t))$$
  
=  $H_{0,GE}^{-1}(q)(y(t) - G_{0,OE}(q)u(t)) = H_{0,GE}^{-1}(q)\eta_0(t).$ 

Using Lemmas 2.1 and 5.4, this gives

$$\begin{split} \Phi_{\varepsilon_0 u}(z) &= H_{0,GE}^{-1}(z) \Phi_{\eta_0 u}(z) = 0, \\ \Phi_{\varepsilon_0}(z) &= H_{0,GE}^{-1}(z) \Phi_{\eta_0}(z) H_{0,GE}^{-1}(z^{-1}) = r_{\eta_0} \end{split}$$

where (5.21) and (5.22b) have been used in the last equality. The results in (5.23) have thus been shown.  $\hfill\square$ 

Corollary 5.1 shows that residual analysis without any further considerations can be used as a validation method also for GE-LTI-SOEs when the corresponding OE-LTI-SOEs are regular since there will be no spurious correlations between u(t) and past  $\varepsilon_0(t - k)$ . In addition, this shows that there will be no need for a closed-loop interpretation of the GE-LTI-SOE in this case.

In this chapter, the basic properties of OE-LTI-SOEs have been discussed. It has been shown that an OE-LTI-SOE is independent of output additive noise that is uncorrelated with the input and the noise-free output. Furthermore, it has been shown that the OE-LTI-SOE often depends only on the odd nonlinearities in a system and that it can be viewed as a linear operator. A class of input signals guaranteeing that the OE-LTI-SOE will be regular, and some properties that the OE-LTI-SOE and GE-LTI-SOE exhibit in this case have also been discussed. In the next chapter, we will turn our attention to NFIR systems with separable inputs.

# 6

## **NFIR Systems with Separable Inputs**

Nonlinear finite impulse response (NFIR) systems are systems that can be written as

$$y(t) = f((u(t-k))_{k=0}^{M}) + w(t)$$
(6.1)

for some  $M \in \mathbb{N}$ . The important special case of a static nonlinearity is obtained for M = 0. Subsystems consisting of static nonlinearities are common in applications. For example, many control systems contain various types of sensor and input nonlinearities that often can be described as static. Furthermore, many nonlinear systems can be approximated by NFIR models. However, we will not discuss such nonlinear approximations here but instead consider how an NFIR system can be approximated by an LTI model.

Intuitively, the natural LTI approximation of an NFIR system is an FIR model. However, the mean-square error optimal LTI approximation, i.e., the OE-LTI-SOE, of such a system will in general be an LTI system with an infinite impulse response. This might not be a problem if the impulse response length M of the NFIR system is known, since it is always possible to estimate an FIR model with the same impulse response length in this case. Although this model might not be the optimal LTI model, it will at least have a structure that probably can be viewed as reasonable compared to the structure of the nonlinear system.

However, in the more realistic case that M is unknown, the structure of the OE-LTI-SOE becomes important. If an NFIR system with impulse response length M has an OE-LTI-SOE that is an FIR model with impulse response length M, it will be rather easy to find an appropriate linear FIR model of this system. When the number of measurements tends to infinity, the parameters of a chosen FIR model will converge to the parameter values given by Theorem 4.2. The problem of finding the impulse response length Mof the NFIR system can thus be solved by estimating linear FIR models with different impulse response lengths. If too large an impulse response length is chosen in the model, the parameters that correspond to the extra terms in the impulse response will simply approach zero asymptotically, just as if the NFIR system would have been a linear FIR system. Hence, it is possible to estimate M without more effort than if the true system would have been linear.

On the other hand, if an NFIR system with impulse response length M has an OE-LTI-SOE with an infinite impulse response length, it will be impossible to estimate Musing only linear approximations. In this case, an increase of the impulse response length in an estimated FIR model will reduce the variance of the model residuals and make the model a better approximation of the OE-LTI-SOE according to Theorem 4.2. Since the OE-LTI-SOE has an infinite impulse response, no information about M can be derived from the FIR approximations of it.

With the previous discussion in mind, it seems that it often should be desirable to preserve the finite impulse response property when an NFIR system is approximated by its OE-LTI-SOE. This is the main topic of this chapter and it turns out that this issue is related to the notion of separable processes. An overview of the theory for separable processes was given in Section 2.4. In this chapter, it will be shown that some random multisines are separable. Since random multisines are a common choice of input signal in identification experiments, this is a theoretical result with practical consequences. Here, the usefulness of random multisines will be illustrated in the case of identification of Hammerstein systems. Furthermore, the notion of higher order separability will be introduced and it will be shown that this type of separability gives a necessary and sufficient condition on the input signal for the OE-LTI-SOE of an arbitrary NFIR system to be an FIR model.

## 6.1 Separability

As was shown in (2.31) in Section 2.4, it is easy to show that separability is a sufficient condition for the invariance property

$$R_{yu}(\tau) = b_0 R_u(\tau), \quad \forall \tau \in \mathbb{Z}$$

from (2.30) to hold. Using z-spectra, the invariance property can be written

$$\Phi_{yu}(z) = b_0 \Phi_u(z),$$

and thus, from Corollary 4.1, it follows that the OE-LTI-SOE of a static nonlinearity is regular and equal to the constant  $b_0$ . Hence, for a separable input, the OE-LTI-SOE of a static nonlinearity is static. This result was shown in Nuttall (1958a) and it will be generalized to NFIR systems in Section 6.3.

Although a number of separable processes can be found in literature (Nuttall, 1958a; McGraw and Wagner, 1968), it seems that the problem of how to construct separable processes has not been extensively studied. The generation of a particular nontrivial separable process is described in the following example.

– Example 6.1 –

Consider a process u defined as

$$u(t) = e(t) + e(t-1),$$

where e is a white process with exponential distribution over the interval  $[-1, +\infty)$  such that E(e(t)) = 0 and  $E(e(t)^2) = 1$ . These properties follow if each random variable e(t) has the probability density function  $p(x) = e^{-(x+1)}$  for  $x \ge -1$ .

Since the process e is white,  $u(t + \tau)$  and u(t) are independent if  $|\tau| > 1$ . Hence,  $E(u(t + \tau)|u(t)) = 0$  for  $|\tau| > 1$ . Furthermore, we have that

$$\begin{split} \mathrm{E}(u(t+1)|u(t)) &= \underbrace{\mathrm{E}(e(t+1)|e(t)+e(t-1))}_{=0} + \mathrm{E}(e(t)|e(t)+e(t-1)) \\ &= \mathrm{E}(e(t)|e(t)+e(t-1)), \\ \mathrm{E}(u(t-1)|u(t)) &= \mathrm{E}(e(t-1)|e(t)+e(t-1)) + \underbrace{\mathrm{E}(e(t-2)|e(t)+e(t-1))}_{=0} \\ &= \mathrm{E}(e(t-1)|e(t)+e(t-1)). \end{split}$$

From these expressions we see that u is separable if e is such that

$$E(e(t)|e(t) + e(t-1) = c) = E(e(t-1)|e(t) + e(t-1) = c) = b \cdot c$$
(6.2)

for some constant b that does not depend on c. We will now show that these equalities hold.

Let X and Y be two independent random variables with probability density functions

$$p_X(x) = \begin{cases} e^{-(x+1)}, & \text{if } x \ge -1, \\ 0, & \text{if } x < -1 \end{cases}$$

and

$$p_Y(y) = \begin{cases} e^{-(y+1)}, & \text{if } y \ge -1, \\ 0, & \text{if } y < -1, \end{cases}$$

and let W = X + Y. Then the joint probability density function for X and W is

$$p_{X,W}(x,w) = p_X(x)p_Y(w-x) = \begin{cases} e^{-(w+2)}, & \text{if } -1 \le x \le w+1, \\ 0, & \text{otherwise.} \end{cases}$$

For  $w \geq -2$ , it follows that

$$p_W(w) = \int_{-1}^{w+1} p_{X,W}(x,w) \, dx = \int_{-1}^{w+1} e^{-(w+2)} \, dx = (w+2)e^{-(w+2)}$$

such that

$$p_W(w) = \begin{cases} (w+2)e^{-(w+2)}, & \text{if } w \ge -2, \\ 0, & \text{if } w < -2. \end{cases}$$

This gives

$$p_{X|W=c}(x) = \begin{cases} \frac{p_{X,W}(x,c)}{p_{W}(c)} = \frac{1}{c+2}, & \text{if } -1 \le x < c+1, \\ 0, & \text{otherwise} \end{cases}$$

and

$$E(X|W=c) = \int_{-1}^{c+1} \frac{x}{c+2} \, dx = \frac{c}{2}.$$
(6.3)

Replacing W with e(t) + e(t - 1) and X with either e(t) or e(t - 1) in (6.3) shows that (6.2) holds with b = 1/2. Hence,

$$E(u(t+1)|u(t) = c) = E(u(t-1)|u(t) = c) = \frac{c}{2}$$

and u is thus separable.

In the previous example, a separable process was obtained by passing a white noise process through a linear FIR filter, i.e., by linear filtering of a separable process. Alternatively, this example can be viewed as an example where the sum of two *dependent* separable processes is separable. Just like the sum of two *independent* separable processes is not separable in general (cf. Section 2.4 or Nuttall (1958a)), linear filtering of a separable process will often result in a nonseparable process. For example, the input signal in Example 4.2 is not separable since it gives a nonstatic OE-LTI-SOE of a static nonlinear system. However, this input signal was generated by linear filtering of a white, and thus separable, signal.

The observation that separability often is lost by linear filtering is an important result that must be considered when a separable process is used in an identification experiment. For example, most of the separable processes in Nuttall (1958a) will not be separable after they have passed through a linear subsystem. However, Gaussian processes, which are a subset of the class of separable processes, have the advantage that the Gaussianity, and thus the separability, will not be affected by linear filtering. This property makes these processes particularly useful for some types of identification problems, as will be described later in Chapter 7.

Separable processes are useful not only for LTI approximations of static nonlinearities but also for LTI approximations of Hammerstein systems. As has been mentioned previously in Section 3.2.2, the initial identification of the LTI part of a Hammerstein system without considering the nonlinearity is an important step in many methods. In the following theorem, it is shown that a scaled version of the LTI subsystem is an OE-LTI-SOE of the system if the input is separable.

#### Theorem 6.1

Consider a Hammerstein system

$$y(t) = G_L(q)v(t) + w(t),$$
 (6.4a)

$$v(t) = f(u(t)), \tag{6.4b}$$

where  $G_L(q)$  is a stable and causal LTI system and where w(t) is measurement noise. Assume that Assumptions A1 and A2 hold if the input is nonperiodic and that Assumption A3 holds if it is periodic. Furthermore, assume that the input is separable and that Assumption A5 holds. Then  $b_0G_L(q)$  with  $b_0 = E(f(u(t))u(t))/R_u(0)$  is an OE-LTI-SOE of this system.

**Proof:** The invariance property

$$R_{vu}(\tau) = \mathbf{E}\big(f(u(t))u(t-\tau)\big) = \mathbf{E}\big(f(u(t))\mathbf{E}\big(u(t-\tau)|u(t)\big)\big)$$
$$= \frac{R_u(\tau)}{R_u(0)}\mathbf{E}\big(f(u(t))u(t)\big) = b_0R_u(\tau)$$
(6.5)

holds since u(t) is separable. Here, Lemma 2.2 has been used in the third equality. The condition that  $G_{0,OE}$  should minimize

$$\mathbf{E}\big((y(t) - G(q)u(t))^2\big)$$

is equivalent to  $G_{0,OE}$  satisfying the Wiener-Hopf condition

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} g_{0,OE}(k) R_u(\tau - k) = 0, \quad \tau \ge 0.$$
(6.6)

From the system description (6.4a), we have

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} g_L(k) R_{vu}(\tau - k) = 0, \quad \forall \tau \in \mathbb{Z}$$

and inserting (6.5) gives

$$R_{yu}(\tau) - \sum_{k=0}^{\infty} b_0 g_L(k) R_u(\tau - k) = 0, \quad \forall \tau \in \mathbb{Z}.$$

Hence,  $b_0G_L(q)$  is an OE-LTI-SOE of the system.

Note that for a nonperiodic input and output satisfying Assumptions A1 and A2, the model  $b_0G_L(q)$  is the unique OE-LTI-SOE of the system.

## 6.2 Separable Random Multisines

Although the separability of a single random phase sinusoid was shown in Nuttall (1958a), there does not seem to be any results about separability of random multisines in literature. However, it turns out that such signals are separable if all amplitudes are constant and equal. This result is proven in the following lemma.

#### Lemma 6.1

A random multisine

$$u(t) = \sum_{k=1}^{Q} A_k \cos(\omega_k t + \psi_k),$$

where all  $A_k$  are constants,  $A_k = \overline{A}$ , and all  $\psi_k$  are independent random variables with uniform distribution on the interval  $[0, 2\pi)$ , is separable.

**Proof:** The signals

$$u_k(t) = A_k \cos(\omega_k t + \psi_k), \quad k = 1, 2, \dots, Q$$

are independent and

$$f_{u_k,1}(\xi_1) = f_{u_l,1}(\xi_1) = J_0(\bar{A}\xi_1),$$
  
$$\sigma_k^2 = \sigma_l^2 = \frac{\bar{A}^2}{2}$$

for all  $k, l \in \{1, 2, ..., Q\}$  from Lemma 2.3. Hence, u(t) is separable according to Theorem 2.2.

Besides the fact that the separability of random multisines is theoretically interesting, it gives a mathematical explanation to why these signals have turned out to be so useful for identification of Hammerstein systems in previous works (Crama and Schoukens, 2001; Crama et al., 2004; Crama and Schoukens, 2004, 2005).

For a separable random multisine, Theorem 6.1 shows that the LTI part of the system can be estimated without compensating for or estimating the nonlinearity at the input of the system. Hence, the model that minimizes the cost function  $V_{N_E,N}$  defined in (2.8) will usually be a good approximation of the LTI subsystem if the number of experiments  $N_E$  is large. Furthermore, Theorem 6.1 shows that the number of excited frequencies does not affect the fact that the frequency response of the LTI subsystem can be estimated consistently at these frequencies without considering the nonlinearity. This result is verified numerically in the following example.

#### -Example 6.2

Consider the Hammerstein system

$$y(t) = G_L(q)v(t) = \frac{1.6 - 1.6q^{-1} + 0.4q^{-2}}{1 - 1.56q^{-1} + 0.96q^{-2}}v(t),$$
(6.7a)

$$v(t) = f(u(t)) = u(t)^3,$$
 (6.7b)

with the input

$$u_1(t) = \sum_{k=1}^{6} \cos(\omega_k t + \psi_k), \tag{6.8}$$

where  $\omega_k = 2\pi k/40$  and where  $\psi_k$  are independent random variables with uniform distribution on the interval  $[0, 2\pi)$ . 500 realizations of the phases have been generated and an input signal with 400 samples has been constructed for each realization. For each input, an identification experiment has been performed where the last periods (40 samples) of the input and output signals have been collected.

Based on the 500 data sets, each consisting of 40 input and output measurements, a nonparametric frequency response estimate  $\hat{G}(e^{i\omega_k})$  has been computed using the least-squares solution (2.13) that minimizes the cost function  $V_{N_E,N}$  in (2.8). A scaled version of this estimate is shown in Figure 6.1 together with the linear part of the system. As can be seen in this figure, the nonparametric frequency response estimate is very close to being a scaled version of  $G_L(e^{i\omega})$ . Actually, the relative errors

$$\rho_k = \frac{|\hat{G}(e^{i\omega_k})/\hat{c}_0 - G_L(e^{i\omega_k})|}{|G_L(e^{i\omega_k})|}, \quad k = 1, 2, \dots, 6,$$

where

$$\hat{c}_0 = \frac{1}{6} \sum_{k=1}^{6} \frac{|\hat{G}(e^{i\omega_k})|}{|G_L(e^{i\omega_k})|},$$

are less than 4% here. An identical identification experiment has been performed with the input

$$u_2(t) = \sum_{k=1}^{6} 2^{5-k} \cos(\omega_k t + \psi_k).$$


**Figure 6.1:** The frequency response of the linear part  $G_L(q)$  of the Hammerstein system from Example 6.2 (solid line) and a scaled version of the nonparametric frequency response estimate (circles).

However, in this case the relative errors were 27%, 5%, 12%, 12%, 12% and 17%, respectively, i.e., significantly larger than for the separable random multisine. Furthermore, for  $u_2(t)$ , the relative errors do not decrease if the number of phase realizations is increased. Hence, it seems that for this input, the best LTI approximation is not a scaled version of  $G_L(q)$ .

Example 6.2 illustrates the main benefit of using separable random multisines for Hammerstein system identification since it shows that for such an input signal, it is rather easy to obtain good estimates of the LTI part of the system. Furthermore, the example indicates that not all random multisines are separable.

#### 6.3 Higher Order Separability

Since separability of an input signal is such a useful property for the analysis of systems with static nonlinearities, it is interesting to investigate if this notion can be generalized. It turns out that it is actually rather straightforward to define a more general type of separability.

#### 6.3.1 Definition and Basic Properties

As a matter of fact, the type of separability in Definition 2.6 should be denoted separability of order one to distinguish it from a more general type of separability. However, in case the order of separability is obvious from the context, we will continue to say that a process is separable whenever there is no risk for ambiguities.

Separability of a process means that certain conditional expectations are linear. A more general version of the previous definition of separability can be written as follows.

**Definition 6.1 (Separability of order** M + 1). Consider an integer  $M \ge 0$  and a stationary stochastic process u with zero mean. This process is separable of order M + 1

if

$$\mathbb{E}(u(t-\tau)|u(t), u(t-1)\dots, u(t-M)) = \sum_{i=0}^{M} a_{\tau,i} u(t-i), \quad \forall \tau \in \mathbb{Z},$$
(6.9)

i.e., if the conditional expectation is linear in u.

Obviously, the previous definition of first order separability is a special case of Definition 6.1.

As was mentioned in Chapter 2, the notion of separability of order one is discussed in detail in Nuttall (1958a) and it is also mentioned briefly (on page 55) that this notion might be extended to separability of higher orders by considering integrals like

$$\int_{-\infty}^{\infty} x_t p(x_t, x_{t-\tau_1}, x_{t-\tau_2}) \, dx_t.$$

However, no further conclusions are drawn in Nuttall (1958a) and to the author's knowledge, no such extension has been made elsewhere.

It is not obvious that all first order separable processes are separable of higher orders as well. However, there are at least a few such cases. For example, since (6.9) is a wellknown property of Gaussian signals (see, for example, Brockwell and Davis, 1987, p. 64), it immediately follows that such signals are separable of order M + 1 for any  $M \in \mathbb{N}$ . Furthermore, it is easy to see that white, possibly non-Gaussian, signals satisfy (6.9) too.

It is straightforward to show that higher order separability implies lower order separability.

#### Theorem 6.2 (Separability of different orders)

If a process u is separable of order M + 1, it is separable of order  $1, 2, \ldots, M$  too.

Proof: It holds that

$$E(u(t-\tau)|u(t), u(t-1)..., u(t-M+1))$$
  
=  $E_{u(t-M)}(E(u(t-\tau)|u(t), u(t-1)..., u(t-M)))$   
=  $E_{u(t-M)}(\sum_{i=0}^{M} a_{\tau,i}u(t-i)) = \sum_{i=0}^{M-1} a_{\tau,i}u(t-i), \quad \forall \tau \in \mathbb{Z}.$ 

Hence, the process is separable of order M. The result follows by induction.

 $\square$ 

#### 6.3.2 Higher Order Separability and OE-LTI-SOEs

We will here consider NFIR systems (6.1) with input signals u(t) that satisfy the conditions in Assumption A1, i.e., real-valued inputs with zero mean, an exponentially bounded covariance function and a z-spectrum with a canonical spectral factorization. For each choice of such a stochastic process u, let  $D_u$  be a class of Lebesgue integrable functions such that

$$D_u = \{f : \mathbb{R}^{M+1} \to \mathbb{R} \mid \mathbb{E}\left(f((u(t-k))_{k=0}^M)\right) = 0,$$
  
$$\mathbb{E}\left(f((u(t-k))_{k=0}^M)^2\right) < \infty,$$
  
$$R_{yu}(\tau) = \mathbb{E}\left(f((u(t-k))_{k=0}^M)u(t-\tau)\right) \text{ exists } \forall \tau \in \mathbb{Z}\}.$$

This class of functions is a straightforward generalization of the corresponding definition for functions on  $\mathbb{R}$  in (2.32). Note that the conditions in the definition of  $D_u$  are weaker than the related conditions on the output signal in Assumption A2. Here, we will use the notation

$$\mathbf{R}_{U} = \begin{pmatrix} R_{u}(0) & R_{u}(1) & \dots & R_{u}(M) \\ R_{u}(1) & R_{u}(0) & \dots & R_{u}(M-1) \\ \vdots & & \vdots \\ R_{u}(M) & R_{u}(M-1) & \dots & R_{u}(0) \end{pmatrix},$$
(6.10)  
$$\mathbf{R}_{YU} = \begin{pmatrix} R_{yu}(0) & R_{yu}(1) & \dots & R_{yu}(M) \end{pmatrix}^{T}.$$

We will here discuss under which conditions the OE-LTI-SOE of an NFIR system will be an FIR model. In this discussion, we will need the notion of the mean-square error optimal FIR model of a system. The following lemma is a classic result (see, for example, Kailath et al., 2000, Theorems 3.2.1 and 3.2.2) and holds for each fixed choice of u. It is included here for the sake of completeness.

#### Lemma 6.2 (FIR approximation)

Consider an input signal u that fulfills the conditions in Assumption A1 and for which  $\mathbf{R}_U$  is positive definite ( $\mathbf{R}_U \succ 0$ ). Then for each NFIR system f in the corresponding class  $D_u$ , there exists a unique linear FIR model

$$G_{0,FIR}(z) = \sum_{k=0}^{M} \bar{b}_f(k) z^{-k}$$

of length M that is an optimal FIR approximation of length M in the mean-square error sense. This FIR model has parameters

$$\bar{B}_f = \left(\bar{b}_f(0) \quad \bar{b}_f(1) \quad \dots \quad \bar{b}_f(M)\right)^T = \mathbf{R}_U^{-1} \mathbf{R}_{YU}$$
(6.11)

and satisfies

$$R_{yu}(\tau) = \sum_{k=0}^{M} \bar{b}_f(k) R_u(\tau - k), \quad \tau = 0, 1, \dots, M.$$
(6.12)

**Proof:** The parameters  $B_f$  in the optimal FIR model of length M are obtained by minimizing

$$E(\left(f((u(t-k))_{k=0}^{M}) - \sum_{k=0}^{M} b(k)u(t-k)\right)^{2})$$

with respect to B. Differentiating this expression with respect to b(i) gives the following equations for the stationary points of the mean-square error criterion

$$-2\left(R_{yu}(i) - \sum_{k=0}^{M} \bar{b}_f(k)R_u(i-k)\right) = 0, \quad i = 0, 1, \dots, M.$$

This can also be written as

$$\mathbf{R}_U \bar{B}_f = \mathbf{R}_{YU}$$

and (6.11) and (6.12) follows readily from this expression.

From (6.12) we see that  $G_{0,FIR}$  can explain the cross-covariance function  $R_{yu}(\tau)$  for  $\tau = 0, 1, \ldots, M$ . However, sometimes it can actually explain the complete cross-covariance function, i.e.,

$$R_{yu}(\tau) = \sum_{k=0}^{M} \bar{b}_f(k) R_u(\tau - k), \quad \forall \tau \in \mathbb{Z}$$
(6.13)

or, equivalently,

$$\Phi_{yu}(z) = G_{0,FIR}(z)\Phi_u(z).$$

In this case, we know from Corollary 4.1 that  $G_{0,FIR}$  is not only the mean-square error optimal FIR approximation of length M of the system, but also the OE-LTI-SOE of the system. It turns out that this will always be true if the input process is separable of order M + 1 and  $f \in D_u$ .

First, we will here consider noise-free NFIR systems, i.e., nonlinear systems with impulse response lengths  $M \ge 0$  that can be written as

$$y(t) = f((u(t-k))_{k=0}^{M}).$$

We will use the following notation

$$\mathbf{R}_{U,\tau} = \begin{pmatrix} R_u(\tau) & R_u(\tau-1) & \dots & R_u(\tau-M) \end{pmatrix}^T,$$

and we will assume that  $\mathbf{R}_U$  (see (6.10)) is a positive definite matrix ( $\mathbf{R}_U \succ 0$ ) such that the vector

$$C_{\tau} = \begin{pmatrix} c_{\tau,0} & c_{\tau,1} & \dots & c_{\tau,M} \end{pmatrix}^T = \mathbf{R}_U^{-1} \mathbf{R}_{U,\tau}$$
(6.14)

is well-defined.

We will now show that the definition of separability implies that the constants  $a_{\tau,i}$  from Definition 6.1 satisfy  $a_{\tau,i} = c_{\tau,i}$ .

#### Lemma 6.3

Consider a signal u which is separable of order M + 1, i.e., that (6.9) holds for some constants  $a_{\tau,i}$ . Then it holds that

$$A_{\tau} = C_{\tau} = \mathbf{R}_{U}^{-1} \mathbf{R}_{U,\tau},\tag{6.15}$$

where

$$A_{\tau} = \begin{pmatrix} a_{\tau,0} & a_{\tau,1} & \dots & a_{\tau,M} \end{pmatrix}^{T}.$$

**Proof:** For k = 0, 1, ..., M, Definition 6.1 gives

$$R_{u}(\tau - k) = E(u(t - k)u(t - \tau))$$
  
=  $E(E(u(t - k)u(t - \tau)|u(t), u(t - 1)..., u(t - M)))$   
=  $E(u(t - k)E(u(t - \tau)|u(t), u(t - 1)..., u(t - M)))$   
=  $\sum_{i=0}^{M} a_{\tau,i}E(u(t - k)u(t - i)) = \sum_{i=0}^{M} a_{\tau,i}R_{u}(k - i).$ 

The previous expression can also be written as

$$\mathbf{R}_U A_{\tau} = \mathbf{R}_{U,\tau}.$$

This shows that (6.15) holds.

The previous lemma shows that separability of order M + 1 is equivalent to that the property

$$E(u(t-\tau)|u(t), u(t-1)..., u(t-M)) = \sum_{i=0}^{M} c_{\tau,i} u(t-i), \quad \forall \tau \in \mathbb{Z}$$
(6.16)

holds. In the next lemma, we will show that separability of u is a necessary and sufficient condition for the equality (6.13) to hold for all  $\tau \in \mathbb{Z}$  and for all  $f \in D_u$ .

#### Lemma 6.4 (Separability of order M + 1)

Consider a fixed  $M \ge 0$  and a certain choice of input signal u that fulfills the conditions in Assumption A1, and for which  $\mathbf{R}_U \succ 0$  and  $\mathrm{E}(|u(t)|) < \infty$ . Let  $\bar{B}_f$  denote the parameters of the mean-square error optimal FIR approximation of length M of each  $f \in D_u$ , i.e.,  $\bar{B}_f = \mathbf{R}_U^{-1} \mathbf{R}_{YU}$  according to Lemma 6.2. Then

$$R_{yu}(\tau) = \sum_{k=0}^{M} \bar{b}_f(k) R_u(\tau - k), \quad \forall \tau \in \mathbb{Z} \text{ and } \forall f \in D_u$$
(6.17)

if and only if u is separable of order M + 1.

**Proof:** Using (6.14) and (6.11), it follows that

$$\sum_{k=0}^{M} \bar{b}_f(k) R_u(\tau - k) = \bar{B}_f^T \mathbf{R}_{U,\tau} = \mathbf{R}_{YU}^T C_\tau = \sum_{i=0}^{M} c_{\tau,i} R_{yu}(i).$$
(6.18)

IF: Assume that u is separable of order M + 1, i.e., that (6.16) holds. By the construction of  $\overline{B}_f$ , the equality (6.17) already holds for  $\tau = 0, 1, \ldots, M$  for all  $f \in D_u$  (cf. (6.11)). Take an arbitrary  $f \in D_u$  and let  $y(t) = f((u(t-k))_{k=0}^M)$ . Furthermore, take an arbitrary  $\tau > M$  or  $\tau < 0$ . Then it follows that

$$R_{yu}(\tau) = E(y(t)u(t-\tau)) = E(E(y(t)u(t-\tau)|u(t), u(t-1), \dots, u(t-M)))$$
  
=  $E(y(t)E(u(t-\tau)|u(t), u(t-1), \dots, u(t-M)))$   
=  $\sum_{i=0}^{M} c_{\tau,i}E(y(t)u(t-i)) = \sum_{i=0}^{M} c_{\tau,i}R_{yu}(i) = \sum_{k=0}^{M} \bar{b}_{f}(k)R_{u}(\tau-k),$ 

where the third equality follows from the fact that y(t) depends only on  $u(t), u(t - 1), \ldots, u(t - M)$  while the fourth equality follows from (6.16) and the last from (6.18). Since both f and  $\tau$  were arbitrary, (6.17) holds for all  $\tau \in \mathbb{Z}$  and for all  $f \in D_u$ .

ONLY IF: Assume that (6.17) holds for a particular u. Take an arbitrary  $\tau > M$  or  $\tau < 0$ . Using (6.18), (6.17) gives the equality

$$\int_{\mathbb{R}^{M+1}} f(x_t, \dots, x_{t-M}) \left( \int_{-\infty}^{\infty} x_{t-\tau} p_{\tau}(x_t, \dots, x_{t-M}, x_{t-\tau}) \, dx_{t-\tau} - \sum_{i=0}^{M} c_{\tau,i} x_{t-i} p(x_t, \dots, x_{t-M}) \right) dx_t \dots dx_{t-M} = 0, \quad \forall f \in D_u,$$

$$(6.19)$$

where p and  $p_{\tau}$  are the joint probability density functions of

$$(u(t), u(t-1), \dots, u(t-M))^T$$

and

$$(u(t), u(t-1), \dots, u(t-M), u(t-\tau))^T$$
,

respectively. Let

$$v_{\tau}(x_{t}, \dots, x_{t-M}) = \int_{-\infty}^{\infty} x_{t-\tau} p_{\tau}(x_{t}, \dots, x_{t-M}, x_{t-\tau}) \, dx_{t-\tau}$$
$$- \sum_{i=0}^{M} c_{\tau,i} x_{t-i} p(x_{t}, \dots, x_{t-M})$$

and define a function

$$f_0(x_t,\ldots,x_{t-M}) = \operatorname{sign}(v_\tau(x_t,\ldots,x_{t-M})) - \mu_0,$$

where

$$\mu_0 = \mathrm{E}\left(\mathrm{sign}\left(v_\tau((u(t-k))_{k=0}^M)\right)\right).$$

Since

$$\mathbb{E}(f_0((u(t-k))_{k=0}^M)) = 0, \\ \mathbb{E}(f_0((u(t-k))_{k=0}^M)^2) < \infty$$

and

$$\begin{aligned} |\mathbf{E} \big( f_0((u(t-k))_{k=0}^M) u(t-\sigma) \big)| \\ &= |\mathbf{E} \big( \operatorname{sign} \big( v_\tau((u(t-k))_{k=0}^M) \big) u(t-\sigma) \big) - \mu_0 \underbrace{\mathbf{E}(u(t-\sigma))}_{=0} | \\ &= |\mathbf{E} \big( \operatorname{sign} \big( v_\tau((u(t-k))_{k=0}^M) \big) u(t-\sigma) \big) | \\ &\leq \mathbf{E} \big( |\operatorname{sign} \big( v_\tau((u(t-k))_{k=0}^M) \big) u(t-\sigma) | \big) \\ &\leq \mathbf{E} (|u(t-\sigma)|) < \infty, \quad \forall \sigma \in \mathbb{Z}, \end{aligned}$$

it follows that  $f_0 \in D_u$ . Hence, (6.19) holds for  $f = f_0$  and this implies that

$$\int_{\mathbb{R}^{M+1}} |v_{\tau}(x_t, \dots, x_{t-M})| dx_t \dots dx_{t-M}$$
$$-\mu_0 \underbrace{\mathbf{E}(u(t-\tau))}_{=0} + \mu_0 \sum_{i=0}^M c_{\tau,i} \underbrace{\mathbf{E}(u(t-i))}_{=0} = 0$$
$$\Rightarrow \int_{\mathbb{R}^{M+1}} |v_{\tau}(x_t, \dots, x_{t-M})| dx_t \dots dx_{t-M} = 0$$
$$\Rightarrow v_{\tau}(x_t, \dots, x_{t-M}) = 0 \quad \text{almost everywhere.}$$

The conditional probability density function of  $u(t - \tau)$  given  $u(t) = x_t, u(t - 1) = x_{t-1}, \ldots, u(t - M) = x_{t-M}$  is

$$p_{\tau,c}(x_{t-\tau}) = \frac{p_{\tau}(x_t, \dots, x_{t-M}, x_{t-\tau})}{p(x_t, \dots, x_{t-M})}$$

if  $p(x_t, \ldots, x_{t-M}) > 0$ . Hence, the fact that

$$v_{\tau}(x_t,\ldots,x_{t-M})=0$$

implies that

$$\int_{-\infty}^{\infty} x_{t-\tau} p_{\tau,c}(x_{t-\tau}) \, dx_{t-\tau} = \sum_{i=0}^{M} c_{\tau,i} x_{t-i}$$

or, equivalently, that (6.16) holds for the chosen  $\tau$ . Since  $\tau$  was arbitrary, (6.16) follows and u is thus separable of order M + 1.

Lemma 6.4 is an extension of the corresponding theorem about separability of order one in Nuttall (1958a). Just like this classic result can be used to study linear approximations of static nonlinearities, Lemma 6.4 can be used to characterize exactly for which input signals the OE-LTI-SOE of an arbitrary NFIR system will be an FIR model.

Besides the results about mean-square error optimal stable and causal LTI predictors, which here have been used to define OE-LTI-SOEs, classic Wiener filtering theory also contains results about mean-square error optimal stable noncausal LTI predictors, usually known as *Wiener smoothers* or noncausal Wiener filters (see, for example, Kailath et al., 2000, Theorem 7.3.1). For a nonlinear system with input u and output y that satisfy Assumptions A1 and A2, these results show that the best, in mean-square error sense, stable but possibly noncausal LTI approximation of this system is given by the ratio

$$\frac{\Phi_{yu}(z)}{\Phi_u(z)}.\tag{6.20}$$

A simple example of a causal nonlinear system with an input such that (6.20) becomes noncausal can be found in Forssell and Ljung (2000). However, for a separable input, this cannot happen since the following result holds.

#### Theorem 6.3

Consider a fixed  $M \ge 0$  and a certain input signal u that fulfills the conditions in Assumption A1, and for which  $\mathbf{R}_U \succ 0$  and  $\mathrm{E}(|u(t)|) < \infty$ . Consider NFIR systems

$$y(t) = y_{nf}(t) + w(t) = f((u(t-k))_{k=0}^{M}) + w(t),$$

where the noise w(t) is such that Assumption A5 is fulfilled for all f. Then the OE-LTI-SOE of such a system will be well-defined and equal to a linear FIR model

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)} = \sum_{k=0}^M \bar{b}_f(k) z^{-k},$$
(6.21)

where  $\bar{B}_f = \mathbf{R}_U^{-1} \mathbf{R}_{YU}$  for all  $f \in D_u$ , if and only if u is separable of order M + 1.

**Proof:** Assumption A5 gives  $\Phi_{yu}(z) = \Phi_{y_{nf}u}(z)$ . Hence, the OE-LTI-SOE is not influenced by the noise term w(t). Since the input satisfies the conditions in Lemma 6.4, we have that

$$\Phi_{y_{nf}u}(z) = \sum_{k=0}^{M} \bar{b}_f(k) z^{-k} \Phi_u(z), \qquad (6.22)$$

where  $\bar{B}_f = \mathbf{R}_U^{-1} \mathbf{R}_{YU}$  for all  $f \in D_u$ , if and only if u is separable of order M + 1. If (6.22) holds for all  $f \in D_u$ , the NFIR systems that correspond to these functions have outputs that satisfy Assumption A2. Hence, the OE-LTI-SOEs of these NFIR systems are well-defined and Corollary 4.1 can be applied to show that

$$G_{0,OE}(z) = \frac{\Phi_{y_{nf}u}(z)}{\Phi_u(z)} = \sum_{k=0}^{M} \bar{b}_f(k) z^{-k}$$

for all  $f \in D_u$ . The theorem has thus been shown.

Theorem 6.3 shows that separability of order M + 1 is a necessary and sufficient condition for the OE-LTI-SOE to be equal to an FIR model of length M for all NFIR systems defined by functions in  $D_u$ . Furthermore, this theorem shows that even if we consider noncausal LTI models, a separable input will give an optimal model that is a causal FIR model.

In many cases, it is possible to shed some light on a theoretical result by interpreting it in a geometrical framework. This can be done also in our case. For a fixed t, we can view the output y(t) and the components of the input signal  $u(\tau)$ ,  $\tau \in \mathbb{Z}$  as vectors in an infinite dimensional inner-product space with the inner product  $\langle u, v \rangle = E(uv)$  (see Brockwell and Davis, 1987).

The output from the OE-LTI-SOE of the NFIR system will in this framework be the orthogonal projection of y(t) into the linear subspace that is spanned by

$$u(t), u(t-1), \ldots, u(t-\infty).$$

From (6.21) we can draw the conclusion that this projection actually lies in the finite dimensional linear subspace that is spanned by

$$u(t), u(t-1), \ldots, u(t-M)$$

if u is separable. This is a compact and intuitive way to describe the theoretical results presented previously in this section.

#### 6.3.3 Identification of Generalized Hammerstein Systems

Just like first order separability can be used when a Hammerstein system is approximated by an LTI model, higher order separability can be used to characterize the OE-LTI-SOE of a more general class of systems that we will call generalized Hammerstein systems. These systems consist of an NFIR system followed by an LTI system and the OE-LTI-SOE of such a system with a separable input is described in the following theorem.

#### Theorem 6.4

Consider a generalized Hammerstein system

$$y(t) = G_L(q)v(t) + w(t),$$
 (6.23a)

$$v(t) = f((u(t-k))_{k=0}^{M}),$$
 (6.23b)

where  $G_L(q)$  is a stable and causal LTI system and where w(t) is measurement noise. Assume that the input to this system is separable of order M+1 and that Assumptions A1, A2 and A5 hold. Then

$$G_{0,OE}(q) = G_L(q)B(q),$$
 (6.24)

where

$$B(q) = \sum_{k=0}^{M} \bar{b}_f(k) q^{-k}$$

and  $\bar{B}_f = \mathbf{R}_U^{-1} \mathbf{R}_{VU}$ .

**Proof:** Assumption A5 implies that

$$\Phi_{yu}(z) = G_L(z)\Phi_{vu}(z)$$

and Lemma 6.4 gives

$$\Phi_{vu}(z) = B(z)\Phi_u(z).$$

Hence,

$$\Phi_{yu}(z) = G_L(z)B(z)\Phi_u(z)$$

and (6.24) follows from Corollary 4.1.

Since B(q) is an FIR model, the previous theorem shows that the denominator of  $G_L(q)$  can be consistently estimated without compensating for the NFIR subsystem.

The focus of this chapter has been on FIR approximations of NFIR systems. The separability of a class of random multisines has been shown. Furthermore, it has been shown that separability of order M + 1 is a necessary and sufficient condition on the input signal for the OE-LTI-SOE of any NFIR system in a rather wide class of systems to be an FIR model. In the next chapter, NFIR systems with Gaussian inputs will be studied and it will be shown that a couple of additional useful properties hold for Gaussian inputs.

7

### **NFIR Systems with Gaussian Inputs**

Random variables and processes with Gaussian distributions play a rather special role in probability theory and applications. They exhibit a large number of properties that simplify many statistical problems or make the solutions to the problems more general. One interesting property that holds if two random variables have a simultaneous Gaussian distribution, is that these variables are independent if and only if they are uncorrelated.

This property implies that a Gaussian process u, whose z-spectrum has a canonical spectral factorization, can always be viewed as if it has been generated by filtering white Gaussian noise through a minimum phase LTI filter. This results follows from the fact that any signal can be viewed as if it has been generated by filtering an *uncorrelated* signal with a minimum phase filter. Hence, the OE-LTI-SOE of an arbitrary causal nonlinear system with a Gaussian input will be

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)}$$

provided that the input and output signals fulfill the conditions in Theorem 5.2. This implies that all properties from Sections 5.5 and 5.6 will hold also for Gaussian input signals. However, in this chapter we will show that OE-LTI-SOEs for Gaussian inputs also have properties that other minimum phase generated inputs do not have.

For example, a Gaussian process is separable of any order. This observation was made in the previous chapter and implies that for an NFIR system with a Gaussian input, the cross-covariance function between y and u can always be written as in (6.13). This result is a kind of generalization of Bussgang's theorem (Theorem 3.1) to NFIR systems. The reason why (6.13) is not a proper generalization of Bussgang's theorem is that it is not obvious that the coefficients  $\bar{b}_f(k)$  can be calculated as expectations of derivatives of fin the same way as  $b_0 = E(f'(u(t)))$  in Theorem 3.1. In the next section, a generalized version of Bussgang's theorem will be shown.

This result implies that the OE-LTI-SOEs of NFIR systems with Gaussian inputs al-

ways will be linear FIR models and that the coefficients in these models will have a natural definition in terms of properties of the true system. This relation between the FIR model parameters and the true system and the fact that, unlike separability, Gaussianity of a process is preserved under linear filtering turn out to be useful in some applications.

It will be shown later in this chapter that these properties of Gaussian signals can be used for structure identification of NFIR systems and for identification of classes of systems that here will be called *generalized Wiener-Hammerstein systems*. However, first we will show the generalization of Bussgang's classic theorem.

#### 7.1 OE-LTI-SOEs of NFIR Systems with Gaussian Input Processes

The generalization of Bussgang's theorem to NFIR systems can be found in, for example, Scarano et al. (1993) and has also previously been used in the research area of stochastic mechanical vibrations (see, for example, Atalik and Utku, 1976; Lutes and Sarkani, 1997). We will however restate the result here under the following technical assumptions.

Assumption A6. Assume that the real-valued functions f(x) and  $p(\tilde{x})$ , where  $x \in \mathbb{R}^N$ and  $\tilde{x} = (x^T, x_{N+1})^T \in \mathbb{R}^{N+1}$ , are such that  $f \cdot p$ ,  $f'_{x_i} \cdot p$  and  $f \cdot \tilde{x}_i \cdot p$ ,  $i = 1, \ldots, (N+1)$ all belong to  $L^1(\mathbb{R}^{N+1})$  and that  $f(x)p(\tilde{x}) \to 0$  when  $|\tilde{x}| \to +\infty$ . (Here,  $f'_{x_i}$  is the partial derivative of f with respect to  $x_i$ ).

**Assumption A7.** Consider two stationary stochastic processes u and y such that  $y(t) = f((u(t-k))_{k=0}^{M})$ . Assume that u is a Gaussian process with zero mean and that E(y(t)) = 0. Form random vectors

$$\omega_{\sigma} = (u(t), u(t-1), \dots, u(t-M), u(t-\sigma))^{T}$$
(7.1)

with  $\sigma < 0$  or  $\sigma > M$ . Let  $P_{\sigma}$  and  $p_{\sigma}$  denote the covariance matrices and joint probability density functions of these vectors, respectively. Assume that det  $P_{\sigma} \neq 0$  and that f and  $p_{\sigma}$  satisfy Assumption A6 for all  $\sigma < 0$  or  $\sigma > M$ .

Assumptions A6 and A7 assure that the input is Gaussian and that the function f(x) does not grow too fast. Assumption A6 holds if, for example, f is a polynomial and p is a Gaussian probability density function. This assumption is used in the following lemma.

Lemma 7.1

Let

$$\tilde{x} = (x^T, x_{N+1})^T = (x_1, x_2, \dots, x_N, x_{N+1})^T$$
(7.2)

be a jointly Gaussian distributed random vector with zero mean and covariance matrix C with det  $C \neq 0$ . Let  $f : \mathbb{R}^N \to \mathbb{R}$  be a differentiable function of x with E(f(x)) = 0 and let p denote the probability density function of  $\tilde{x}$ . Furthermore, assume that f and p satisfy Assumption A6. Then

$$\mathcal{E}(f(x)\tilde{x}) = Cw,\tag{7.3}$$

where

$$w = \begin{pmatrix} \mathcal{E}(f'_{x_1}(x)) \\ \mathcal{E}(f'_{x_2}(x)) \\ \vdots \\ \mathcal{E}(f'_{x_N}(x)) \\ 0 \end{pmatrix}.$$

**Proof:** Factorize C as  $C = \tilde{Q}\tilde{Q}^T$  and define a new stochastic vector z as  $z = \tilde{Q}^{-1}\tilde{x}$ . Then z is jointly normally distributed with zero mean and a covariance matrix that is equal to the identity matrix. Let Q denote the matrix that is obtained from  $\tilde{Q}$  by removing the last row. Then x = Qz and we get

$$\begin{split} \mathbf{E}(f(x)\tilde{x}) &= \tilde{Q}\mathbf{E}(f(x)\tilde{Q}^{-1}\tilde{x}) = \tilde{Q}\mathbf{E}(f(Qz)z) \\ &= \tilde{Q} \begin{pmatrix} \mathbf{E}(\frac{\partial f(Qz)}{\partial z_1}) \\ \mathbf{E}(\frac{\partial f(Qz)}{\partial z_2}) \\ \vdots \\ \mathbf{E}(\frac{\partial f(Qz)}{\partial z_N+1}) \end{pmatrix} = \tilde{Q}\tilde{Q}^T \begin{pmatrix} \mathbf{E}(f'_{x_1}(x)) \\ \mathbf{E}(f'_{x_2}(x)) \\ \vdots \\ \mathbf{E}(f'_{x_N}(x)) \\ 0 \end{pmatrix} = Cw. \end{split}$$

The third equality follows from the fact that  $E(h(z)z_i) = E(h'_{z_i}(z))$  when z has an N(0, I) distribution. This equality holds since

$$\int_{-\infty}^{\infty} g(r) r e^{-r^2/2} dr = \left[ -g(r) e^{-r^2/2} \right]_{r=-\infty}^{\infty} + \int_{-\infty}^{\infty} g'(r) e^{-r^2/2} dr.$$

Furthermore, the fourth equality in the derivation above follows from the chain rule, which can be written here as

$$\frac{\partial f(Qz)}{\partial z_i} = \frac{\partial f(Qz)}{\partial x_1} Q_{1i} + \frac{\partial f(Qz)}{\partial x_2} Q_{2i} + \ldots + \frac{\partial f(Qz)}{\partial x_N} Q_{Ni}.$$

Lemma 7.1 is used in the following generalization of Bussgang's theorem.

#### Theorem 7.1

Let  $y(t) = f((u(t-k))_{k=0}^{M})$  be an NFIR system with a stationary Gaussian process u as input. Assume that u and y satisfy Assumption A7. Then it follows that

$$R_{yu}(\tau) = \sum_{k=0}^{M} b(k) R_u(\tau - k), \quad \forall \tau \in \mathbb{Z},$$
(7.4)

where

$$b(k) = \mathbf{E} \left( f'_{u(t-k)} \left( (u(t-j))_{j=0}^{M} \right) \right).$$

**Proof:** Choose an arbitrary  $\sigma < 0$  or  $\sigma > M$  and let

$$x = (u(t), u(t-1), \dots, u(t-M))^T$$

and  $x_{N+1} = u(t - \sigma)$  in Lemma 7.1. Then Equation (7.3) gives

$$E(y(t)\begin{pmatrix}u(t)\\u(t-1)\\\vdots\\u(t-M)\\u(t-\sigma)\end{pmatrix}) = \begin{pmatrix}R_{u}(0) & R_{u}(1) & \dots & R_{u}(M) & R_{u}(\sigma)\\R_{u}(1) & R_{u}(0) & \dots & R_{u}(M-1) & R_{u}(\sigma-1)\\\vdots & \vdots & \ddots & \vdots & \vdots\\R_{u}(M) & R_{u}(M-1) & \dots & R_{u}(0) & R_{u}(\sigma-M)\\R_{u}(\sigma) & R_{u}(\sigma-1) & \dots & R_{u}(\sigma-M) & R_{u}(0)\end{pmatrix}w,$$
(7.5)

where  $w_{i+1} = \mathbb{E}(f'_{u(t-i)}((u(t-k))_{k=0}^M))$  for  $i = 0, \ldots, M$  and  $w_{M+2} = 0$ . Equation (7.5) can be written more compactly as

$$R_{yu}(\tau) = \sum_{k=0}^{M} b(k) R_u(\tau - k), \quad \tau = 0, 1, \dots, M, \sigma,$$

where  $b(k) = w_{k+1} = \mathbb{E}(f'_{u(t-k)}((u(t-j))_{j=0}^M))$ . As  $\sigma$  was chosen arbitrarily, this relation holds for all  $\tau \in \mathbb{Z}$ .

Using z-transforms, the result (7.4) can also be written as

$$\Phi_{yu}(z) = B(z)\Phi_u(z),\tag{7.6}$$

where  $B(z) = \sum_{k=0}^{M} b(k) z^{-k}$ . This relation can be used to characterize the OE-LTI-SOE of an NFIR system with a Gaussian input. The OE-LTI-SOE is in general obtained by the Wiener filter construction in (4.3). However, from (7.6) we see that the ratio  $\Phi_{yu}(z)/\Phi_u(z)$  is stable and causal if the nonlinear system is an NFIR system with a Gaussian input. Hence, with Corollary 4.1 in mind we can state the following theorem.

#### Theorem 7.2

Consider an NFIR system

$$y(t) = f((u(t-k))_{k=0}^{M}) + w(t)$$

with a Gaussian input u(t) such that Assumptions A1, A2, A5 and A7 are satisfied. Then the OE-LTI-SOE of this system is the linear FIR model

$$G_{0,OE}(z) = \frac{\Phi_{yu}(z)}{\Phi_u(z)} = \sum_{k=0}^M b(k) z^{-k},$$
(7.7)

where

$$b(k) = \mathbb{E}\left(f'_{u(t-k)}\left((u(t-j))_{j=0}^{M}\right)\right).$$
(7.8)

The fact that (7.8) holds for a Gaussian input but not for a general separable input might seem like a minor difference. However, it will be shown in the next section that (7.8) can be rather useful if the purpose of estimating a linear model is to obtain information about the structure of the nonlinear system. In this case, a Gaussian process is a more suitable choice of input signal than a general separable process. Furthermore, Gaussianity of a process is preserved under linear filtering while separability in general is not. An application where this fact is crucial will also be described in the next section.

#### 7.2 Applications

The characterization (7.7) of the OE-LTI-SOE of an NFIR system with a Gaussian input is not only theoretically interesting, but can also be useful in some applications of system identification. We will here briefly discuss two such applied identification problems.

#### 7.2.1 Structure Identification of NFIR Systems

The most obvious application of the result (7.7) is perhaps to use it for guidance when an NFIR system is going to be identified. However, linear models are not useful for all types of NFIR systems. Any NFIR system can be written as a sum of an even and an odd function. Since all Gaussian probability density functions with zero mean are even functions, the OE-LTI-SOE of an NFIR system is only influenced by the odd part of the system (cf. Lemma 5.2).

Hence, we will here only consider odd NFIR systems, i.e., NFIR systems  $y(t) = f((u(t - n_k - j))_{i=0}^M)$  where

$$f((-u(t-n_k-j))_{j=0}^M) = -f((u(t-n_k-j))_{j=0}^M).$$

When such an odd NFIR system is going to be identified, it is in general not obvious how the time delay  $n_k$  and order M should be estimated in an efficient way. However, if the input is Gaussian and sufficiently many measurements can be collected,  $n_k$  and M can both be obtained from an impulse response estimate. Such an estimate can be computed very efficiently by means of the least-squares method.

Furthermore, if only a few of the input terms

$$u(t-n_k), u(t-n_k-1), \dots, u(t-n_k-M)$$

enter the system in a nonlinear way, it might be interesting to know which these terms are. If a nonlinear model of the system is desired, this knowledge can be used to reduce the complexity of the proposed model. A coefficient b(j) in (7.7) will be invariant of the input properties if the corresponding input term u(t - j) only affects the system linearly, while an input term that affects the system in a nonlinear way will have an input dependent *b*-coefficient in (7.7).

This fact makes it possible to extract information about which nonlinear terms are present in the system simply by looking at the differences between FIR models that have been estimated with different Gaussian input signals. The coefficients that correspond to an input term that enters the system in a nonlinear way will be different in these estimates, provided that the covariance functions of the inputs are different. This idea is used in the following example.

Example 7.1 –

Consider the nonlinear system  $y(t) = u(t) + u(t-1)^3$  and assume that the input to this system is Gaussian and such that the conditions in Theorem 7.2 are fulfilled. Then the OE-LTI-SOE of this system will be

$$G_{0,OE}(q) = b(0) + b(1)q^{-1},$$



Figure 7.1: A generalized Wiener-Hammerstein system.

where b(0) = 1 and  $b(1) = 3R_u(0)$ . If the variance of the input is changed, b(1) will change too, while b(0) will remain equal to one. Hence, it is easy to see which input signal component affects y(t) in a nonlinear way.

An overview of other methods for structure identification of nonlinear systems can be found in Haber and Unbehauen (1990).

#### 7.2.2 Identification of Generalized Wiener-Hammerstein Systems

It has been mentioned previously in Section 3.2.2 that Bussgang's theorem has been used to show important results concerning the identification of Hammerstein and Wiener systems (see, for example, Billings and Fakhouri, 1982). In principle, these results state that an estimated LTI model will converge to a scaled version of the linear part of a Hammerstein or Wiener system when the number of measurements tends to infinity, provided that the input is Gaussian. These results simplify the identification of Wiener and Hammerstein systems significantly.

Hence, it is interesting to investigate if the result (7.7) about the OE-LTI-SOEs of NFIR systems can be used to prove similar results for extended classes of systems. In this section, we will study a type of systems that we will call generalized Wiener-Hammerstein systems.

More specifically, we will call a nonlinear system a generalized Wiener-Hammerstein system if it consists of an LTI system  $n(t) = G_1(q)u(t)$  followed by an NFIR system  $v(t) = f((n(t-k))_{k=0}^M)$  followed by an LTI system  $y(t) = G_2(q)v(t)$  as is shown in Figure 7.1. The following corollary to Theorem 7.2 shows that the OE-LTI-SOE of such a system has a certain structure.

#### **Corollary 7.1**

Consider a generalized Wiener-Hammerstein system  $y(t) = G_2(q)v(t) + w(t)$  where  $v(t) = f((n(t-k))_{k=0}^M)$  and  $n(t) = G_1(q)u(t)$  and where  $G_1(q)$  and  $G_2(q)$  are stable and causal LTI systems. Assume that u(t) is Gaussian and that u(t) and y(t) fulfill Assumptions A1, A2 and A5. Assume also that n(t) and v(t) fulfill Assumptions A1, A2 and A7. Then the OE-LTI-SOE of this system is

$$G_{0,OE}(z) = G_2(z)B(z)G_1(z),$$
(7.9)

where  $B(z) = \sum_{k=0}^{M} b(k) z^{-k}$  and

$$b(k) = \mathbf{E} \left( f'_{n(t-k)} \left( (n(t-j))_{j=0}^{M} \right) \right).$$

Proof: We have

$$\Phi_{yu}(z) = G_2(z)\Phi_{vu}(z), \tag{7.10a}$$

$$\Phi_{vn}(z) = \Phi_{vu}(z)G_1(z^{-1}), \tag{7.10b}$$

$$\Phi_n(z) = G_1(z)\Phi_u(z)G_1(z^{-1}).$$
(7.10c)

In addition, Theorem 7.2 gives that

$$\Phi_{vn}(z) = B(z)\Phi_n(z). \tag{7.11}$$

Inserting (7.10b) and (7.10c) in (7.11) gives

$$\Phi_{vu}(z) = B(z)G_1(z)\Phi_u(z), \tag{7.12}$$

and inserting (7.10a) in (7.12) gives

$$\Phi_{yu}(z) = G_2(z)B(z)G_1(z)\Phi_u(z).$$

Hence, (7.9) follows from Corollary 4.1.

Corollary 7.1 shows that the OE-LTI-SOE of a generalized Wiener-Hammerstein system with a Gaussian input will be  $G_2(z)B(z)G_1(z)$ , and hence an estimated output error model will approach this model when the number of measurements tends to infinity. In particular, as B(z) is an FIR model, this shows that the denominator of the estimated model will approach the product of the denominators of  $G_1$  and  $G_2$  if the degree of the model denominator polynomial is correct.

We will thus get consistent estimates of the poles of  $G_1$  and  $G_2$  despite the presence of the NFIR system. This is particularly useful if either  $G_1$  or  $G_2$  is equal to one, i.e., if we have either a generalized Hammerstein or a generalized Wiener system. The consistency of the pole estimates for a generalized Hammerstein system is verified numerically in Example 7.2.

— Example 7.2 —

Consider a generalized Hammerstein system

$$y(t) = G(q)f(u(t), u(t-1)) + w(t)$$

where

$$G(q) = \frac{1}{1 + 0.6q^{-1} + 0.1q^{-2}},$$
  
$$f(u(t), u(t-1)) = \arctan(u(t)) \cdot u(t-1)^2$$

and where w(t) is white Gaussian noise with E(w(t)) = 0 and  $E(w(t)^2) = 1$ .

Let the input u(t) be generated by linear filtering of a white Gaussian process e(t) with E(e(t)) = 0 and  $E(e(t)^2) = 1$  such that

$$u(t) = \frac{1 - 0.8q^{-1} + 0.1q^{-2}}{1 - 0.2q^{-1}}e(t),$$

and assume that e(t) and w(s) are independent for all  $t, s \in \mathbb{Z}$ .

This input signal has been used in an identification experiment where a data set consisting of 100 000 measurements of u(t) and y(t) was collected. The large number of measurements has been chosen since the convergence towards the OE-LTI-SOE might be slow. A linear output error model  $\hat{G}_{OE}$  with  $n_b = n_f = 2$  and  $n_k = 0$  has been estimated from this data set and the result was

$$\hat{G}_{OE}(q) = \frac{0.7481 - 0.7164q^{-1}}{1 + 0.6045q^{-1} + 0.1031q^{-2}}.$$
(7.13)

As can easily be seen from (7.13), the denominator of  $\hat{G}_{OE}(q)$  is indeed close to the denominator of G(q). This is exactly what one would expect as the previous theoretical discussion give that the OE-LTI-SOE of the generalized Hammerstein system is the product between G(q) and an FIR model B(q). The MATLAB code that has been used in this example is available in Appendix B.2.

The following example verifies Corollary 7.1 also for a particular generalized Wiener system.

#### Example 7.3 –

Consider a generalized Wiener system consisting of the same linear and nonlinear blocks as the generalized Hammerstein system in Example 7.2 but with the linear block before the nonlinear, i.e.,

$$y(t) = f(n(t), n(t-1)) + w(t),$$
  
 $n(t) = G(q)u(t),$ 

where

$$G(q) = \frac{1}{1 + 0.6q^{-1} + 0.1q^{-2}},$$
  
$$f(n(t), n(t-1)) = \arctan(n(t)) \cdot n(t-1)^2$$

and where w(t) is white Gaussian noise with E(w(t)) = 0 and  $E(w(t)^2) = 1$ .

Let the input u(t) be generated in the same way as in Example 7.2, i.e.,

$$u(t) = \frac{1 - 0.8q^{-1} + 0.1q^{-2}}{1 - 0.2q^{-1}}e(t),$$

where e(t) is a white Gaussian process with E(e(t)) = 0 and  $E(e(t)^2) = 1$  such that e(t) and w(s) are independent for all  $t, s \in \mathbb{Z}$ .

An identification experiment has been performed on this generalized Wiener system with a realization of this u(t) as input and 100 000 measurements of u(t) and y(t) have been collected. A linear output error model  $\hat{G}_{OE}(q)$  with  $n_b = n_f = 2$  and  $n_k = 0$  has been estimated from the measurements and the result was

$$\hat{G}_{OE}(q) = \frac{0.9292 - 2.066q^{-1}}{1 + 0.5971q^{-1} + 0.09784q^{-2}}.$$
(7.14)

From (7.14) we can see that the denominator of  $G_{OE}(q)$  is close to the denominator of G(q) also when the data has been generated by a generalized Wiener system. The MATLAB code that has been used in this example is available in Appendix B.3.

In this chapter we have studied OE-LTI-SOEs of NFIR systems with Gaussian inputs. We have shown that the OE-LTI-SOE of such a system is always an FIR model with certain coefficients and that this fact can be used for structure identification of NFIR systems. More specifically, it can be used to tell which input signal components that actually affect the output in a nonlinear way. Furthermore, we have shown that the OE-LTI-SOE of a generalized Wiener-Hammerstein system always will be the product of the LTI parts of the system and an FIR model. Hence, it is possible to estimate the denominator polynomials of the linear parts consistently without compensating for any nonlinearities.

In the next chapter, OE-LTI-SOEs of nonlinear systems that are almost linear will be studied. In particular, the sensitivity of the OE-LTI-SOE to small nonlinearities will be investigated and it will be shown that this sensitivity in some cases depends on how non-Gaussian the input signal is.

8

## **Almost Linear Systems**

When system identification is used to model real-life systems, it is very common to neglect the presence of small nonlinearities in the true system. This works well in many cases but, as we will see in the beginning in this chapter, LTI approximations of almost linear systems can sometimes exhibit a rather strange behavior. Later in this chapter, a convergence result that holds when the nonlinearities tend to zero will also be shown. Finally, a bound on the distance between the OE-LTI-SOE of an almost linear NFIR system and the linear part of that system will be given.

#### 8.1 Almost Linear Systems

The use of a linear model is very natural when the true system is close to being linear. In many cases, the behavior of an almost linear system can be understood, at least intuitively, from the theory of linear systems. Hence, it is a legitimate question to ask whether this linear intuition can be extended also to OE-LTI-SOEs for almost linear systems. An almost linear system will here be defined as a system that for a certain input signal can be written

$$y(t) = G_l(q)u(t) + \alpha y_n(t) + w(t),$$

where the linear term  $G_l(q)u(t)$  has much larger variance than the nonlinear term  $\alpha y_n(t)$ . Here, the parameter  $\alpha$  defines the size of the nonlinear part of the system and w(t) is a noise term.

If the nonlinear contribution to the output is small for a certain input, one might assume that the corresponding OE-LTI-SOE would be close to the linear part of the system in some sense. However, as we will see in the following example, this is not always the case.

#### — Example 8.1 —

Consider the nonlinear system

$$y(t) = y_l(t) + \alpha y_n(t)$$
  

$$y_l(t) = u(t),$$
  

$$y_n(t) = u(t)^3.$$

The output from this system consists of a linear part,  $y_l(t)$ , and a nonlinear part,  $\alpha y_n(t)$ , whose size is controlled by the parameter  $\alpha$ . Here, the transfer function  $G_l(q)$  of the linear part is equal to one. For bounded input signals, small values of  $\alpha$  will give a system output that is close to the output from  $G_l$ . In this sense,  $\alpha$  defines how close the nonlinear system is to the linear system  $G_l(q)$ . Let the input signal be

$$u(t) = L_m(q, c)e(t),$$
 (8.1)

where

$$L_m(q,c) = \left(1 - cq^{-1}\right)^2 = 1 - 2cq^{-1} + c^2q^{-2}, \quad 0 < c < 1$$
(8.2)

and where e(t) is a white noise process with uniform distribution over the interval [-1, 1]. For all c with 0 < c < 1, the input is bounded, -4 < u(t) < 4. For this input, a small value of  $\alpha$  like, for example,  $\alpha = 0.01$  will give an output that is very similar to the output from  $G_l$ , i.e., the output when  $\alpha = 0$ . This can be seen in Figure 8.1a for a particular realization of the input signal. However, the small differences between these output signals will sometimes give rise to totally different OE-LTI-SOEs.

Since the input is generated by filtering white noise through a minimum phase filter, Theorem 5.2 gives that the OE-LTI-SOE can be written

$$G_{0,OE}(z,\alpha,c) = \frac{\Phi_{yu}(z,\alpha,c)}{\Phi_u(z,c)} = \underbrace{G_l(z)}_{-1} + \alpha \frac{\Phi_{y_n e}(z,c)}{L_m(z,c)R_e(0)}.$$
(8.3)

If  $y_n(t)$  is expanded we get

$$y_n(t) = e(t)^3 - 6ce(t)^2 e(t-1) + 3c^2 e(t)^2 e(t-2) + 12c^2 e(t)e(t-1)^2 - 12c^3 e(t)e(t-1)e(t-2) + 3c^4 e(t)e(t-2)^2 - 8c^3 e(t-1)^3 + 12c^4 e(t-1)^2 e(t-2) - 6c^5 e(t-1)e(t-2)^2 + c^6 e(t-2)^3.$$

Using the fact that  $E(e(t)^2) = \frac{1}{3}$  and  $E(e(t)^4) = \frac{1}{5}$  and that e(t) and e(t - k) are independent when  $k \neq 0$ , the cross-covariance function  $R_{y_n e}(\tau, c)$  can be calculated as

$$\begin{split} R_{y_n e}(0,c) &= \mathcal{E}(y_n(t)e(t)) = \mathcal{E}(e(t)^4) + 12c^2\mathcal{E}(e(t)^2e(t-1)^2) \\ &\quad + 3c^4\mathcal{E}(e(t)^2e(t-2)^2) = \frac{1}{5} + 12c^2\frac{1}{9} + 3c^4\frac{1}{9} = \frac{1}{15}\left(3 + 20c^2 + 5c^4\right), \\ R_{y_n e}(1,c) &= \mathcal{E}(y_n(t)e(t-1)) = -6c\mathcal{E}(e(t)^2e(t-1)^2) - 8c^3\mathcal{E}(e(t-1)^4) \\ &\quad - 6c^5\mathcal{E}(e(t-1)^2e(t-2)^2) = -6c\frac{1}{9} - 8c^3\frac{1}{5} - 6c^5\frac{1}{9} \\ &= -\frac{1}{15}\left(10c + 24c^3 + 10c^5\right), \end{split}$$

$$\begin{split} R_{y_n e}(2,c) &= \mathcal{E}(y_n(t)e(t-2)) = 3c^2 \mathcal{E}(e(t)^2 e(t-2)^2) + 12c^4 \mathcal{E}(e(t-1)^2 e(t-2)^2) \\ &+ c^6 \mathcal{E}(e(t-2)^4) = 3c^2 \frac{1}{9} + 12c^4 \frac{1}{9} + c^6 \frac{1}{5} = \frac{1}{15} \left( 5c^2 + 20c^4 + 3c^6 \right), \\ R_{y_n e}(\tau,c) &= 0 \quad \forall \tau \in \mathbb{Z} \setminus \{0, 1, 2\}. \end{split}$$

Inserted in (8.3) this gives

$$G_{0,OE}(z,\alpha,c) = = 1 + \frac{\alpha}{5} \cdot \frac{\left(3 + 20c^2 + 5c^4\right) - \left(10c + 24c^3 + 10c^5\right)z^{-1} + \left(5c^2 + 20c^4 + 3c^6\right)z^{-2}}{1 - 2cz^{-1} + c^2z^{-2}}.$$
(8.4)

Let  $\Delta_G(z, \alpha, c) \triangleq \sum_{k=0}^{\infty} \delta_G(k, \alpha, c) z^{-k} = G_{0,OE}(z, \alpha, c) - G_l(z)$ . Then the static gain of  $\Delta_G$  is

$$\Delta_G(1,\alpha,c) = \frac{\alpha}{5} \cdot \frac{3 - 10c + 25c^2 - 24c^3 + 25c^4 - 10c^5 + 3c^6}{(1-c)^2}.$$
 (8.5)

From (8.5) we see that the numerator of  $\Delta_G(1, \alpha, c)$  approaches  $12\alpha$  when  $c \to 1$ , i.e., for c close to 1 we have  $\Delta_G(1, \alpha, c) \approx \frac{12\alpha}{5(1-c)^2}$ . This implies that no matter how small  $\alpha > 0$  we select, we can always make  $\Delta_G(1, \alpha, c)$  arbitrarily large by choosing a c sufficiently close to 1. That is, no matter how linear the system is, there is always a bounded input signal such that its OE-LTI-SOE is far from  $G_l(e^{i\omega})$  for  $\omega = 0$ . The difference between  $|G_{0,OE}(e^{i\omega}, 0.01, 0.99)|$  and  $|G_l(e^{i\omega})|$  is shown in Figure 8.1b. Furthermore, since

$$|\Delta_G(1,\alpha,c)| = |\sum_{k=0}^{\infty} \delta_G(k,\alpha,c)| \le \sum_{k=0}^{\infty} |\delta_G(k,\alpha,c)|,$$

it follows that for any  $\alpha > 0$ , the  $l^1$ -norm of the impulse response of  $\Delta_G$  can be made arbitrarily large by an appropriate choice of c. It can also be shown that, for any fixed  $\alpha > 0$ ,

$$\sum_{k=0}^{\infty} \delta_G(k,\alpha,c)^2 = \frac{1}{2\pi} \int_{-\pi}^{\pi} |\Delta_G(e^{i\omega},\alpha,c)|^2 \, d\omega$$

can be made arbitrarily large by taking a c close to 1.

The previous example is a clear indication that the OE-LTI-SOE not always can be understood from linear theory. There is no corresponding behavior in the linear case, since a small linear, time-invariant deviation from  $G_l$  in Example 8.1 only would have given rise to exactly the same small deviation in the OE-LTI-SOE. It should be noted that for small  $\alpha$  values,  $G_l$  is a good model of the true system and it can predict the output very well from the input signal. Despite this, the prediction error estimate will not converge to  $G_l$ but to  $G_{0,OE}$  when the number of measurements tends to infinity.

In the case of linear undermodeling, i.e., when a linear system is approximated by a model of lower order, the optimal approximation depends only of the spectral density of the input Ljung (1999) and not on the distribution of the input signal components.



(a) It is hard to distinguish the output y(t) (dashed) of the nonlinear system (with  $\alpha = 0.01$  and c = 0.99) in Example 8.1 from the output  $y_l(t) = u(t)$  (solid) of the linear part of that system for a particular realization of the input signal.



(b) The frequency response of the OE-LTI-SOE  $G_{0,OE}(e^{i\omega}, 0.01, 0.99)$  (dashed) differs from the one of the linear part  $G_l(e^{i\omega})$  (solid) of the system in Example 8.1.

**Figure 8.1:** The frequency response of the OE-LTI-SOE can be far from the response of the linear part of the system also when the nonlinear contributions to the output are small.

However, when a nonlinear system is approximated with a linear model, the distribution of the input affects the result significantly. For example, in Example 8.1 the input is generated from the white uniformly distributed signal e(t). If this signal is replaced with a white Gaussian signal with the same variance, the corresponding OE-LTI-SOE will be completely different. This is shown in the next example.

#### — Example 8.2

Consider once again the nonlinear system in Example 8.1 but now with an input u(t) generated by filtering a white Gaussian process e(t) with zero mean and variance 1/3 through the filter  $L_m(q,c)$  in (8.2). Since the system is static and the input is Gaussian, Theorem 3.1 gives that

$$R_{yu}(\tau, \alpha, c) = R_u(\tau) + \alpha \mathbb{E}(3u(t)^2)R_u(\tau)$$
  
=  $(1 + \alpha(1 + 4c^2 + c^4))R_u(\tau),$  (8.6)

where we have used that  $E(u(t)^2) = (1 + 4c^2 + c^4)/3$ . Since (8.6) implies that

$$\Phi_{yu}(z, \alpha, c) = (1 + \alpha(1 + 4c^2 + c^4))\Phi_u(z),$$

the OE-LTI-SOE of the system is

$$G_{0,OE}(z,\alpha,c) = (1 + \alpha(1 + 4c^2 + c^4))$$

for this particular input. Hence, the fact that the input is Gaussian implies the OE-LTI-SOE is static. Furthermore, it is obvious that the deviation of the OE-LTI-SOE from the linear part (i.e., from 1) is bounded in this case. With  $\Delta_G(z, \alpha, c) = G_{0,OE}(z, \alpha, c) - G_l(z) = \alpha(1 + 4c^2 + c^4)$  and 0 < c < 1, we have

$$|\Delta_G(z,\alpha,c)| < 6|\alpha|.$$

Here, unlike in Example 8.1, we cannot make  $\Delta_G$  large for an arbitrarily small  $\alpha > 0$  by selecting c < 1 sufficiently close to 1. For example,  $\alpha = 0.01$  and c = 0.99 give

$$G_{0,OE}(z, 0.01, 0.99) \approx 1.0588$$
 (8.7)

and  $|\Delta_G(z, \alpha, c)| \approx 0.0588$ , which can be compared with the large  $\Delta_G$  that is obtained when e(t) is uniformly distributed (see Figure 8.1b).

From Example 8.2 we see that the fact that the input is Gaussian can, at least in some cases, prevent the OE-LTI-SOE from being far from the linear part of an almost linear system. Probably, this is often a desirable property and hence one could argue that Gaussian inputs should always be used in identification experiments where slightly nonlinear systems are approximated with LTI models. However, an input with a distribution that is similar to a Gaussian distribution might also work fine. This is illustrated in the next example.

#### Example 8.3 –

Again, consider the nonlinear system in Example 8.1 with an input u(t) generated by filtering a white process e(t) with zero mean and variance 1/3 through the filter  $L_m(q, c)$ in (8.2). Let c = 0.99 and  $\alpha = 0.01$ . In the two previous examples, it has been shown that this choice of parameters and a uniformly distributed e(t) give an OE-LTI-SOE with the frequency response shown in Figure 8.1b while a Gaussian e(t) gives the OE-LTI-SOE in (8.7). Here, it will be shown that for non-Gaussian choices of e(t) with distributions that are more Gaussian than the uniform distribution, the OE-LTI-SOEs will have frequency responses that are closer to the linear part of the system. Let

$$e_M(t) = \frac{1}{\sqrt{M}} \sum_{k=1}^M \tilde{e}_k(t), \quad M \in \mathbb{Z}_+,$$

where  $\tilde{e}_k(t)$  are independent white signals with uniform distribution over the interval [-1,1] and zero mean. In this way,  $E(e_M(t)^2) = 1/3$  for all M and the central limit theorem implies that  $e_M(t)$  will become more Gaussian for larger M. Let

$$u_M(t) = L_m(q, 0.99)e_M(t)$$

be inputs to the nonlinear system and let the OE-LTI-SOE of the system for each of these inputs be denoted by  $G_{0,OE,M}(z)$ . In this way,  $u_1(t)$  and  $G_{0,OE,1}(z)$  correspond to the input and the OE-LTI-SOE in Example 8.1, respectively. Since all inputs have been generated in the same way as this input,

$$G_{0,OE,M}(z) = \frac{b_0(M) + b_1(M)z^{-1} + b_2(M)z^{-2}}{1 - 1.98z^{-1} + 0.9801z^{-2}}$$



**Figure 8.2:** The frequency responses of the estimated output error models  $\hat{G}_{0,OE,G}(z)$  (dash-dotted),  $\hat{G}_{0,OE,1}(z)$  (thin dashed),  $\hat{G}_{0,OE,2}(z)$  (solid) and  $\hat{G}_{0,OE,8}(z)$  (thick dashed) from Example 8.3.

for all M. Furthermore, let  $u_G(t)$  denote the Gaussian input in Example 8.2 that is obtained when e(t) is a white Gaussian input with zero mean and variance 1/3 and let  $G_{0,OE,G}(z)$  be the corresponding OE-LTI-SOE (i.e., the OE-LTI-SOE in (8.7)).

Data sets with 50 000 measurements of  $u_M(t)$  for M = 1, 2 and 8,  $u_G(t)$  and of the corresponding system outputs have been generated in MATLAB and an output error model  $\hat{G}_{0,OE,M}(z)$  with  $n_b = 3$ ,  $n_f = 2$  and  $n_k = 0$  has been estimated for each data set. The frequency responses of these estimated models are shown in Figure 8.2.

By comparing this figure with Figure 8.1b, it can be seen that  $\hat{G}_{0,OE,1}(z)$  indeed is close to the exact OE-LTI-SOE. Furthermore, Figure 8.2 shows that the OE-LTI-SOEs for the non-Gaussian inputs get closer to the OE-LTI-SOE for the Gaussian input if the input has a more Gaussian-like distribution.

The results in Example 8.3 indicate that the distance between the OE-LTI-SOE and the linear part of an almost linear system depends on how non-Gaussian the input is. With this in mind, it seems preferable to use inputs that are as Gaussian as possible in this case.

Examples 8.1 to 8.3 show that the OE-LTI-SOE in some cases can be far from the linear part of the system. This can in some circumstances be an undesirable property, e.g., if the OE-LTI-SOE is supposed to be used as a basis for robust control design. Such a design puts restrictions on the control laws in order to guarantee the stability of the resulting true closed-loop system, despite the presence of model errors.

Assume that the true system is almost linear in the sense that it deviates from an LTI system  $G_l$  by a nonlinearity with a small gain. In this case,  $G_l$  is a very good basis for robust control design, since the small nonlinearity often only gives rise to rather mild restrictions on the controller. However, if an OE-LTI-SOE that is far from  $G_l$  is used for the controller design, the restrictions on the controller might become much harder, since the gain of the model error now is large. Hence, it is interesting to investigate under what circumstances we can guarantee that the OE-LTI-SOE will be close to the linear part of the system when the nonlinearities are small. Some answers to this question will be given in this chapter.

Consider a system  $y(t) = f((u(t-k))_{k=0}^{\infty}, \alpha)$  where  $\alpha$ , just like in the previous examples, is a parameter that defines the size of the nonlinear part of f. Let  $S_A$  denote the set of all inputs such that they, and the system outputs they generate, fulfill Assumptions A1 and A2. Assume that f is continuous at  $\alpha = 0$  and that

$$f((u(t-k))_{k=0}^{\infty}, 0)$$

is a stable and causal LTI system  $G_l$ , i.e.,

$$f((u(t-k))_{k=0}^{\infty}, 0) = \sum_{k=0}^{\infty} g_l(k)u(t-k) = G_l(q)u(t).$$
(8.8)

Let  $G_{0,OE}(z, \alpha)$  denote the OE-LTI-SOE that is obtained for a particular input signal u and a particular  $\alpha$ . The conclusion that we can draw from Example 8.1 is that we cannot in general assume that, for example,

$$\sup_{u \in S_A} |G_{0,OE}(e^{i\omega}, \alpha) - G_l(e^{i\omega})|$$
(8.9)

will approach 0 when  $\alpha \to 0$  for a fixed  $\omega \in [-\pi, \pi]$ . For some systems we can, whenever there is a small nonlinear term in the system output, find a *u* for which the OE-LTI-SOE is far from  $G_l$ . This property of the OE-LTI-SOE makes it different from, for example, a linearization based on a Taylor series expansion. Since such a linearization is based only on local properties of the nonlinear system, e.g., for  $u(t) \equiv 0$ , it will converge uniformly when  $\alpha \to 0$ . However, as we will see in the next section, it is possible to show that also OE-LTI-SOEs exhibit some, less general, convergence properties.

#### 8.2 A Convergence Result

Despite the fact that it even for an almost linear system is not true that the OE-LTI-SOEs will be close to the linear part of the system for all inputs that fulfill Assumptions A1 and A2, it is possible to say something about the behavior of the OE-LTI-SOE for a more restricted class of input signals, whose main feature is that their spectral densities are strictly positive. This is done in the following theorem.

#### Theorem 8.1

Consider a nonlinear system  $y(t) = f((u(t-k))_{k=0}^{\infty}, \alpha) + w(t)$  such that the function  $f((x(t-k))_{k=0}^{\infty}, \alpha) \to f((x(t-k))_{k=0}^{\infty}, 0) = \sum_{k=0}^{\infty} g_l(k)x(t-k)$  uniformly on the set of sequences  $M_f = \{(x(t-k))_{k=0}^{\infty} \mid |x(t-k)| < u_{max} \forall k \in \mathbb{N}\}$  when  $\alpha \to 0$ . Assume that the limit  $G_l(q)$  is a stable and causal LTI system. Let  $S_f$  denote the set of stochastic input signals that fulfill the following conditions

- (i)  $P(|u(t)| \ge u_{max}) = 0$ ,
- (ii)  $\operatorname{E}(|u(t)|) \leq m_c < \infty$ ,
- (iii)  $\Phi_u(e^{i\omega}) \ge \mu_c > 0$  for all  $\omega \in [-\pi, \pi]$ ,
- (iv) u(t) is such that  $G_{0,OE}(z,\alpha)$  is regular for all  $\alpha$  with  $|\alpha| < \alpha_{max}$ ,

- (v) u(t) and  $y(t) = f((u(t-k))_{k=0}^{\infty}, \alpha) + w(t)$  fulfill Assumptions A1, A2 and A5 for all  $\alpha$  with  $|\alpha| < \alpha_{max}$ ,
- (vi)  $\exists M_c \in \mathbb{Z}_+, \lambda_c, 0 \leq \lambda_c < 1, K_c > 0$  such that  $|R_{yu}(\tau, \alpha)| < K_c \lambda_c^{|\tau|}$  when  $|\tau| > M_c \,\forall \alpha \text{ with } |\alpha| < \alpha_{max}$ ,

where  $u_{max}$ ,  $m_c$ ,  $\mu_c$ ,  $\alpha_{max}$ ,  $M_c$ ,  $\lambda_c$  and  $K_c$  are given constants. Then

$$\sup_{u \in S_f} \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}, \alpha) - G_l(e^{i\omega})|^n \, d\omega \to 0, \quad \alpha \to 0, \quad n = 1, 2.$$
(8.10)

**Proof:** Take an arbitrary  $\tau \in \mathbb{Z}$ . Then

$$\begin{split} \sup_{u \in S_{f}} |R_{yu}(\tau, \alpha) - R_{yu}(\tau, 0)| \\ &= \sup_{u \in S_{f}} \left| \mathrm{E}(\left( f((u(t-k))_{k=0}^{\infty}, \alpha) - \sum_{k=0}^{\infty} g_{l}(k)u(t-k)\right) u(t-\tau)) \right| \\ &\leq \sup_{u \in S_{f}} \mathrm{E}(\left| f((u(t-k))_{k=0}^{\infty}, \alpha) - \sum_{k=0}^{\infty} g_{l}(k)u(t-k)\right| |u(t-\tau)|) \\ &\leq \sup_{u \in S_{f}} \mathrm{E}(\sup_{x \in M_{f}} \left| f((x(t-k))_{k=0}^{\infty}, \alpha) - \sum_{k=0}^{\infty} g_{l}(k)x(t-k)\right| |u(t-\tau)|) \\ &= \sup_{x \in M_{f}} \left| f((x(t-k))_{k=0}^{\infty}, \alpha) - \sum_{k=0}^{\infty} g_{l}(k)x(t-k)\right| \sup_{u \in S_{f}} \mathrm{E}(|u(t-\tau)|) \\ &\leq \sup_{x \in M_{f}} \left| f((x(t-k))_{k=0}^{\infty}, \alpha) - \sum_{k=0}^{\infty} g_{l}(k)x(t-k)\right| \sup_{u \in S_{f}} \mathrm{E}(|u(t-\tau)|) \\ &\leq \sup_{x \in M_{f}} \left| f((x(t-k))_{k=0}^{\infty}, \alpha) - \sum_{k=0}^{\infty} g_{l}(k)x(t-k)\right| m_{c} \to 0, \quad \alpha \to 0. \end{split}$$

Here, we have used (i) and (ii) in the second and last inequality, respectively. Since  $\tau$  was arbitrary it follows that

$$\sup_{u \in S_f} |R_{yu}(\tau, \alpha) - R_{yu}(\tau, 0)| \to 0, \quad \alpha \to 0, \quad \forall \tau \in \mathbb{Z}.$$
(8.11)

Now we need to show that  $\sup_{u \in S_f} \int_{-\pi}^{\pi} |\Phi_{yu}(e^{i\omega}, \alpha) - \Phi_{yu}(e^{i\omega}, 0)|^2 d\omega \to 0$  when  $\alpha \to 0$ . Take an arbitrary  $\varepsilon > 0$ . By Parseval's identity, which holds for all  $\alpha$  with  $|\alpha| < \alpha_{max}$  according to (vi), we get

$$\sup_{u \in S_f} \int_{-\pi}^{\pi} |\Phi_{yu}(e^{i\omega}, \alpha) - \Phi_{yu}(e^{i\omega}, 0)|^2 d\omega$$
  
= 
$$\sup_{u \in S_f} 2\pi \sum_{\tau = -\infty}^{\infty} |R_{yu}(\tau, \alpha) - R_{yu}(\tau, 0)|^2$$
  
= 
$$\sup_{u \in S_f} 2\pi \sum_{\tau = -C_0}^{C_0} |R_{yu}(\tau, \alpha) - R_{yu}(\tau, 0)|^2 + Q(C_0, \alpha).$$

Choose  $C_0$  such that the term  $Q(C_0, \alpha)$  is less than  $\varepsilon/2$  for all  $\alpha$  with  $|\alpha| < \alpha_{max}$ . (This is possible since the tails of the series above will be small for all  $\alpha$  according to (vi)). Then, from (8.11) it follows that  $\exists \delta_{\varepsilon} > 0$  such that

$$|\alpha| < \delta_{\varepsilon} \quad \Rightarrow \quad \sup_{u \in S_f} 2\pi \sum_{\tau = -C_0}^{C_0} |R_{yu}(\tau, \alpha) - R_{yu}(\tau, 0)|^2 < \varepsilon/2.$$

Thus  $\sup_{u \in S_f} \int_{-\pi}^{\pi} |\Phi_{yu}(e^{i\omega}, \alpha) - \Phi_{yu}(e^{i\omega}, 0)|^2 d\omega < \varepsilon$  if  $|\alpha| < \delta_{\varepsilon}$  and since  $\varepsilon$  was arbitrary we get

$$\sup_{u\in S_f} \int_{-\pi}^{\pi} |\Phi_{yu}(e^{i\omega},\alpha) - \Phi_{yu}(e^{i\omega},0)|^2 \, d\omega \to 0, \quad \alpha \to 0.$$
(8.12)

Using (iii) and (iv) together with (8.12), we obtain

$$\begin{split} \sup_{u \in S_f} & \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}, \alpha) - G_l(e^{i\omega})|^2 \, d\omega \\ &= \sup_{u \in S_f} \int_{-\pi}^{\pi} \frac{|\Phi_{yu}(e^{i\omega}, \alpha) - \Phi_{yu}(e^{i\omega}, 0)|^2}{\Phi_u(e^{i\omega})^2} \, d\omega \\ &\leq \sup_{u \in S_f} \frac{1}{\mu_c^2} \int_{-\pi}^{\pi} |\Phi_{yu}(e^{i\omega}, \alpha) - \Phi_{yu}(e^{i\omega}, 0)|^2 \, d\omega \to 0, \quad \alpha \to 0 \end{split}$$

and thus we have shown (8.10) for n = 2. Finally, Schwarz inequality gives

$$\begin{split} \sup_{u \in S_f} &\int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}, \alpha) - G_l(e^{i\omega})| \, d\omega \\ &\leq \sup_{u \in S_f} \sqrt{2\pi} \left( \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}, \alpha) - G_l(e^{i\omega})|^2 \, d\omega \right)^{1/2} \to 0, \quad \alpha \to 0. \end{split}$$

Hence, we have shown (8.10) also for n = 1.

Theorem 8.1 gives conditions on the system and set of input signals that guarantee a uniform convergence of the OE-LTI-SOEs when  $\alpha$  tends to zero, i.e., when the system becomes linear. The reason why we cannot apply this theorem in Example 8.1 is that the set of input signals generated by (8.1) for all c with 0 < c < 1 does not fulfill condition (iii) since  $\Phi_u(1) = \frac{(1-c)^2}{3}$ . Hence, there is no constant  $\mu_c > 0$  such that all considered input signals fulfill  $\Phi_u(e^{i\omega}) \ge \mu_c$  for all  $\omega$  in the interval  $[-\pi, \pi]$ .

Furthermore, Theorem 8.1 shows that it does not matter if  $G_{0,OE}(z)$  and  $G_l(z)$  have different orders. For input signals in  $S_f$ ,  $G_{0,OE}(z)$  will still approach  $G_l(z)$  in a well-behaved way according to (8.10) when the nonlinearities tend to zero.

#### 8.3 Almost Linear NFIR Systems

The theorem in the previous section tells us that the OE-LTI-SOE will converge to the linear part of the system when the nonlinearities tend to zero if  $u \in S_f$ , but not how fast this convergence is. In order to be able to derive an upper bound on the distance between the OE-LTI-SOE and the linear part of a system with a nonzero nonlinearity of a certain size, we will have to make some new restrictions on the types of systems and excitation signals.

Hence, we will in this section only consider NFIR systems with white input signals that can be written like  $y(t) = f((u(t-k))_{k=0}^M) + w(t)$  and that are close to a linear system  $z(t) = \sum_{k=0}^{M} g_l(k)u(t-k) + w(t)$ . The following theorem gives an upper bound on the distance between the OE-LTI-SOE and the linear part of such a nonlinear system.

#### Theorem 8.2

Consider an NFIR system

$$y(t) = f((u(t-k))_{k=0}^{M}) + w(t),$$

where u(t) is a white input signal and where f satisfies

$$\left| f((u(t-k))_{k=0}^{M}) - \sum_{k=0}^{M} g_l(k)u(t-k) \right| < a, \quad \forall t \in \mathbb{Z}$$
(8.13)

for every realization of u. Assume that the output y(t) together with u(t) fulfill the conditions in Assumptions A1, A2 and A5. Then

$$\int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G_l(e^{i\omega})| \, d\omega < a2\pi\sqrt{(M+1)} \left(\frac{\mathrm{E}(|u(t)|)}{\mathrm{E}(u(t)^2)}\right),\tag{8.14a}$$

$$\int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G_l(e^{i\omega})|^2 \, d\omega < a^2 2\pi (M+1) \left(\frac{\mathrm{E}(|u(t)|)}{\mathrm{E}(u(t)^2)}\right)^2.$$
(8.14b)

**Proof:** We start by proving the following inequality

$$\left| R_{yu}(\tau) - \sum_{k=0}^{M} g_l(k) R_u(\tau - k) \right| < a \mathbb{E}(|u(t-\tau)|), \quad \forall \tau \in \mathbb{Z}.$$
(8.15)

Take an arbitrary  $\tau \in \mathbb{Z}$ . Then Assumption A5 gives

$$\begin{aligned} \left| R_{yu}(\tau) - \sum_{k=0}^{M} g_{l}(k) R_{u}(\tau - k) \right| \\ &= \left| \mathrm{E}(y_{nf}(t) u(t - \tau)) - \sum_{k=0}^{M} g_{l}(k) \mathrm{E}(u(t - k) u(t - \tau)) \right| \\ &= \left| \mathrm{E}\Big( \Big( y_{nf}(t) - \sum_{k=0}^{M} g_{l}(k) u(t - k) \Big) u(t - \tau) \Big) \right| \end{aligned}$$

$$\leq \mathbf{E}(|y_{nf}(t) - \sum_{k=0}^{M} g_l(k)u(t-k)||u(t-\tau)|) < a\mathbf{E}(|u(t-\tau)|).$$

Since  $\tau$  was arbitrary, (8.15) follows. The assumption that u(t) consists of independent random variables implies that  $\Phi_u(e^{i\omega}) = R_u(0)$ ,

$$\sum_{k=0}^{M} g_l(k) R_u(\tau - k) = g_l(\tau) R_u(0), \quad 0 \le \tau \le M,$$

and that  $R_{yu}(\tau) = 0$  when  $\tau > M$  or  $\tau < 0$ . This, together with Parseval's identity and Equation (8.15) give

$$\begin{split} & \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G_l(e^{i\omega})|^2 \, d\omega \\ &= \frac{1}{R_u(0)^2} \int_{-\pi}^{\pi} |\Phi_{yu}(e^{i\omega}) - G_l(e^{i\omega})R_u(0)|^2 \, d\omega \\ &= \frac{2\pi}{R_u(0)^2} \sum_{\tau=0}^{M} |R_{yu}(\tau) - g_l(\tau)R_u(0)|^2 \\ &< a^2 2\pi (M+1) \left(\frac{\mathbf{E}(|u(t)|)}{\mathbf{E}(u(t)^2)}\right)^2. \end{split}$$

Finally, Schwarz inequality gives

$$\int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G_l(e^{i\omega})| \, d\omega \le \left( \int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G_l(e^{i\omega})|^2 \, d\omega \right)^{1/2} \sqrt{2\pi} \\ < a 2\pi \sqrt{(M+1)} \left( \frac{\mathcal{E}(|u(t)|)}{\mathcal{E}(u(t)^2)} \right).$$

r		

Theorem 8.2 tells us that the distance

$$\int_{-\pi}^{\pi} |G_{0,OE}(e^{i\omega}) - G_l(e^{i\omega})| \, d\omega$$

between the OE-LTI-SOE and  $G_l$  is less than a bound that is proportional to a, i.e., to the upper bound on the size of the nonlinearity in (8.13). Furthermore, this theorem shows the effect on the OE-LTI-SOE of a scaling of the input signal.

The use of  $\tilde{u} = \nu u$  as input instead of u will result in a new OE-LTI-SOE. The distance in (8.14a) between this new OE-LTI-SOE and  $G_l$  will have an upper bound that

is  $\frac{1}{|\nu|}$  times the original bound. When a white input signal is used, it is thus possible to reduce the distance between the OE-LTI-SOE and the linear part of a nonlinear FIR system that fulfill (8.13) simply by scaling the input signal. This is natural, since the relative error gets smaller for large inputs.

The main objective of this chapter has been to describe the behavior of OE-LTI-SOEs for almost linear systems. It has been shown that the OE-LTI-SOE sometimes can be far from the linear part of such a system and that the distribution of the input is very important for this behavior. Furthermore, a convergence result has been derived and, finally, a bound on the distance between the OE-LTI-SOE of an almost linear NFIR system with a white input signal has been presented.

# 9

## Discussion

A number of results about LTI-SOEs have been shown in the previous chapters of this thesis. Since some of these results have practical implications, it might be appropriate to compile some guidelines for the user. This will be done in this chapter.

It should once again be pointed out that the conclusions that have been drawn in Chapters 4 to 8 concern *asymptotic* properties of prediction-error model estimates, i.e., properties when the number of measurements and, in some cases, the number of experiments tends to infinity. Hence, the results are usually only applicable to identification problems where large data sets are used.

Previously, we have seen that there can be remaining correlation between the input and the model residuals both for the OE-LTI-SOE and the GE-LTI-SOE. If no additional knowledge about the system structure is available, this correlation might be taken as an indication that the system operates in closed loop.

However, for some classes of input signals, this cannot happen for an open-loop nonlinear system. If the input has been generated by filtering white noise through a minimum phase filter, there will be no spurious correlation between the input and the residuals of the OE-LTI-SOE and the GE-LTI-SOE according to Lemma 5.4 and Corollary 5.1, respectively. Furthermore, these input signals have the following properties.

- Each minimum phase generated input signal is optimal over a class of other input signals in the sense that the variance of the residuals of the OE-LTI-SOE is minimized (see Theorem 5.3).
- Residual and spectral analysis can be used to validate an estimated model and to see if it is sufficiently close to the OE-LTI-SOE of the system (see Section 5.5.2).
- A minimum phase generated input signal is a good choice of reference signal for closed-loop identification using the two-step method since it implies that only a causal S(z) has to be estimated (see Section 5.5.3).

If the nonlinear system is a generalized Hammerstein system, it might be desirable to use a separable input signal. These signals have the following properties.

- For a separable input signal there will be no extra dynamics in the OE-LTI-SOE. Hence, the OE-LTI-SOE of an NFIR system will be an FIR model of the same order and the OE-LTI-SOE of a generalized Hammerstein system will be the cascade product of an FIR model and the linear part of the system (see Theorems 6.3 and 6.4). Hence, it is possible to estimate the denominator polynomial of the linear part consistently without compensating for the nonlinearities.
- In particular, for a separable input, the OE-LTI-SOE of a Hammerstein system will be a scaled version of the linear part of the system (see Theorem 6.1).

One particular choice of a separable input signal is Gaussian noise with arbitrary color. This class of input signals is also a subset of the class of minimum phase generated signals and it has the following properties in addition to the ones above.

- For a Gaussian input signal, the OE-LTI-SOE of an NFIR system will be an FIR model with coefficients that are expectations of the partial derivatives of the non-linear function (see Theorem 7.2). This simplifies structure identification of such a system significantly.
- Furthermore, the OE-LTI-SOE of a generalized Hammerstein, Wiener or Wiener-Hammerstein system will be the cascade product of the linear parts of that system and an FIR model (see Corollary 7.1). Again, this implies that it is possible to estimate the denominator polynomials of the linear parts consistently without compensating for the nonlinearities.

With the properties mentioned above in mind, it is possible to give a general advice concerning LTI approximations of nonlinear systems using the prediction-error method. Although there might be circumstances where this advice is not suitable, it is at least applicable to the identification problems studied in this thesis.

Use a Gaussian signal, a separable random multisine or any other separable signal as the input in the identification experiment. If this is not possible, use any other signal generated by filtering white noise through a minimum phase filter. Collect a large data set by using either one long realization of the input or a large number of different realizations, depending on the type of input signal used.

## Part II Identification and Control
# 10 Robust Control

A number of properties of LTI approximations of nonlinear systems have been discussed in the previous chapters. In most cases, these properties have been considered from a system identification point of view, i.e., the focus has been more on theoretical approximation issues and on questions like under which circumstances the LTI model will inherit structural properties from the true system. However, it is important to remember that in most applications, the modeling of a system is done for a specific purpose. Different modeling issues are relevant depending on this purpose and on the type of application.

In this chapter, some aspects of robust control design based on LTI models of nonlinear systems will be discussed. In most modern control design methods, the controller is calculated based on a model of the true systems. Such a controller is *robust* to model errors if it can be guaranteed that the differences between the model and the system will not cause instability when the controller is used with the true system. Control design based on an approximate model has been studied for a long time and there are thus a large number of different robustness results.

For example, when both the true system and the model are linear, and thus also the model error, several methods exist where the robustness issues can be considered explicitly in the design process (Zhou et al., 1996). Furthermore, many engineering guidelines concerning controller tuning and classic methods, such as loop shaping and state feedback, have intrinsic robustness properties. Typically, controllers designed with any of these methods will be robust to at least some linear model errors.

Also the common case when the true system is nonlinear and the design is based on a linear model has been studied previously. In principle, it is possible to analyze this setup using the small gain theorem (Vidyasagar, 1993; Sastry, 1999; Khalil, 2002). However, such an analysis will often be rather pessimistic. Intuitively, the result in the small gain theorem is that a closed-loop system will be stable if the product of the gains of the subsystems in the loop is less than one (see, for example, Sastry, 1999, p. 147). The gain  $\|\mathcal{S}\|$  of a system  $\mathcal{S}$  with input u(t) and output y(t) can be defined as the smallest constant

 $\beta$  for which

$$\|y\| \le \alpha + \beta \|u\|, \quad \forall u(t), \tag{10.1}$$

where  $\alpha$  is a constant and where  $\|\cdot\|$  denotes the signal norm

$$||z||^2 = \sum_{t=0}^{\infty} z(t)^T z(t).$$

The small gain theorem can be used to prove a robustness result that says that the true system is stable if the product of the gain of the model error and the gain from an output disturbance to u in the closed-loop system is less than one (Sastry, 1999, pp. 150-151). However, this result is most useful when the gain of the model error is small compared to the gain of the model.

As a simple example, consider a saturation system, i.e., the nonlinear system

$$y(t) = \begin{cases} 1, & u(t) > 1, \\ u(t), & |u(t)| \le 1, \\ -1, & u(t) < -1 \end{cases}$$

and assume that a linear approximation

$$\hat{y}(t) = b_0 u(t)$$

of this system should be defined such that the gain of the model error is as small as possible. However, it is easy to see that for every choice of  $b_0$ , the gain of the model error is equal to  $b_0$ . Of course, a model of a system is not very useful if the model error is of the same size as the model.

The same problem as in this example occurs when most nonlinear systems are approximated with a linear model. However, there are many successful real-life applications of control design based on an LTI model of a nonlinear system and thus, the robustness result based on the small gain theorem seems to be too pessimistic. In this chapter, it will be shown that a more useful theoretical robustness result can be obtained if the definition of the gain of a system is modified. The method used here is closely related to optimal control (see, for example, Bryson and Ho, 1975; van der Schaft, 1992). Furthermore, different approaches to robust stability of linear systems with input saturation have been investigated previously (Kim and Bien, 1994; Henrion and Tarbouriech, 1999).

It should also be mentioned that there are many results about system identification for control purposes available in the literature (Hjalmarsson, 2005). However, we will not discuss identification of LTI models for control design for nonlinear systems in this chapter, but merely point out that LTI models really can be useful for this purpose and that robust control design can be done also when a bound on the model error is known only for a subset of all possible input signals. Most of the results in this chapter have previously been published in Glad et al. (2005) and some related results can be found also in Glad et al. (2004). In the next section, the robustness result will first be described for general models while the special case of LTI models and saturated inputs will be studied in Section 10.2.

#### **10.1 Robust Control for Constrained Inputs**

Consider a general state-space model  $\mathcal{M}$  that can be written as

$$x_{\mathcal{M}}(t+1) = f_{\mathcal{M}}(x_{\mathcal{M}}(t), u(t)),$$
  

$$y_{\mathcal{M}}(t) = h_{\mathcal{M}}(x_{\mathcal{M}}(t), u(t)),$$
  

$$x_{\mathcal{M}}(0) = x_{\mathcal{M},0}$$
(10.2)

of a system with input  $u(t) \in \mathbb{R}^m$  and output  $y(t) \in \mathbb{R}^p$ . Note that the model (10.2) contains an assumption about the initial state. The output y(t) from the true system can be written

$$y(t) = y_{\mathcal{M}}(t) + \varepsilon(t), \qquad (10.3)$$

where the signal  $\varepsilon(t)$  contains both the effects of model errors and external disturbances. Assume that the simple model error model

$$u(t) \in U, \ 0 \le t \le N \Rightarrow \|\varepsilon\|_N \le \alpha(N) + \beta \|u\|_N \tag{10.4}$$

holds. Here, U is a subset of  $\mathbb{R}^m$ ,  $\alpha$  is a function with  $\alpha(N) \ge 0$  for all  $N, \beta \ge 0$  is a constant and  $\|\cdot\|_N$  denotes the truncated norm

$$||z||_N^2 = \sum_{t=0}^N z(t)^T z(t).$$

For example, the function  $\alpha(N)$  can be used to describe both the effects of erroneous initial conditions in the model (10.2) and bounded external disturbances. By comparing (10.4) with (10.1), it is obvious that (10.4) can be used as an alternative gain definition, valid only for a restricted class of inputs. Hence,  $\beta$  is an upper bound on the gain of the model error for these restricted inputs. It should be mentioned also that, although the model error model (10.4) looks simple, it can be a demanding task to establish it for a particular system.

Assume that a controller is to be designed using the model  $\mathcal{M}$  in (10.2) and (10.3) and with the objective to keep a signal z(t) small despite the influence from the error signal  $\varepsilon(t)$ . Here, the signal z(t) is assumed to be the output of a filter  $\mathcal{W}$  with y(t) as input. Furthermore, it is assumed that the filter  $\mathcal{W}$  can be written as

$$\begin{aligned} x_{\mathcal{W}}(t+1) &= f_{\mathcal{W}}(x_{\mathcal{W}}(t)) + g_{\mathcal{W}}(x_{\mathcal{W}}(t))y(t), \\ z(t) &= h_{\mathcal{W}}(x_{\mathcal{W}}(t)), \\ x_{\mathcal{W}}(0) &= x_{\mathcal{W},0}. \end{aligned}$$
(10.5)

The model  $\mathcal{M}$  in (10.2), which can be used when  $u(t) \in U$ , the filter  $\mathcal{W}$  in (10.5) and the output description (10.3) can be combined into a complete state-space model

$$\begin{aligned} x(t+1) &= f(x(t), u(t)) + n(x(t))\varepsilon(t), \\ y(t) &= h(x(t), u(t)) + \varepsilon(t), \\ z(t) &= m(x(t)), \\ x(0) &= x_0, \end{aligned}$$
(10.6)

where

 $x(t) = \begin{pmatrix} x_{\mathcal{M}}(t) \\ x_{\mathcal{W}}(t) \end{pmatrix}$ 

and

$$\begin{split} f(x(t), u(t)) &= \begin{pmatrix} f_{\mathcal{M}}(x_{\mathcal{M}}(t), u(t)) \\ f_{\mathcal{W}}(x_{\mathcal{W}}(t)) + g_{\mathcal{W}}(x_{\mathcal{W}}(t))h_{\mathcal{M}}(x_{\mathcal{M}}(t), u(t)) \end{pmatrix},\\ n(x(t)) &= \begin{pmatrix} 0 \\ g_{\mathcal{W}}(x_{\mathcal{W}}(t)) \end{pmatrix},\\ h(x(t), u(t)) &= h_{\mathcal{M}}(x_{\mathcal{M}}(t), u(t)),\\ m(x(t)) &= h_{\mathcal{W}}(x_{\mathcal{W}}(t)),\\ x_0 &= \begin{pmatrix} x_{\mathcal{M},0} \\ x_{\mathcal{W},0} \end{pmatrix}. \end{split}$$

Let

$$J_N = \sum_{t=0}^N (m(x(t))^T m(x(t)) + u(t)^T u(t) - \gamma^2 \varepsilon(t)^T \varepsilon(t))$$

and consider the problem of finding u(t) = k(x(t)) such that  $J_N \leq 0$  for all  $N \geq 0$ . One approach for solving this problem is described in the following theorem.

#### Theorem 10.1

Consider the model (10.6) and suppose that there is a positive semidefinite function V(x(t)) and a control law u = k(x) such that u(t) = k(x(t)) implies that  $u(t) \in U$  for all  $t \ge 0$  and that

$$m(x)^T m(x) + k(x)^T k(x) - \gamma^2 \varepsilon^T \varepsilon + V(f(x, k(x)) + n(x)\varepsilon) - V(x) \le 0$$
(10.7)

for all x and for all  $\varepsilon$ . Here,  $\gamma$  is a nonnegative constant. Then the inequality

$$||z||_{N}^{2} + ||u||_{N}^{2} \le V(x(0)) + \gamma^{2} ||\varepsilon||_{N}^{2}$$
(10.8)

holds for every signal  $\varepsilon$  and for every  $N \ge 0$  when u(t) = k(x(t)).

**Proof:** Since the equality

$$V(x(0)) - V(x(N+1)) + \sum_{k=0}^{N} (V(x(k+1)) - V(x(k))) = 0$$

holds for any function V, we have that

$$J_N - V(x(0)) = \sum_{t=0}^N \left( m(x(t))^T m(x(t)) + k(x(t))^T k(x(t)) - \gamma^2 \varepsilon(t)^T \varepsilon(t) \right) + V\left( f(x(t), k(x(t))) + n(x(t))\varepsilon(t) \right) - V(x(t)) \right) - V(x(N+1)) \le 0,$$

where we have used (10.7) and the fact that V is positive semidefinite in the last inequality. The fact that

$$J_N - V(x(0)) \le 0$$

implies that (10.8) holds for every  $\varepsilon$ .

Theorem 10.1 describes a discrete-time control design problem where a function k(x) should be found. For this control design problem to be solved, the inequality (10.7) must hold for all x and all  $\varepsilon$ . However, if V is assumed to be a quadratic function,  $V(x) = x^T P x$ , for some choice of a symmetric, positive semidefinite matrix P, it turns out that  $\varepsilon$  can be eliminated from (10.7). In this case, the robust control design problem involves finding a solution to an inequality for all x, i.e., to solve an easier problem. Using a quadratic V such that  $Q = (\gamma^2 I - n^T P n)$  is positive definite, we have

$$\begin{split} z^T z + u^T u - \gamma^2 \varepsilon^T \varepsilon + f^T P f + f^T P n \varepsilon + \varepsilon^T n^T P f + \varepsilon^T n^T P n \varepsilon - x^T P x \\ &= z^T z + u^T u + f^T P f + f^T P n Q^{-1} n^T P f - x^T P x \\ &- (\varepsilon - Q^{-1} n^T P f)^T Q (\varepsilon - Q^{-1} n^T P f) \\ &\leq z^T z + u^T u + f^T P f + f^T P n Q^{-1} n^T P f - x^T P x. \end{split}$$

Hence, the condition (10.7) in Theorem 10.1 is satisfied if a positive semidefinite matrix P and a function k(x) can be found such that  $Q(x, P) = \gamma^2 I - n(x)^T P n(x)$  is positive definite and

$$m(x)^{T}m(x) + k(x)^{T}k(x) + f(x,k(x))^{T}Pf(x,k(x)) + f(x,k(x))^{T}Pn(x)Q(x,P)^{-1}n(x)^{T}Pf(x,k(x)) - x^{T}Px \le 0$$
(10.9)

for all x.

Intuitively, the result (10.8) says that the gain of the closed-loop system from  $\varepsilon$  to z and u is less than  $\gamma$  whenever Theorem 10.1 can be applied. Since this theorem also implies that  $u(t) \in U$  for all  $t \ge 0$ , it can be used to prove the following result, which can be viewed as a version of the small gain theorem.

#### Theorem 10.2

Consider a system and a model such that the model error model (10.4) holds and assume that the robust control design problem in Theorem 10.1 has been solved for some  $\gamma \ge 0$ such that  $\gamma\beta < 1$ . If the controller u = k(x) from Theorem 10.1 is used for the true system, the obtained closed-loop system satisfies

$$\|z\|_N^2 \le V(x(0)) + \frac{\gamma^2 \alpha(N)^2}{1 - \gamma^2 \beta^2}.$$
(10.10)

**Proof:** Inserting the model error model (10.4) in (10.8) gives

$$||z||_{N}^{2} + ||u||_{N}^{2} \le V(x(0)) + \gamma^{2}(\alpha(N) + \beta ||u||_{N})^{2}.$$

By completing squares, this expression can be rewritten as

$$\begin{aligned} \|z\|_{N}^{2} + (1 - \gamma^{2}\beta^{2})(\|u\|_{N} - \frac{\gamma^{2}\alpha(N)\beta}{1 - \gamma^{2}\beta^{2}})^{2} &\leq V(x(0)) + \gamma^{2}\alpha(N)^{2} + \frac{\gamma^{4}\alpha(N)^{2}\beta^{2}}{1 - \gamma^{2}\beta^{2}} \\ &= V(x(0)) + \frac{\gamma^{2}\alpha(N)^{2}}{1 - \gamma^{2}\beta^{2}} \end{aligned}$$

and since  $\gamma\beta < 1$ , deleting the positive term on the left hand side gives the result in (10.10).

Theorem 10.2 shows that  $||z||_N^2$  has an upper bound that depends on the size of  $\alpha(N)$  in the model error model (10.4). If  $\alpha(N)$  is bounded,  $0 \le \alpha(N) \le \alpha_0$  for all  $N \ge 0$ , (10.10) implies that the true closed-loop system is stable in the sense that

$$\lim_{N \to \infty} \|z\|_N < \infty.$$

If  $\varepsilon(t)$  contains a signal component, which can be both a result of model errors and an external disturbance, with an upper bound at each time instant, the choice  $\alpha(N) = \alpha_0 \sqrt{N}$  is natural. For such an  $\alpha(N)$ , Theorem 10.2 can be used to show that the power of z(t) will be bounded.

A nice property of the controller u = k(x) is that the control signal is computed only from the states of the model. Hence, no observer has to be designed and the complete controller can be written as

$$\begin{aligned} x(t+1) &= f(x(t), k(x(t))) + n(x(t))(y(t) - h(x(t), k(x(t)))), \\ u(t) &= k(x(t)), \\ x(0) &= x_0 \end{aligned}$$

or, equivalently, as

$$\begin{aligned} x_{\mathcal{M}}(t+1) &= f_{\mathcal{M}}\left(x_{\mathcal{M}}(t), k\left((x_{\mathcal{M}}(t)^{T}, x_{\mathcal{W}}(t)^{T})^{T}\right)\right), \\ x_{\mathcal{W}}(t+1) &= f_{\mathcal{W}}\left(x_{\mathcal{W}}(t)\right) + g_{\mathcal{W}}\left(x_{\mathcal{W}}(t)\right) y(t), \\ u(t) &= k((x_{\mathcal{M}}(t)^{T}, x_{\mathcal{W}}(t)^{T})^{T}), \\ x_{\mathcal{M}}(0) &= x_{\mathcal{M},0}, \\ x_{\mathcal{W}}(0) &= x_{\mathcal{W},0}. \end{aligned}$$

However, it is important to verify that it is possible to compute the states in this controller with sufficiently high accuracy before using it in practice. For example, it should always be verified that the state-space model used in the controller is stable.

In the next section, robust control design for a nonlinear system based on an LTI model will be discussed.

#### 10.2 Robust Control Using LTI Models

In this section, the special case of LTI models with input saturation will be considered. Assume that (10.6) can be written as

$$x(t+1) = Ax(t) + Bu(t) + N\varepsilon(t),$$
  

$$y(t) = Cx(t) + Du(t) + \varepsilon(t),$$
  

$$z(t) = Mx(t),$$
  

$$x(0) = x_0,$$
  
(10.11)

where A, B, C, D, M and N are matrices. This implies that the condition (10.9), which can be used instead of (10.7) if

$$Q(P) = \gamma^2 I - N^T P N$$

is positive definite, can be written

$$x^{T}M^{T}Mx + u^{T}u + x^{T}A^{T}PAx + x^{T}A^{T}PBu + u^{T}B^{T}PAx + u^{T}B^{T}PBu + (Ax + Bu)^{T}PNQ(P)^{-1}N^{T}P(Ax + Bu) - x^{T}Px \le 0.$$
 (10.12)

Furthermore, this condition can be rewritten as

$$x^{T}(A^{T}PA - P + M^{T}M + A^{T}PNQ(P)^{-1}N^{T}PA - T(P)^{T}W(P)^{-1}T(P))x + (u + W(P)^{-1}T(P)x)^{T}W(P)(u + W(P)^{-1}T(P)x) \le 0,$$
(10.13)

where

$$W(P) = I + B^T P B + B^T P N Q(P)^{-1} N^T P B,$$
  

$$T(P) = B^T P A + B^T P N Q(P)^{-1} N^T P A.$$

Consider the case of a single input single output system and let

$$u(t) = k(x(t)) \triangleq \begin{cases} -u_0, & W(P)^{-1}T(P)x(t) > u_0, \\ -W(P)^{-1}T(P)x(t), & |W(P)^{-1}T(P)x(t)| \le u_0, \\ u_0, & W(P)^{-1}T(P)x(t) < -u_0, \end{cases}$$
(10.14)

where  $u_0 > 0$  is a constant such that  $U = \{v \in \mathbb{R} \mid |v| \le u_0\}$ . When

$$W(P)^{-1}T(P)x(t) > u_0$$

we obtain the condition

$$x^{T}(A^{T}PA - P + M^{T}M + A^{T}PNQ(P)^{-1}N^{T}PA - T(P)^{T}W(P)^{-1}T(P))x + (-u_{0} + W(P)^{-1}T(P)x)^{T}W(P)(-u_{0} + W(P)^{-1}T(P)x) = x^{T}(A^{T}PA - P + M^{T}M + A^{T}PNQ(P)^{-1}N^{T}PA)x + u_{0}^{2}W(P) - 2u_{0}T(P)x \le 0$$
(10.15)

by inserting the expression for the control signal in (10.14) into (10.13). Since

$$u_0^2 W(P) - 2u_0 T(P)x = u_0 W(P)(u_0 - 2W(P)^{-1}T(P)x) \le 0,$$

condition (10.15) is satisfied if

$$x^T (A^T P A - P + M^T M + A^T P N Q(P)^{-1} N^T P A) x \le 0.$$

Using, similar calculations, this result can also be obtained when

$$W(P)^{-1}T(P)x(t) < -u_0.$$

If a positive definite matrix P such that

$$A^T P A - P + M^T M + A^T P N Q(P)^{-1} N^T P A \preccurlyeq 0, \qquad (10.16a)$$

$$Q(P) = \gamma^2 - N^T P N > 0$$
 (10.16b)

can be found, the conditions in Theorem 10.1 are thus satisfied for  $V(x) = x^T P x$  and the control law in (10.14). Here,  $\preccurlyeq 0$  means negative semidefiniteness of the left hand side. Hence, for an LTI model, a solution to the robust control design problem can be found by solving the Riccati inequality (10.16a) and checking that the solution satisfies (10.16b). This approach is illustrated in the following example.

#### \_\_\_\_\_ Example 10.1 \_\_\_\_\_

Consider the slightly nonlinear system

$$\begin{split} y(t) &= y_l(t) + \delta y_n(t) + w(t), \\ y_l(t) &= G_L(q)u(t) = \frac{2q-1}{q^2 - q + 0.2}u(t), \\ y_n(t) &= u(t)^3, \end{split}$$

where  $\delta = -1/15$  is a constant and w(t) is a disturbance. Assume that the linear transfer function  $G_L(q)$  is known and that we want to use it for robust control design where the objective is to reduce the effects of low-frequent disturbances w(t) on y(t). This control objective is formulated using the filter

$$z(t) = W(q)y(t) = \frac{0.2}{q - 0.9}y(t).$$

The model  $G_L(q)$  and the filter W(q) give the complete state-space model

$$\begin{split} x(t+1) &= Ax(t) + Bu(t) + N\varepsilon(t), \\ y(t) &= Cx(t) + \varepsilon(t), \\ z(t) &= Mx(t), \\ x(0) &= 0, \end{split}$$

where

$$A = \begin{pmatrix} 1.0 & -0.4 & 0\\ 0.5 & 0 & 0\\ 0.2 & -0.2 & 0.9 \end{pmatrix},$$
$$B = \begin{pmatrix} 2\\ 0\\ 0 \\ 0 \end{pmatrix},$$
$$N = \begin{pmatrix} 0\\ 0\\ 0.2 \end{pmatrix},$$
$$C = \begin{pmatrix} 1 & -1 & 0 \end{pmatrix},$$
$$M = \begin{pmatrix} 0 & 0 & 1 \end{pmatrix}.$$

For example, with  $\gamma = 2.1$ , the matrix

$$P = \begin{pmatrix} 2.11 & -1.67 & 3.64 \\ -1.67 & 1.34 & -2.95 \\ 3.64 & -2.95 & 7.95 \end{pmatrix}$$

is one solution to the Riccati inequality (10.16a) obtained for this state-space model. Furthermore, this choice of P satisfies (10.16b). This solution has been found using MAT-LAB and the Riccati equation solver dare. With  $G_L(q)$  as model, the gain of the nonlinear model error is  $\beta = u_0^2/15$ . In order to achieve  $\gamma\beta < 1$ , the input has to be saturated.

For example, the saturation can be at  $u_0 = 2.5$  since this value gives the gain

$$\beta = \frac{5}{12} < \frac{10}{21} = \frac{1}{\gamma}$$

of the model error. The resulting control law is

$$u(t) = k(x(t)) = \begin{cases} -2.5, & Lx(t) > 2.5, \\ -Lx(t), & |Lx(t)| \le 2.5, \\ 2.5, & Lx(t) < -2.5, \end{cases}$$

where

$$L = (0.429 \quad -0.337 \quad 0.709)$$

and the complete controller can be written

$$\begin{aligned} x(t+1) &= Ax(t) + Bk(x(t)) + N(y(t) - Cx(t)), \\ u(t) &= k(x(t)), \\ x(0) &= 0. \end{aligned}$$

The closed-loop system that is obtained when this controller is used for the true nonlinear system has been simulated using MATLAB. In this simulation, the true system was initialized at zero and the disturbance w(t) was a unit step at t = 5, i.e.,

$$w(t) = \begin{cases} 1, & t \ge 5, \\ 0, & t < 5. \end{cases}$$

This corresponds to a function

$$\alpha(N) = \begin{cases} \sqrt{N-4}, & N \ge 5, \\ 0, & N < 5 \end{cases}$$

in the model error model. The output signal from the simulation experiment is shown in Figure 10.1. As can be seen in this figure, the feedback controller reduces the stationary effect of the disturbance with more than 60%. Furthermore, from the design of the controller, we know that there is no risk of instability due to the model errors.

Example 10.1 shows that an input saturation might be needed to limit the size of the model error in a nonlinear closed-loop system. In this example, removing the restrictions on the input signal would result in a model error with infinite gain.



**Figure 10.1:** The output signal from the closed-loop system in Example 10.1 when a unit step disturbance is added to the output at t = 5.

#### 10.3 Discussion

In this chapter, some issues concerning robust control design have been discussed. In particular, a gain definition valid only for a restricted set of input signals has been presented. In some cases, this gain definition gives less pessimistic results than the standard gain definition concerning, for example, robust control using LTI models of nonlinear systems. It has been shown that a robust control law can be found by solving (10.7) in Theorem 10.1 for a  $\gamma > 0$  that is smaller than the inverse of the gain of the model error for restricted inputs. Hence, the main difference between the approach presented here and an approach based on the small gain theorem is the use of the alternative gain definition.

Furthermore, the special case of LTI models of nonlinear systems has been investigated. In this case, the robust control design problem can be solved by finding a solution to the Riccati inequality (10.16a). The method presented here for robust control using linear models of nonlinear systems illustrates one general feature of this type of problems, namely that some restrictions on the input signal typically have to be added in order to reduce the size of the nonlinear model error.

# 11

## Mixed Parametric Nonparametric Identification

The OE-LTI-SOE of a nonlinear system is defined as the LTI model that minimizes the mean-square error and considering only this criterion, there is thus no better LTI approximation. However, the mean-square error is not always a suitable measure of how good or accurate a model is. In the previous chapters, it has been shown that the OE-LTI-SOE of a nonlinear system sometimes can be very sensitive to small nonlinearities. Since this effect is usually undesirable, it is interesting to investigate how it can be reduced. If the input signal to the system can be modified, a careful input design can give an OE-LTI-SOE that is less sensitive to small nonlinearities. For example, as was discussed in Chapter 6, a separable input signal might be useful in some cases. However, if the user is not free to design the input signal, other ways of improving the LTI approximation are necessary.

Of course, one approach for obtaining an LTI approximation of a nonlinear system is to estimate first a nonlinear model. When an accurate nonlinear model of the system has been found, it can be linearized analytically to provide the user with both an LTI model and a detailed mathematical description of the model errors. However, estimating a nonlinear model of a system can be a challenging task where the user has many options. There are various methods for nonlinear system identification and usually a number of nonlinear model structures that can be used with each method (see, for example, Giannakis and Serpedin, 2001).

The use of linear models of complex, nonlinear systems is common in many real-life applications of system identification. For example, tuning of control loops in process industry is often done based on approximate linear models. The reason why nonlinear models are not desired can be that such models are considered too time consuming to estimate. Hence, there are applications where an improved method designed for estimation of LTI models of nonlinear systems might be useful.

In the standard prediction-error method, the obtained estimates of the model parameters are such that the correlation between the model residuals and the past and present input signal components is minimized. Actually, this correlation is zero for the OE-LTI- SOE (cf. Corollary 4.2). However, if there are small nonlinearities in the system, it might be better not to try to model all correlation between the input and the output but to allow for a small correlated term in the model residuals. This idea was proposed in Mäkilä (2005), but no method was presented there. Here, however, we will discuss one such method for identification of nonlinear autoregressive systems with external inputs (NARX systems) (Sjöberg et al., 1995). This method is described also in Roll et al. (2005a).

NARX systems are a straightforward generalization of linear ARX systems that has been used in many applications. For an NARX system, the mean-square error optimal one step ahead predictor is a nonlinear function of a finite number of past output and past and present input components. Using a version of the prediction-error method (Ljung, 1999), we will here simultaneously estimate both a nonparametric NARX model and a parametric ARX model such that their sum give an as good prediction of the output as possible. Related model structures have been used in semiparametric or partially linear models (see, for example, Heckman, 1988; Chen et al., 2001).

The proposed method can be viewed both as a way to handle insignificant nonlinearities when a linear model is estimated and as a method for nonlinear system identification. A nonparametric nonlinear model will always be estimated but can be ignored if only a linear model is desired. It is interesting to consider nonparametric methods for nonlinear system identification since the assumptions about the true system are usually weaker for such methods than for parametric methods. For a nonlinear system, it can be hard to tell in advance whether a specific assumption about, for example, the shape of the nonlinearities is reasonable or not. Here, the only assumption about the true NARX system is that its nonlinearities are Lipschitz continuous.

This assumption makes it possible to use an approach where the identification problem is formulated as a quadratic programming (QP) problem. By solving this problem, both the parameters of the linear ARX model and the nonparametric NARX model can be estimated at the same time. A version of this idea, without the linear, parametric part, has previously been used for nonparametric regression and for maximum likelihood estimation of unknown parameters in probability density functions (Bertsimas et al., 1999). Other methods for nonparametric regression can be found in, for example, Fan and Gijbels (1996). Lipschitz conditions are a common way to guarantee that a function, or some of its derivatives, will be smooth. For example, functions with a Lipschitz continuous gradient can be identified using local modeling such that the worst-case mean-square error is minimized (Roll et al., 2005b).

The method presented in this chapter can sometimes make the estimate of the linear model more robust against nonlinearities in the system since the nonparametric NARX model can compensate for some of the nonlinear effects. A related concept is the notion of unknown but bounded noise and set membership identification (Garulli et al., 1999), since a bounded nonlinearity might affect the system output in a similar way as such a noise term.

#### 11.1 Identification of NARX and NOE Systems

Consider an NARX system with input u(t) and output y(t) that can be written as

$$y(t) = \theta_0^T \varphi(t) + r_0(\varphi(t)) + e(t),$$
 (11.1)

where

$$\varphi(t) = \begin{pmatrix} -y(t-1) \\ \vdots \\ -y(t-n_a) \\ u(t-n_k) \\ \vdots \\ u(t-n_k-n_b) \end{pmatrix}$$
(11.2)

is a regression vector and e(t) is white noise. The constant vector  $\theta_0$  defines a linear ARX part of the system while the function  $r_0$  can be nonlinear. Assume that  $(e(t))_{t=-\infty}^{\infty}$  and  $(\varphi(t))_{t=-\infty}^{\infty}$  are stationary stochastic processes such that the signal components e(s) and  $\varphi(s)$  are independent for every  $s \in \mathbb{Z}$  and that  $r_0$  is a Lipschitz continuous function with Lipschitz constant  $L_0$ , i.e., that

$$|r_0(\varphi_1) - r_0(\varphi_2)| \le L_0 |\varphi_1 - \varphi_2|, \ \forall \varphi_1, \varphi_2 \in \mathbb{R}^n, \tag{11.3}$$

where  $n = n_a + n_b + 1$ . Furthermore, assume that a data set  $(\varphi(t), y(t))_{t=1}^N$  consisting of N measurements of the regression vector and the system output is available.

Using this data set, estimates  $\hat{\theta}_N$  and  $\hat{r}_N$  of  $\theta_0$  and  $r_0$ , respectively, can be obtained by solving the QP problem

$$\begin{array}{ll} \underset{\theta_N,\rho_N}{\text{minimize}} & \frac{1}{N} \sum_{t=1}^{N} (y(t) - \theta_N^T \varphi(t) - \rho_N(t))^2 \\ \text{subject to} & \rho_N(t) - \rho_N(s) \leq L |\varphi(t) - \varphi(s)|, \\ & \forall s, t \in \{1, 2, \dots, N\} \\ & \pm \theta_{N,i} \leq m_{\theta,i}, \\ & \forall i \in \{1, 2, \dots, n_a + n_b + 1\}. \end{array}$$

$$(11.4)$$

In this problem,  $\theta_N$  is a vector with  $n_a + n_b + 1$  elements  $\theta_{N,i}$  and  $\rho_N$  is a vector with N elements  $\rho_N(t)$  that can be viewed as estimates of  $r_0(\varphi(t))$ . The constraints on the variables  $\rho_N(t)$  imply that these variables will satisfy

$$|\rho_N(t) - \rho_N(s)| \le L|\varphi(t) - \varphi(s)|$$

for all  $s, t \in \{1, 2, ..., N\}$ . If the variables  $\rho_N(t)$  are viewed as samples from some function, this means that a Lipschitz condition holds for the sample points  $(\varphi(t))_{t=1}^N$ . Note that N of the constraints on  $\rho_N$  in (11.4) are trivial ( $0 \le 0$ ) and present in (11.4) only for notational convenience. These constraints can be removed without changing the solution of the problem. Furthermore, the constraints on  $\theta_N$  are needed mainly when consistency is proved and can be removed in most applications. In these cases, the QP problem

$$\begin{array}{ll} \underset{\theta_N,\rho_N}{\text{minimize}} & \frac{1}{N} \sum_{t=1}^{N} (y(t) - \theta_N^T \varphi(t) - \rho_N(t))^2 \\ \text{subject to} & \rho_N(t) - \rho_N(s) \leq L |\varphi(t) - \varphi(s)|, \\ & \forall s, t \in \{1, 2, \dots, N\}, \end{array}$$
(11.5)

which has fewer constraints, can be solved instead of (11.4).

An optimal solution  $(\hat{\theta}_N, \hat{\rho}_N)$  to the problem (11.4) or (11.5) can be used to construct one step ahead predictions

$$\hat{y}_N(t) = \hat{\theta}_N^T \varphi(t) + \hat{\rho}_N(t)$$
(11.6)

of the system output for the observed regression vectors  $(\varphi(t))_{t=1}^N$ . In order to obtain a predictor that can be used for an arbitrary regression vector, the nonparametric function estimate  $\hat{\rho}_N$  has to be interpolated.

When  $\varphi(t)$  is a scalar, linear interpolation is probably the most natural type of interpolation. However, for  $\varphi(t) \in \mathbb{R}^n$  with n > 1, linear interpolation of the variables  $\hat{\rho}_N(t)$ will in general not result in a function that satisfies the Lipschitz condition for the choice of L used in (11.4). Instead, for n > 1, an estimate  $\hat{r}_N$  of  $r_0$  can be defined as

$$\hat{r}_N(\varphi) = \frac{1}{2} \max_{1 \le t \le N} (\hat{\rho}_N(t) - L|\varphi - \varphi(t)|) + \frac{1}{2} \min_{1 \le t \le N} (\hat{\rho}_N(t) + L|\varphi - \varphi(t)|) \quad (11.7)$$

using a similar construction as in Bertsimas et al. (1999). The function  $\hat{r}_N$  is Lipschitz continuous since it is the mean of two Lipschitz continuous functions. Of course, other interpolation methods that can be used instead of (11.7) as long as they guarantee that the resulting function will be Lipschitz continuous with the Lipschitz constant L. Using  $\hat{\theta}_N$  and  $\hat{r}_N$ , a general one step ahead predictor function

$$\hat{f}_N(\varphi) = \hat{\theta}_N^T \varphi + \hat{r}_N(\varphi) \tag{11.8}$$

can be constructed.

At first sight, it might seem that the N + n variables used in the problem (11.4) and for the construction of the model (11.8) are too many since there are only N measurements. However, thanks to the randomness of the disturbance e(t) in (11.1), the constraints in (11.4) will impose an averaging effect on the nonparametric function estimate. Without these constraints, one optimal solution to (11.4) is  $\theta_N = 0$ ,  $\rho_N(t) = y(t)$  for  $t = 1, 2, \ldots, N$ . Of course, since the measurements of the output are noisy, such a solution does not give a good model of the true system. By adding constraints like in (11.4), two variables  $\rho_N(t)$  and  $\rho_N(s)$  are allowed to differ only marginally from each other if the distance  $|\varphi(t) - \varphi(s)|$  between the corresponding regression vectors is small. In this way, the  $\rho$  variables are imposed to have similar properties as samples from the true Lipschitz continuous function  $r_0$ . If the set of regression vectors gets more dense when N increases,  $\hat{\theta}_N^T \varphi(t) + \hat{\rho}_N(t)$  will approach  $\theta_0^T \varphi(t) + r_0(\varphi(t))$ . For an intuitive understanding of this convergence, consider a small region in  $\mathbb{R}^n$  which contains many regression vectors. The corresponding  $\rho$  variables will with a high probability be close to the mean of  $y(t) - \theta_N^T \varphi(t)$  since the constraints in (11.4) implies that the  $\rho$  variables should have values close to each other. The consistency of the predictor function estimator (11.8) will be discussed in Section 11.2.

Several types of extensions can be made to the identification method presented here. For example, if any prior knowledge about the true system can be written as linear constraints on  $\theta_N$  and  $\rho_N$ , this knowledge can easily be incorporated in the QP problem (11.4). Examples of such prior knowledge are:

• Bounds on the function  $r_0$  are known in a subset of its domain.

- The function  $r_0$  is only piecewise Lipschitz continuous (with known breakpoints).
- The function  $r_0$  satisfies different Lipschitz conditions in different parts of its domain.
- The function  $r_0$  is known to be odd or even.
- An expression for the function  $r_0$  is known in a subset of its domain.

Sometimes it could be interesting to estimate only a Lipschitz continuous function. Such an estimate can be obtained by setting all  $m_{\theta,i} = 0$  in (11.4). Hence, a nonparametric function estimate can be calculated by solving the QP problem

$$\begin{array}{ll} \underset{\rho_N}{\text{minimize}} & \frac{1}{N} \sum_{t=1}^{N} (y(t) - \rho_N(t))^2 \\ \text{subject to} & \rho_N(t) - \rho_N(s) \le L |\varphi(t) - \varphi(s)|, \\ & \forall s, t \in \{1, 2, \dots, N\}. \end{array}$$
(11.9)

The construction of  $\hat{r}_N$  using the interpolation method (11.7) can be used also in this case.

An advantage with the presented method is that the underlying optimization problem (11.4) is convex (Boyd and Vandenberghe, 2004). This convexity follows from the fact that only NARX models are considered. However, OE models have been the standard choice in the earlier chapters of this thesis and thus it would be nice if the mixed parametric and nonparametric method could be generalized to handle this case as well. It turns out that such a generalization is easy to formulate, but that it gives a nonconvex optimization problem. This problem can be formulated as

$$\begin{array}{ll} \underset{\theta_{N},\rho_{N}}{\text{minimize}} & \frac{1}{N} \sum_{t=1}^{N} (y(t) - G(q,\theta_{N})u(t) - \rho_{N}(t))^{2} \\ \text{subject to} & \rho_{N}(t) - \rho_{N}(s) \leq L |\varphi(t) - \varphi(s)|, \\ & \forall s, t \in \{1, 2, \dots, N\} \\ & \theta_{N} \in \mathcal{D}_{\theta}, \end{array} \tag{11.10}$$

where  $\varphi(t)$  now only contains input components,  $G(q, \theta_N)$  is a parameterized LTI model and  $\mathcal{D}_{\theta}$  is a set of parameters that give stable and causal models. The model in this problem has a particular nonlinear output error (NOE) structure such that the model output can be written as the sum of the outputs from a linear OE model and an NFIR model.

In most cases, the optimization problem (11.10) is nonconvex and thus, the problem of finding its global minimum seems hard. However, it is easy to use a procedure where the problems

A: Keep  $\rho_N$  fixed and estimate the parameters  $\theta_N$  by solving (11.10) for these variables.

**B:** Keep  $\theta_N$  fixed and estimate the parameters  $\rho_N$  by solving (11.10) for these variables.

are solved iteratively until convergence. Problem A is a standard OE modeling problem while problem B is equivalent to an NFIR version of (11.9). The method can, for example, be initiated with  $\rho_N = 0$ . Usually, this procedure will give a model that corresponds to a local minimum of (11.10), but we will not analyze the convergence properties of this method here. However, the consistency of the presented nonparametric identification methods based on (11.4) and (11.9) will be shown in the next section.

#### 11.2 Consistency

The identification methods given by (11.4) and (11.9), respectively, turn out to have rather attractive asymptotic properties when the number of measurements tends to infinity. In this section, it will be shown that the prediction function estimator defined by the solution to the optimization problem (11.4) under fairly general conditions is consistent. Since (11.9) is a special case of (11.4), the consistency of the nonparametric function estimator without a linear part follows from the more general case.

A related consistency result for the estimator defined by (11.9) is shown in Bertsimas et al. (1999) using results from Vapnik (1998). However, the result in the following theorem is based on different assumptions and is shown using an alternative proof.

#### Theorem 11.1

Consider data sets  $(\varphi(t), y(t))_{t=1}^N$  generated from the nonlinear system

$$y(t) = \theta_0^T \varphi(t) + r_0(\varphi(t)) + e(t) \triangleq f_0(\varphi(t)) + e(t), \qquad (11.11)$$

where e(t) is a white stationary stochastic process with zero mean and bounded variance  $\sigma^2$  and where  $\varphi(t)$  is a stationary stochastic process. For each data set, let  $\hat{\theta}_N$  and  $\hat{\rho}_N(t)$  be the optimal solution to (11.4). Furthermore, let  $\hat{f}_N$  be the predictor function given by this solution, i.e.,

$$\hat{f}_N(\varphi) = \hat{\theta}_N^T \varphi(t) + \hat{r}_N(\varphi),$$

where  $\hat{r}_N$  is defined in (11.8). Suppose that

1.  $\varphi(t) \in \Phi$ , where  $\Phi$  is a compact set such that the probability density function  $p(\varphi)$ for  $\varphi(t)$  is positive for all  $\varphi \in \Phi$  and that for any  $\varepsilon > 0$ ,  $\Phi$  can be partitioned

$$\Phi = \bigcup_{i=1}^{d} \Phi_i, \tag{11.12}$$

where  $\varphi_1, \varphi_2 \in \Phi_i \Rightarrow |\varphi_1 - \varphi_2| \leq \varepsilon$  and  $p_i = P(\varphi(t) \in \Phi_i) > 0$  for all  $i = 1, 2, \ldots, d$ ,

2. the stochastic process  $\varphi(t)$  is such that  $N_i/N \to p_i$  when  $N \to \infty$  w.p.1 for all i in any  $\varepsilon$ -partitioning (11.12) where

$$N_i = \operatorname{card}(T_i) \text{ and } T_i = \{t \mid \varphi(t) \in \Phi_i, \ t \le N\},$$
(11.13)

- *3.* e(t) and  $\varphi(t)$  are independent, but  $\varphi(t)$  may depend on past e(s),
- 4.  $|r_0(\varphi_1) r_0(\varphi_2)| \leq L|\varphi_1 \varphi_2|$  for all  $\varphi_1, \varphi_2 \in \Phi$ ,
- 5.  $|f_0(\varphi_1) f_0(\varphi_2)| \leq \tilde{L}|\varphi_1 \varphi_2|$  for all  $\varphi_1, \varphi_2 \in \Phi$ , where  $\tilde{L} = L + M_{\theta}$  and

$$M_{\theta}^{2} = \sum_{i=1}^{n_{a}+n_{b}+1} m_{\theta,i}^{2}.$$

Then

$$\lim_{N \to \infty} \frac{1}{N} \sum_{t=1}^{N} (\hat{f}_N(\varphi(t)) - f_0(\varphi(t)))^2 = 0 \quad w.p.1$$
(11.14)

and

$$\hat{f}_N(\varphi) \to f_0(\varphi)$$
 uniformly on  $\Phi$  as  $N \to \infty$  w.p.1. (11.15)

**Proof:** Take an arbitrary  $\varepsilon > 0$  and consider an  $\varepsilon$ -partitioning such that the first assumption is satisfied. Let  $I_{\Phi_i}(\varphi)$  be the indicator function for the set  $\Phi_i$ , i.e.,

$$I_{\Phi_i}(\varphi) = \begin{cases} 1, & \varphi \in \Phi_i, \\ 0, & \text{otherwise.} \end{cases}$$

Consider arbitrary realizations of the processes  $\varphi(t)$  and e(t). With probability one, these realizations are such that  $N_i/N \to p_i$  as  $N \to \infty$  and that

$$\lim_{N \to \infty} \frac{1}{p_i N} \sum_{t \in T_i} e(t) = \lim_{N \to \infty} \frac{1}{p_i N} \sum_{t=1}^N I_{\Phi_i}(\varphi(t)) e(t) = 0,$$
(11.16a)

$$\lim_{N \to \infty} \frac{1}{p_i N} \sum_{t \in T_i} |e(t)| = \lim_{N \to \infty} \frac{1}{p_i N} \sum_{t=1}^N I_{\Phi_i}(\varphi(t)) |e(t)| \le C,$$
(11.16b)

$$\lim_{N \to \infty} \frac{1}{N} \sum_{t=1}^{N} e(t)^2 = \sigma^2$$
 (11.16c)

for some constant C and for all i = 1, 2, ..., d. The limits (11.16) follow from the law of large numbers. Let

$$\tilde{e}(t) = I_{\Phi_i}(\varphi(t))e(t),$$

and consider two arbitrary different time instants s and t. Without loss of generality, we can assume that s > t. Since  $\varphi(t)$ ,  $\varphi(s)$  and e(t) are all independent of e(s), this implies that

$$\mathbf{E}(\tilde{e}(t)\tilde{e}(s)) = \mathbf{E}(I_{\Phi_i}(\varphi(t))e(t)I_{\Phi_i}(\varphi(s)))\mathbf{E}(e(s)) = 0,$$

i.e., that  $\tilde{e}(t)$  and  $\tilde{e}(s)$  are uncorrelated. Similarly, with

$$e^{*}(t) = I_{\Phi_{i}}(\varphi(t))(|e(t)| - \mathrm{E}(|e(t)|)),$$

 $e^*(t)$  and  $e^*(s)$  can be shown to be uncorrelated. Furthermore, the variances of  $\tilde{e}(t)$  and  $e^*(t)$  are finite. Hence, the version of the strong law of large numbers in Theorem 5.1.2 in Chung (1974) imply that (11.16a) and (11.16b) hold. The convergence of the series in (11.16c) follows from the strong law of large numbers for independent variables (Chung, 1974, Theorem 5.4.2).

For two fixed realizations of  $\varphi(t)$  and e(t) for which (11.16) holds, we can thus find

an  $N'(\varepsilon)$  such that

$$\left| \frac{1}{p_i N} \sum_{t \in T_i} e(t) \right| \le \varepsilon, \quad \forall i \in \{1, 2, \dots, d\},$$
(11.17a)

$$\frac{1}{p_i N} \sum_{t \in T_i} |e(t)| \le 2C, \quad \forall i \in \{1, 2, \dots, d\},$$
(11.17b)

$$\frac{1}{N}\sum_{t=1}^{N} e(t)^2 \le 2\sigma^2$$
(11.17c)

for all  $N > N'(\varepsilon)$ . This result follows since the partitioning is finite for any  $\varepsilon$ .

The fourth and fifth assumption in the theorem imply that

$$f_N(\varphi(t)) \triangleq \theta_N^T \varphi(t) + \rho_N(t) = f_0(\varphi(t))$$

is a feasible choice of function in the optimization problem (11.4), either with  $\rho_N(t) = r_0(\varphi(t))$  and  $\theta_N = \theta_0$  or sometimes with some smaller  $\theta_N$  and larger  $\rho_N(t)$ :s. Hence, we have

$$\frac{1}{N}\sum_{t=1}^{N}(y(t) - \hat{f}_N(\varphi(t)))^2 \le \frac{1}{N}\sum_{t=1}^{N}(y(t) - f_0(\varphi(t))^2) = \frac{1}{N}\sum_{t=1}^{N}e(t)^2,$$

which means that

$$\frac{1}{N}\sum_{t=1}^{N}(f_0(\varphi(t)) - \hat{f}_N(\varphi(t)))^2 \le \left|\frac{2}{N}\sum_{t=1}^{N}e(t)(f_0(\varphi(t)) - \hat{f}_N(\varphi(t)))\right|.$$
 (11.18)

Note first that, by applying Cauchy-Schwarz inequality to the right hand side, we obtain that

$$\frac{1}{N}\sum_{t=1}^{N}(f_0(\varphi(t)) - \hat{f}_N(\varphi(t)))^2 \le \frac{4}{N}\sum_{t=1}^{N}e(t)^2.$$

Since  $f_0$  is bounded by  $C_{f_0} = \sup_{\varphi \in \Phi} |f_0(\varphi)|$  and (11.17c) holds for all  $N > N'(\varepsilon)$ ,  $\hat{f}_N(\varphi)$  as defined by (11.7) and (11.8) must be bounded too. Hence, we can choose a constant  $C_{\hat{f}}$ , such that for  $N > N'(\varepsilon)$  we have  $C_{\hat{f}} > \sup_{\varphi \in \Phi} |\hat{f}_N(\varphi)|$ .

Since  $N_i \to \infty$  for all *i*, there will be many  $\varphi(t)$  in every set  $\Phi_i$  in the  $\varepsilon$ -partitioning. Choose  $t_i^* \in T_i$  and let

$$f_i = f_0(\varphi(t_i^*)),$$
$$\hat{f}_{N,i} = \hat{f}_N(\varphi(t_i^*)).$$

This means that for  $t \in T_i$ , it holds that

$$|f_0(\varphi(t)) - f_i| \le \tilde{L}\varepsilon$$

and

$$|\hat{f}_N(\varphi(t)) - \hat{f}_{N,i}| \le \tilde{L}\varepsilon.$$

Inserting this into the expression (11.18) gives

$$\begin{split} &\frac{1}{N}\sum_{t=1}^{N}(f_{0}(\varphi(t))-\hat{f}_{N}(\varphi(t)))^{2} \\ &\leq \left|\frac{2}{N}\sum_{t=1}^{N}e(t)(f_{0}(\varphi(t))-\hat{f}_{N}(\varphi(t)))\right| \\ &= \left|\frac{2}{N}\sum_{i=1}^{d}\sum_{t\in T_{i}}e(t)(f_{0}(\varphi(t))-f_{i}+f_{i}-\hat{f}_{N}(\varphi(t))+\hat{f}_{N,i}-\hat{f}_{N,i})\right. \\ &= \left|2\sum_{i=1}^{d}p_{i}\left(\left(\frac{1}{p_{i}N}\sum_{t\in T_{i}}e(t)\right)\left(f_{i}-\hat{f}_{N,i}\right)\right. \\ &+ \left(\frac{1}{p_{i}N}\sum_{t\in T_{i}}e(t)(f_{0}(\varphi(t))-f_{i}-\hat{f}_{N}(\varphi(t))+\hat{f}_{N,i})\right)\right)\right| \\ &\leq 2\sum_{i=1}^{d}p_{i}\left(\varepsilon\max_{i}|f_{i}-\hat{f}_{N,i}|+\frac{1}{p_{i}N}\sum_{t\in T_{i}}|e(t)|2\tilde{L}\varepsilon\right) \\ &\leq C'\varepsilon \qquad \text{for } N > N'(\varepsilon), \end{split}$$

where  $C' = 2C_{f_0} + 2C_{\hat{f}} + 8\tilde{L}C$ . Since  $\varepsilon$  and the realizations were arbitrary, (11.14) has been proven.

We will now prove that the result (11.14) implies the uniform convergence in (11.15). First, we will consider pointwise convergence. Consider arbitrary realizations of  $\varphi(t)$  and e(t). With probability one these realizations are such that the second assumption in the theorem is satisfied and that the convergence in (11.14) holds. Consider two arbitrary realizations where these limits hold and assume that  $\hat{f}_N$  does not converge pointwise to  $f_0(\varphi)$  on  $\Phi$ , i.e., that there exists a  $\varphi_0$  in  $\Phi$ , a  $\delta > 0$  and an infinite strictly increasing sequence of integers  $K_j$ ,  $j \in \mathbb{Z}_+$ , such that

$$|\hat{f}_{K_j}(\varphi_0) - f_0(\varphi_0)| > \delta$$

for all j. Consider a  $\delta/4\hat{L}$ -partitioning of  $\Phi$  such that the two first assumptions are satisfied, i.e., a partitioning where

$$|\varphi_1, \varphi_2 \in \Phi_i \Rightarrow |\varphi_1 - \varphi_2| < \frac{\delta}{4\tilde{L}}$$

and where  $N_i/N \to p_i$  for all i = 1, 2, ..., d. Consider the set  $\Phi_l$  that contains  $\varphi_0$ . For every  $\varphi$  in  $\Phi_l$ , it holds that

$$\begin{aligned} |\hat{f}_{K_{j}}(\varphi) - f_{0}(\varphi)| &\geq |\hat{f}_{K_{j}}(\varphi_{0}) - f_{0}(\varphi_{0})| - |\hat{f}_{K_{j}}(\varphi) - \hat{f}_{K_{j}}(\varphi_{0})| - |f_{0}(\varphi_{0}) - f_{0}(\varphi)| \\ &\geq \delta - \tilde{L}\frac{\delta}{4\tilde{L}} - \tilde{L}\frac{\delta}{4\tilde{L}} = \frac{\delta}{2}. \end{aligned}$$
(11.19)

Furthermore, from the second assumption in the theorem, it holds that  $K_{j,l}/K_j \rightarrow p_l$ when  $j \rightarrow \infty$ , where  $K_{j,l}$  is the number of  $\varphi(t)$  in  $\Phi_l$  when the total number of measurements is  $K_j$ . The convergence

$$\lim_{N \to \infty} \frac{1}{N} \sum_{t=1}^{N} (\hat{f}_N(\varphi(t)) - f_0(\varphi(t)))^2 = 0$$

implies that

$$\lim_{j \to \infty} \frac{1}{K_j} \sum_{t=1}^{K_j} (\hat{f}_{K_j}(\varphi(t)) - f_0(\varphi(t)))^2 = 0.$$

However, using (11.19), it follows that

T /

$$\frac{1}{K_j} \sum_{t=1}^{K_j} (\hat{f}_{K_j}(\varphi(t)) - f_0(\varphi(t)))^2 \ge \frac{1}{K_j} \sum_{t \in T_l} (\hat{f}_{K_j}(\varphi(t)) - f_0(\varphi(t)))^2 \\\ge \frac{1}{K_j} \sum_{t \in T_l} \frac{\delta^2}{4} = \frac{\delta^2 K_{j,l}}{4K_j}.$$

Since

$$\frac{\delta^2 K_{j,l}}{4K_j} \to \frac{\delta^2 p_l}{4} > 0, \quad j \to \infty,$$

we have a contradiction. Thus,  $\hat{f}$  must converge pointwise to  $f_0$  on  $\Phi$ .

It turns out that pointwise convergence gives uniform convergence in this case. Take an arbitrary  $\tilde{\varepsilon} > 0$  and assume that  $\tilde{f}_N(\varphi)$  converges pointwise to  $f_0(\varphi)$  on  $\Phi$ . Select a finite number of points  $\tilde{\varphi}_k$ ,  $k = 1, 2, ..., d_{\tilde{\varepsilon}}$  in  $\Phi$  such that for every point  $\varphi$  in  $\Phi$ ,

$$|\varphi - \tilde{\varphi}_k| < \frac{\tilde{\varepsilon}}{3\tilde{L}}$$

for some k. Choose an  $N_{\tilde{\varepsilon}}$  such that for all k it holds that

$$|\hat{f}_N(\tilde{\varphi}_k) - f_0(\tilde{\varphi}_k))| < \frac{\tilde{\varepsilon}}{3}, \quad \forall N > N_{\tilde{\varepsilon}}.$$

Hence, for an arbitrary point  $\varphi$  in  $\Phi$ , there is a k such that

$$\begin{split} |\hat{f}_{N}(\varphi) - f_{0}(\varphi)| &\leq |\hat{f}_{N}(\varphi) - \hat{f}_{N}(\tilde{\varphi}_{k})| + |f_{0}(\tilde{\varphi}_{k}) - f_{0}(\varphi)| + |\hat{f}_{N}(\tilde{\varphi}_{k}) - f_{0}(\tilde{\varphi}_{k})| \\ &< \tilde{L}\frac{\tilde{\varepsilon}}{3\tilde{L}} + \tilde{L}\frac{\tilde{\varepsilon}}{3\tilde{L}} + \frac{\tilde{\varepsilon}}{3} = \tilde{\varepsilon}, \quad \forall N > N_{\tilde{\varepsilon}}, \end{split}$$

where we have used that both  $f_0$  and  $\hat{f}_N$  satisfy a Lipschitz condition with Lipschitz constant  $\tilde{L}$ . Since  $\tilde{\varepsilon}$  and  $\varphi$  were arbitrary it follows that  $\hat{f}_N$  converges uniformly to  $f_0$ .  $\Box$ 

From the fourth and fifth assumption in Theorem 11.1, it can be seen that the constant L in the QP problem must be an upper bound on the true Lipschitz constant for the nonlinear function  $r_0$  and that  $\tilde{L} = L + M_{\theta}$  must be an upper bound on the Lipschitz constant of the true predictor function  $f_0$  for the consistency result to hold. These conditions are quite intuitive since the estimated function must be allowed to vary at least as much as the true function. However, it is interesting to see that the choice  $\theta_N = \theta_0$  does not have to be a feasible point to (11.4) since the linear part of the system (11.1) can be modeled by the nonparametric function  $\hat{r}_N$ , provided L is large enough. However, although a too hard bound on  $\theta_N$  might not ruin the consistency, an unnecessarily large L will give a less smooth function estimate with a finite number of measurements. One of the main benefits of including also a linear parametric term in the model structure is that the smoothness of the estimate of the nonlinear part in that case will not depend on how large the linear part of the system is.

The fact that the linear part of the nonlinear system can be described by the nonparametric nonlinear part of the model explains why Theorem 11.1 does not discuss consistency for the individual linear and nonlinear estimators. In general, these estimators are not consistent since the separation of the system into a linear and a nonlinear part is not unique if a too large Lipschitz constant L is used. Some further properties of the proposed mixed parametric and nonparametric method will be discussed in the examples in the next section.

#### 11.3 Examples

The previously presented method for combined parametric and nonparametric estimation of NARX or NOE systems has been used in a couple of numerical examples.

#### 11.3.1 NARX Models

The first example concerns identification of a static nonlinearity. Although this example is very simple, it illustrates a number of general properties of the proposed method.

#### — Example 11.1 –

Consider the system

$$y(t) = 0.4u(t) + r_0(u(t)) + e(t), \qquad (11.20)$$

where both u(t) and e(t) are white noise processes and independent of each other. The input u(t) has uniform distribution on the interval [-10, 10] while the noise e(t) is normally distributed such that its mean is zero and its variance is 25. The nonlinearity in this system is

$$r_0(u(t)) = \frac{40}{5+|u(t)|} \left(\frac{u(t)}{1+|u(t)|} - \frac{u(t)-3}{1+|u(t)-3|} - \frac{u(t)+6}{1+|u(t)+6|} + \frac{3}{28}\right).$$

This function is Lipschitz continuous with  $L_0 = 7.4$  and bounded since  $|r'_0(x)| < 7.4$ and  $|r_0(x)| < 3.1$ , for all  $x \in \mathbb{R}$ .

A small data set consisting of 40 measurements of the input and output in (11.20) has been generated and is shown in Figure 11.1. Note that the shape of the nonlinear function is not obvious in this figure. The method (11.5) with L = 7.4 has been used with this data set and linear interpolation has been used to construct  $\hat{r}_N$ . The resulting predictor function  $\hat{f}_N(\varphi)$  is shown in Figure 11.2. From this figure, it seems that the function estimate has



**Figure 11.1:** The values of y(t) plotted against u(t) for the data set with 40 measurements used in Example 11.1.

managed to pick up some key features of the true function, despite the small number of measurements.

In this case, the L value used in the method is equal to  $L_0$ . In a more realistic example, the true Lipschitz constant would typically be unknown. An alternative would then be to divide the data set into estimation data and validation data and try different values of L. By evaluating the predictor (11.8) on the validation data for different choices of Lipschitz constant, it would be possible to find a good choice of L.

If this approach would be used for the identification of the system (11.20), the resulting Lipschitz constant L would probably be smaller than  $L_0$ . This results follows since the variations in  $\hat{r}_N(\varphi)$  in regions where  $|r'_0(\varphi)|$  is rather small can be reduced if a bias is accepted in regions where  $|r'_0(\varphi)|$  is close to  $L_0$ .

A larger data set consisting of 500 measurements of the input and output in (11.20) has also been generated and a couple of models have been estimated using an extended version of (11.5) where bounds  $\pm \rho_N(t) \leq 4$  have been added. One model was estimated using L = 15 and the resulting predictor function is shown in Figure 11.3a. The choices L = 7.4 and L = 4 gave the results shown in Figure 11.3b and 11.3c, respectively. From these figures, it seems that the function estimates contain no significant systematic errors and that a larger value of L gives more variations. Note that for L = 4, the true function estimate gives a rather good approximation of  $r_0$  anyway.

In the case with L = 15, the obtained estimate of the linear regression parameter  $\theta_0 = 0.4 \text{ was } \hat{\theta}_N = 0.31$  while L = 7.4 gave  $\hat{\theta}_N = 0.33$  and L = 4 gave  $\hat{\theta}_N = 0.39$ . Using the same data set but with a completely linear model, the least-squares method gave an estimate  $\hat{\theta}_{LS} = 0.23$ . Hence, it seems that including a bounded nonlinear Lipschitz continuous term in the model sometimes can improve the estimate of the linear part.

Although the estimated nonlinear predictor functions in the previous example are rela-



**Figure 11.2:** The predictor function estimated from 40 measurements (dashed) and the true predictor function (solid) from Example 11.1.

tively good approximations of the true nonlinear function, they are not very smooth. The fact that the function estimate  $\hat{r}_N$  will not be differentiable everywhere in its domain might be a problem in some applications. However, the nonparametric estimate presented here can still be useful as an initial estimate of the nonlinearities.

The method (11.5) combined with the interpolation (11.7) has also been used on a NARX system where the regression vector consists of two past output components and one input component. The results of this numerical experiment are described in Roll et al. (2005a) and indicates that the proposed method gives useful estimates of the predictor function also when  $\varphi(t)$  is a vector.

#### 11.3.2 NOE Models

The iterative method described in Section 11.1 for finding a local minimum to the NOE optimization problem (11.10) has been used on data from the nonlinear system in Example 8.1. The results of this experiment are described in the following example.

— Example 11.2 –

Consider again the nonlinear system

$$y(t) = y_l(t) + \alpha y_n(t),$$
  

$$y_l(t) = u(t),$$
  

$$y_n(t) = u(t)^3,$$

which was studied in Examples 8.1, 8.2 and 8.3. Just like in Example 8.1, assume that the input signal is

$$u(t) = L_m(q, c)e(t),$$

where

$$L_m(q,c) = (1 - cq^{-1})^2 = 1 - 2cq^{-1} + c^2q^{-2}, \quad 0 < c < 1,$$



**Figure 11.3:** The predictor function estimated from 500 measurements for three choices of *L* (dashed) and the true predictor function (solid) from Example 11.1.

e(t) is a white noise process with uniform distribution over the interval [-1, 1] and where c = 0.99 and  $\alpha = 0.01$ .

A realization of this input has been generated and a data set with 50 000 input and output measurements has been collected. The iterative method described after (11.10) in Section 11.1 for finding a local minimum to (11.10) has been used with this data set with Lipschitz constant L = 0.48 and additional bounds  $|\rho_N(t)| \le 0.64$ . These values correspond to the true bounds on the nonlinearity for inputs in the interval [-4, 4]. The orders of the estimated OE models have been  $n_b = 3$ ,  $n_f = 2$  and  $n_k = 0$ , i.e., equal to the orders of the OE-LTI-SOE of this system (cf. Example 8.1).

In order to minimize the computation time, only 2 000 measurements have been used to calculate each  $\hat{\rho}_N$ . However, different measurements have been used for each estimation of the nonlinearity, i.e., the measurements for  $t = 1, 2, \ldots, 2000$  were used to calculate the first  $\hat{\rho}_N$ , the measurements for  $t = 2001, 2002, \ldots, 4000$  were used to calculate the next one, etc. Furthermore, the largest and the smallest input components and the corresponding output components have been added to each of these small data sets, if they were not already included.

Some of the resulting nonparametric function estimates are shown in Figure 11.4. As can be seen from Figure 11.4a, the residuals from the first LTI model contain a lot of information about the unmodeled nonlinearity. However, the difference between the estimated LTI model and the linear part of the system implies that the residuals only can



**Figure 11.4:** The residuals from the LTI models (grey) and the corresponding nonparametric function estimates (black) from some of the iteration steps in Example 11.2. In this case, the output from the true system contains no measurement noise.

give a blurred picture of the nonlinearity. In Figures 11.4b to 11.4d, it can be seen that the residual plots get sharper after each iteration. After nine iterations, there is a close fit between the estimated and the true nonlinearity.

The frequency responses of some of the estimated LTI models are shown in Figure 11.5. After nine iterations, the estimated LTI model is essentially a constant. Hence, it seems that the nonlinear term in the model is able to describe a significant part of the true nonlinearity and thus to reduce its effect on the estimated LTI model.

The iterative method has also been tested on a data set with the same realization of the input but where a realization of Gaussian measurement noise with zero mean and variance 0.0025 has been added to the output. In this case, the same orders of the OE model and bounds on the  $\rho$  variables as for the noise-free case have been used. The residuals from the LTI model and the estimated nonlinearity from the first iteration are shown in Figure 11.6. The estimate of the nonlinearity seems to be able to describe the true nonlinearity rather well also in this case. The frequency responses of some of the estimated LTI models are



**Figure 11.5:** The estimated LTI models of the noise-free system in Example 11.2 from the first (solid), second (dashed) and ninth (dash-dotted) iteration.



**Figure 11.6:** The residuals from the initial LTI model (grey) and the corresponding nonparametric function estimate (black) for the system with measurement noise in Example 11.2.

shown in Figure 11.7. Again, it seems that the sensitivity of the LTI models to the small nonlinearity has been reduced significantly.



**Figure 11.7:** The estimated LTI models of the system with measurement noise in Example 11.2 from the first (solid), second (dashed) and ninth (dash-dotted) iteration.

#### 11.4 A General Perspective

As was mentioned previously in Section 11.1, it is easy to incorporate various kinds of prior knowledge into the identification problem. In fact, we can regard the presented approach as a special instance of the more general identification problem

$$\begin{array}{ll} \underset{\theta_N,\rho_N}{\text{minimize}} & \frac{1}{N} \sum_{t=1}^{N} (y(t) - \theta_N^T \varphi(t) - \rho_N(t))^2 \\ \text{subject to} & A \begin{pmatrix} \rho_N \\ \theta \end{pmatrix} \preccurlyeq b, \end{array}$$
(11.21)

where  $\preccurlyeq$  denotes component-wise inequality. This is still a convex QP problem. An interesting special case of (11.21) is

$$\begin{array}{ll} \underset{\theta_{N},\rho_{N}}{\text{minimize}} & \frac{1}{N} \sum_{t=1}^{N} (y(t) - \theta_{N}^{T} \varphi(t) - \rho_{N}(t))^{2} \\ \text{subject to} & |\rho_{N}(t)| \leq M, \\ & \forall t \in \{1, 2, \dots, N\}. \end{array}$$
(11.22)

It turns out that minimizing (11.22) gives exactly the same linear part as using an  $\varepsilon$ -insensitive norm for identification of ARX models, i.e.,

$$\underset{\theta_N}{\text{minimize}} \quad \frac{1}{N} \sum_{t=1}^{N} \left| y(t) - \theta_N^T \varphi(t) \right|_{\varepsilon}^k, \tag{11.23}$$

with

$$|x|_{\varepsilon} = \begin{cases} 0, & |x| \le \varepsilon, \\ |x| - \varepsilon, & |x| > \varepsilon \end{cases}$$

and with k = 2 and  $\varepsilon = M$ . This norm (or the corresponding norm with k = 1) is often used in support vector machines (Vapnik, 1998), and similar approaches are also used in

robust adaptive control (Peterson and Narendra, 1982). To see the equivalence between (11.22) and (11.23), define

$$\bar{r}(t,\theta) = \begin{cases} M, & y(t) - \theta^T \varphi(t) > M, \\ y(t) - \theta^T \varphi(t), & -M \le y(t) - \theta^T \varphi(t) \le M, \\ -M, & y(t) - \theta^T \varphi(t) < -M. \end{cases}$$

Then we can write (11.23) as

$$\underset{\theta_N}{\text{minimize}} \ \frac{1}{N} \sum_{t=1}^{N} \left| y(t) - (\theta_N^T \varphi(t) + \bar{r}(t, \theta_N)) \right|^k$$

On the other hand it is easy to see that, for a given  $\theta_N$ , the minimum of (11.22) is obtained precisely when  $\rho_N(t) = \bar{r}(t, \theta_N)$ . Since  $\bar{r}(t, \theta_N)$  automatically has a magnitude not greater than M, the desired equivalence follows.

The advantage with using the formulation (11.22) instead of (11.23) is that the explicit representation of  $\rho_N$  again makes it possible to combine different types of requirements on the nonlinearity.

#### 11.5 Discussion

In this chapter, a method has been proposed for identification of NARX systems using a model structure with a parametric linear part and a nonparametric nonlinear part. The model estimate is computed by solving a convex optimization problem where the only assumption on the nonlinearities in the system is that they are Lipschitz continuous. Theorem 11.1 gives a direct proof of the consistency of this method and its usefulness has also been illustrated in examples.

An iterative method for identification of a class of NOE systems has also been presented. The model structure that can be used with this method contains a linear OE part and a nonparametric NFIR part. This NOE identification method can sometimes reduce the sensitivity of OE models to small nonlinearities (see Example 11.2).

A nice property of both presented methods is that it is easy to include prior system knowledge in them if this knowledge can be written as linear constraints. Furthermore, since a nonparametric model of the nonlinearities is used, the methods might be particularly useful for computing initial estimates of unknown nonlinear systems where the shapes of the nonlinearities are completely unknown.

# 12 Conclusions

In this thesis, system identification using LTI models of nonlinear systems with random inputs has been studied. The main focus has been on analysis of LTI models that are optimal approximations in the sense that they minimize a mean-square error criterion. This is a very rich research field with many theoretically and practically interesting problems to consider.

Here, properties of the optimal LTI approximations for different types of input signals have been studied. For example, several useful properties of the LTI models have been shown for classes of minimum phase filtered white noise inputs, Gaussian inputs and separable inputs. Furthermore, LTI approximations of nonlinear systems have been discussed from a robust control point of view and a method for mixed parametric and nonparametric identification has been proposed.

However, it must be admitted that this thesis mainly concerns theoretical approximation aspects and that no particular application has been considered. Hence, it would be interesting to investigate LTI models for some real-life applications in order to see which are the dominating effects of unmodeled nonlinearities. Hopefully, such a study would show that at least some of the theoretical results in this thesis are nice illustrations of the following classic saying, which has been attributed to many famous scientists.

There is nothing more practical than a good theory.

### **Bibliography**

- T. S. Atalik and S. Utku. Stochastic linearization of multi-degree-of-freedom non-linear systems. Earthquake Engineering and Structural Dynamics, 4:411–420, 1976.
- D. P. Atherton. *Nonlinear Control Engineering*. Van Nostrand Reinhold, New York, student edition, 1982.
- E.-W. Bai. Frequency domain identification of Hammerstein models. IEEE Transactions on Automatic Control, 48(4):530–542, 2003a.
- E.-W. Bai. Frequency domain identification of Wiener models. Automatica, 39(9):1521– 1530, 2003b.
- E.-W. Bai. Decoupling the linear and nonlinear parts in Hammerstein model identification. Automatica, 40(4):671–676, 2004.
- E.-W. Bai and D. Li. Convergence of the iterative Hammerstein system identification algorithm. *IEEE Transactions on Automatic Control*, 49(11):1929–1940, 2004.
- A. V. Balakrishnan. On a characterization of processes for which optimal mean-square systems are of specified form. *IRE Transactions on Information Theory*, 6(4):490–500, 1960.
- J. F. Barrett and D. G. Lampard. An expansion for some second-order probability distributions and its application to noise problems. *IRE Transactions on Information Theory*, 1(1):10–15, 1955.
- J. S. Bendat. Nonlinear Systems Techniques and Applications. John Wiley & Sons, 1998.
- D. Bertsimas, D. Gamarnik, and J. N. Tsitsiklis. Estimation of time-varying parameters in statistical models: An optimization approach. *Machine Learning*, 35(3):225–245, 1999.

- S. A. Billings and S. Y. Fakhouri. Theory of separable processes with applications to the identification of nonlinear systems. *Proceedings of the IEE*, 125(9):1051–1058, 1978.
- S. A. Billings and S. Y. Fakhouri. Identification of systems containing linear dynamic and static nonlinear elements. *Automatica*, 18(1):15–26, 1982.
- R. C. Booton, Jr. Nonlinear control systems with random inputs. *IRE Transactions on Circuit Theory*, 1(1):9–18, 1954.
- S. Boyd and L. Vandenberghe. Convex Optimization. Cambridge University Press, Cambridge, 2004.
- P. J. Brockwell and R. A. Davis. *Time Series: Theory and Methods*. Springer-Verlag, New York, 1987.
- J. L. Brown, Jr. On a cross-correlation property for stationary random processes. IRE Transactions on Information Theory, 3(1):28–31, 1957.
- J. W. Brown and R. V. Churchill. *Complex Variables and Applications*. McGraw Hill, Singapore, 1996.
- A. E. Bryson, Jr. and Y.-C. Ho. Applied Optimal Control. Taylor & Francis, 1975.
- J. J. Bussgang. Crosscorrelation functions of amplitude-distorted Gaussian signals. Technical Report 216, MIT Research Laboratory of Electronics, Cambridge, Massachusetts, 1952.
- X. Chen, J. Racine, and N. R. Swanson. Semiparametric ARX neural-network models with an application to forecasting inflation. *IEEE Transactions on Neural Networks*, 12 (4):674–683, 2001.
- K. L. Chung. A Course in Probability Theory. Academic Press, New York, second edition, 1974.
- P. Crama and J. Schoukens. Initial estimates of Wiener and Hammerstein systems using multisine excitation. *IEEE Transactions on Instrumentation and Measurement*, 50(6): 1791–1795, 2001.
- P. Crama and J. Schoukens. Hammerstein-Wiener system estimator initialization. Automatica, 40(9):1543–1550, 2004.
- P. Crama and J. Schoukens. Computing an initial estimate of a Wiener-Hammerstein system with a random phase multisine excitation. *IEEE Transactions on Instrumentation and Measurement*, 54(1):117–122, 2005.
- P. Crama, J. Schoukens, and R. Pintelon. Generation of enhanced initial estimates for Hammerstein systems. Automatica, 40(7):1269–1273, 2004.
- T. P. Dobrowiecki and J. Schoukens. Practical choices in the FRF measurement in presence of nonlinear distortions. *IEEE Transactions on Instrumentation and Measurement*, 50(1):2–7, 2001.

- M. Enqvist. Some results on linear models of nonlinear systems. Licentiate thesis no. 1046. Department of Electrical Engineering, Linköpings universitet, Linköping, Sweden, 2003.
- M. Enqvist. Benefits of the input minimum phase property for linearization of nonlinear systems. In Proceedings of the International Symposium on Nonlinear Theory and its Applications, pages 618–621, Bruges, Belgium, October 2005.
- M. Enqvist. Identification of Hammerstein systems using separable random multisines. Submitted to the 14th IFAC Symposium on System Identification, Newcastle, Australia, March 2006.
- M. Enqvist and L. Ljung. Estimating nonlinear systems in a neighborhood of LTIapproximants. In Proceedings of the 41st IEEE Conference on Decision and Control, pages 1005–1010, Las Vegas, Nevada, December 2002.
- M. Enqvist and L. Ljung. Linear models of nonlinear FIR systems with Gaussian inputs. In Proceedings of the 13th IFAC Symposium on System Identification, pages 1910– 1915, Rotterdam, The Netherlands, August 2003.
- M. Enqvist and L. Ljung. LTI approximations of slightly nonlinear systems: Some intriguing examples. In Proceedings of the 6th IFAC Symposium on Nonlinear Control Systems, pages 639–644, Stuttgart, Germany, September 2004.
- M. Enqvist and L. Ljung. Linear approximations of nonlinear FIR systems for separable input processes. Automatica, 41(3):459–473, 2005.
- C. Evans and D. Rees. Nonlinear distortions and multisine signals Part I: Measuring the best linear approximation. *IEEE Transactions on Instrumentation and Measurement*, 49(3):602–609, 2000a.
- C. Evans and D. Rees. Nonlinear distortions and multisine signals Part II: Minimizing the distortion. *IEEE Transactions on Instrumentation and Measurement*, 49(3):610– 616, 2000b.
- J. Fan and I. Gijbels. Local Polynomial Modelling and Its Applications. Chapman & Hall, 1996.
- U. Forssell and L. Ljung. A projection method for closed-loop identification. IEEE Transactions on Automatic Control, 45(11):2101–2106, 2000.
- W. A. Gardner. Introduction to Random Processes. Macmillan Publishing Company, New York, 1986.
- A. Garulli, A. Tesi, and A. Vicino, editors. *Robustness in Identification and Control.* Lecture Notes in Control and Information Sciences. Springer-Verlag, 1999.
- G. B. Giannakis and E. Serpedin. A bibliography on nonlinear system identification. Signal Processing, 81(3):533–580, 2001.

- S. T. Glad, A. Helmersson, and L. Ljung. Robustness guarantees for linear control designs with an estimated nonlinear model error model. *International Journal of Robust and Nonlinear Control*, 14(11):959–970, 2004.
- S. T. Glad, A. Helmersson, M. Enqvist, and L. Ljung. Controllers for amplitude limited model error models. In *Proceedings of the 16th IFAC World Congress*, Prague, Czech Republic, July 2005.
- T. Glad and L. Ljung. *Control Theory Multivariable and Nonlinear Methods*. Taylor & Francis, London and New York, 2000.
- C. D. Gorman and J. Zaborszky. Functional calculus in the theory of nonlinear systems with stochastic signals. *IEEE Transactions on Information Theory*, 14(4):528–531, 1968.
- A. Gut. An Intermediate Course in Probability. Springer-Verlag, New York, 1995.
- R. Haber and H. Unbehauen. Structure identification of nonlinear dynamic systems A survey on input/output approaches. *Automatica*, 26(4):651–677, 1990.
- N. E. Heckman. Minimax estimates in a semiparametric model. *Journal of the American Statistical Association*, 83(404):1090–1096, 1988.
- D. Henrion and S. Tarbouriech. LMI relaxations for robust stability of linear systems with saturating controls. *Automatica*, 35(9):1599–1604, 1999.
- H. Hjalmarsson. From experiment design to closed-loop control. *Automatica*, 41(3): 393–438, 2005.
- I. M. Horowitz. Quantitative Feedback Design Theory. QFT Publications, Boulder, Colorado, 1993.
- I. W. Hunter and M. J. Korenberg. The identification of nonlinear biological systems: Wiener and Hammerstein cascade models. *Biological Cybernetics*, 55:135–144, 1986.
- A. H. Jazwinski. Stochastic Processes and Filtering Theory. Academic Press, New York, 1970.
- T. Kailath. Linear Systems. Prentice Hall, Englewood Cliffs, New Jersey, 1980.
- T. Kailath, A. H. Sayed, and B. Hassibi. *Linear Estimation*. Prentice Hall, Upper Saddle River, New Jersey, 2000.
- H. K. Khalil. *Nonlinear Systems*. Prentice Hall, Upper Saddle River, New Jersey, third edition, 2002.
- J.-H. Kim and Z. Bien. Robust stability of uncertain linear systems with saturating actuators. *IEEE Transactions on Automatic Control*, 39(1):202–207, 1994.
- M. J. Korenberg. Identifying noisy cascades of linear and static nonlinear systems. In Proceedings of the 7th IFAC Symposium on Identification and System Parameter Estimation, pages 421–426, York, UK, 1985.

- L. Ljung. Convergence analysis of parametric identification methods. IEEE Transactions on Automatic Control, 23(5):770–783, 1978.
- L. Ljung. System Identification: Theory for the User. Prentice Hall, Upper Saddle River, New Jersey, second edition, 1999.
- L. Ljung. Estimating linear time-invariant models of nonlinear time-varying systems. *European Journal of Control*, 7(2-3):203–219, 2001.
- L. Ljung and T. Glad. Modeling of Dynamic Systems. Prentice Hall, Englewood Cliffs, New Jersey, 1994.
- L. D. Lutes and S. Sarkani. *Stochastic Analysis of Structural and Mechanical Vibrations*. Prentice Hall, Upper Saddle River, New Jersey, 1997.
- D. K. McGraw and J. F. Wagner. Elliptically symmetric distributions. *IEEE Transactions* on Information Theory, 14(1):110–120, 1968.
- P. M. Mäkilä. Optimal approximation and model quality estimation for nonlinear systems. In Proceedings of the 13th IFAC Symposium on System Identification, pages 1904– 1909, Rotterdam, The Netherlands, August 2003a.
- P. M. Mäkilä. Squared and absolute errors in optimal approximation of nonlinear systems. *Automatica*, 39(11):1865–1876, 2003b.
- P. M. Mäkilä. On optimal LTI approximation of nonlinear systems. IEEE Transactions on Automatic Control, 49(7):1178–1182, 2004a.
- P. M. Mäkilä. Stabilization and Modelling of Input-Output Systems. PhD thesis, Tampere University of Technology, Tampere, Finland, 2004b.
- P. M. Mäkilä. LTI modelling of NFIR systems: near-linearity and control, LS estimation and linearization. Automatica, 41(1):29–41, 2005.
- P. M. Mäkilä and J. R. Partington. On linear models for nonlinear systems. Automatica, 39(1):1–13, 2003.
- P. M. Mäkilä and J. R. Partington. Least-squares LTI approximation of nonlinear systems and quasistationarity analysis. *Automatica*, 40(7):1157–1169, 2004.
- K. S. Narendra and P. G. Gallman. An iterative method for the identification of nonlinear systems using a Hammerstein model. *IEEE Transactions on Automatic Control*, 11(3): 546–550, 1966.
- A. H. Nuttall. Theory and application of the separable class of random processes. Technical Report 343, MIT Research Laboratory of Electronics, Cambridge, Massachusetts, 1958a.
- A. H. Nuttall. Theory and Application of the Separable Class of Random Processes. PhD thesis, MIT, Cambridge, Massachusetts, 1958b.

- A. Papoulis. Probability, Random Variables and Stochastic Processes. McGraw Hill, second edition, 1984.
- J. R. Partington and P. M. Mäkilä. On system gains for linear and nonlinear systems. Systems & Control Letters, 46(2):129–136, 2002.
- R. K. Pearson. *Discrete-Time Dynamic Models*. Oxford University Press, New York, 1999.
- B. B. Peterson and K. S. Narendra. Bounded error adaptive control. *IEEE Transactions* on *Automatic Control*, 27(6):1161–1168, 1982.
- R. Pintelon and J. Schoukens. *System Identification: A Frequency Domain Approach*. IEEE Press, New York, 2001.
- R. Pintelon and J. Schoukens. Measurement and modelling of linear systems in the presence of non-linear distortions. *Mechanical Systems and Signal Processing*, 16(5):785– 801, 2002.
- R. Pintelon, J. Schoukens, W. Van Moer, and Y. Rolain. Identification of linear systems in the presence of nonlinear distortions. *IEEE Transactions on Instrumentation and Measurement*, 50(4):855–863, 2001.
- R. Pintelon, J. Schoukens, and Y. Rolain. Uncertainty of transfer function modelling using prior estimated noise models. *Automatica*, 39(10):1721–1733, 2003.
- J. Roll, M. Enqvist, and L. Ljung. Consistent nonparametric estimation of NARX systems using convex optimization. In *Proceedings of the 44th IEEE Conference on Decision and Control and the European Control Conference*, Seville, Spain, December 2005a. (To appear).
- J. Roll, A. Nazin, and L. Ljung. Nonlinear system identification via direct weight optimization. Automatica, 41(3):475–490, 2005b.
- W. J. Rugh. *Linear System Theory*. Prentice Hall, Upper Saddle River, New Jersey, second edition, 1996.
- L. Råde and B. Westergren. *Mathematics Handbook for Science and Engineering*. Studentlitteratur, Lund, Sweden, third edition, 1995.
- S. Sastry. Nonlinear Systems Analysis, Stability and Control. Springer-Verlag, New York, 1999.
- G. Scarano, D. Caggiati, and G. Jacovitti. Cumulant series expansion of hybrid nonlinear moments of *n* variates. *IEEE Transactions on Signal Processing*, 41(1):486–489, 1993.
- M. Schetzen. The Volterra & Wiener Theories of Nonlinear Systems. John Wiley & Sons, 1980.
- J. Schoukens, T. Dobrowiecki, and R. Pintelon. Parametric and nonparametric identification of linear systems in the presence of nonlinear distortions - A frequency domain approach. *IEEE Transactions on Automatic Control*, 43(2):176–190, 1998.
- J. Schoukens, R. Pintelon, Y. Rolain, and T. Dobrowiecki. Frequency response function measurements in the presence of nonlinear distortions. *Automatica*, 37(6):939–946, 2001.
- J. Schoukens, R. Pintelon, and T. Dobrowiecki. Linear modeling in the presence of nonlinear distortions. *IEEE Transactions on Instrumentation and Measurement*, 51(4): 786–792, 2002.
- J. Schoukens, J. G. Nemeth, P. Crama, Y. Rolain, and R. Pintelon. Fast approximate identification of nonlinear systems. *Automatica*, 39(7):1267–1274, 2003.
- J. Schoukens, T. Dobrowiecki, and R. Pintelon. Estimation of the risk for an unstable behaviour of feedback systems in the presence of nonlinear distortions. *Automatica*, 40(7):1275–1279, 2004a.
- J. Schoukens, J. G. Németh, G. Vandersteen, R. Pintelon, and P. Crama. Linearization of nonlinear dynamic systems. *IEEE Transactions on Instrumentation and Measurement*, 53(4):1245–1248, 2004b.
- J. Schoukens, R. Pintelon, T. Dobrowiecki, and Y. Rolain. Identification of linear systems with nonlinear distortions. *Automatica*, 41(3):491–504, 2005a.
- J. Schoukens, J. Swevers, J. Paduart, D. Vaes, K. Smolders, and R. Pintelon. Initial estimates for block structured nonlinear systems with feedback. In *Proceedings of the International Symposium on Nonlinear Theory and its Applications*, Bruges, Belgium, October 2005b.
- J. Sjöberg, Q. Zhang, L. Ljung, A. Benveniste, B. Deylon, P.-Y. Glorennec, H. Hjalmarsson, and A. Juditsky. Nonlinear black-box modeling in system identification: a unified overview. Automatica, 31(12):1691–1724, 1995.
- P. Stoica. On the convergence of an iterative algorithm used for Hammerstein system identification. *IEEE Transactions on Automatic Control*, 26(4):967–969, 1981.
- P. M. J. Van den Hof and R. J. P. Schrama. An indirect method for transfer function estimation from closed-loop data. *Automatica*, 29(6):1523–1527, 1993.
- A. J. van der Schaft.  $L_2$ -gain analysis of nonlinear systems and nonlinear state feedback  $H_{\infty}$  control. *IEEE Transactions on Automatic Control*, 37(6):770–784, 1992.
- G. Vandersteen and J. Schoukens. Measurement and identification of nonlinear systems consisting of linear dynamic blocks and one static nonlinearity. *IEEE Transactions on Automatic Control*, 44(6):1266–1271, 1999.
- G. Vandersteen, Y. Rolain, and J. Schoukens. Non-parametric estimation of the frequencyresponse functions of the linear blocks of a Wiener-Hammerstein model. *Automatica*, 33(7):1351–1355, 1997.
- G. Vandersteen, Y. Rolain, and J. Schoukens. An identification technique for data acquisition characterization in the presence of nonlinear distortions and time base distortions. *IEEE Transactions on Instrumentation and Measurement*, 50(5):1355–1363, 2001.

- K. Vanhoenacker and J. Schoukens. Detection of nonlinear distortions with multisine excitations in the case of nonideal behavior of the input signal. *IEEE Transactions on Instrumentation and Measurement*, 52(3):748–753, 2003.
- K. Vanhoenacker, T. Dobrowiecki, and J. Schoukens. Design of multisine excitations to characterize the nonlinear distortions during FRF-measurements. *IEEE Transactions* on Instrumentation and Measurement, 50(5):1097–1102, 2001.
- V. N. Vapnik. Statistical Learning Theory. John Wiley & Sons, New York, 1998.
- M. Vidyasagar. Nonlinear Systems Analysis. Prentice Hall, Englewood Cliffs, New Jersey, second edition, 1993.
- N. Wiener. Extrapolation, Interpolation and Smoothing of Stationary Time Series. Technology Press and Wiley, New York, 1949.
- K. Zhou, J. C. Doyle, and K. Glover. Robust and Optimal Control. Prentice Hall, Upper Saddle River, New Jersey, 1996.

A

## **Calculations for Example 4.3**

Since  $E(e(t)^3) = 0$  and  $E(e(t)^4) = 3$ , we get

$$\begin{aligned} R_u(0) &= \mathcal{E}(u(t)^2) = \mathcal{E}((e(t) + e(t-1)^2 - 1)^2) \\ &= \mathcal{E}(e(t)^2) + \mathcal{E}(e(t-1)^4) + 1 - 2\mathcal{E}(e(t-1)^2) = 1 + 3 + 1 - 2 = 3, \\ R_u(\pm 1) &= \mathcal{E}(u(t)u(t-1)) = 0, \\ R_u(\tau) &= 0, \quad \forall \tau \in \mathbb{Z} \setminus \{-1, 0, 1\}. \end{aligned}$$

Furthermore, since  $E(e(t)^5) = E(e(t)^7) = 0$ ,  $E(e(t)^6) = 15$  and  $E(e(t)^8) = 105$ , we get

$$\begin{aligned} R_{yu}(0) &= \mathrm{E}(y(t)u(t)) \\ &= \mathrm{E}(\left(e(t)^2 + e(t-1)^4 + 2e(t)e(t-1)^2 - 2e(t) - 2e(t-1)^2 - 2\right) \\ &\cdot \left(e(t) + e(t-1)^2 - 1\right)) \\ &= 2\mathrm{E}(e(t)^2e(t-1)^2) - 2\mathrm{E}(e(t)^2) + \mathrm{E}(e(t-1)^6) - \mathrm{E}(e(t-1)^4) \\ &- 2\mathrm{E}(e(t-1)^4) + 2\mathrm{E}(e(t-1)^2) \\ &= 2 - 2 + 15 - 3 - 2 \cdot 3 + 2 = 8, \\ R_{yu}(1) &= \mathrm{E}(y(t)u(t-1)) \\ &= \mathrm{E}(\left(e(t)^2 + e(t-1)^4 + 2e(t)e(t-1)^2 - 2e(t) - 2e(t-1)^2 - 2\right) \\ &\cdot \left(e(t-1) + e(t-2)^2 - 1\right)) = 0, \\ R_{yu}(-1) &= \mathrm{E}(y(t)u(t+1)) \\ &= \mathrm{E}(\left(e(t)^2 + e(t-1)^4 + 2e(t)e(t-1)^2 - 2e(t) - 2e(t-1)^2 - 2\right) \\ &\cdot \left(e(t+1) + e(t)^2 - 1\right)) = \mathrm{E}(e(t)^4) - \mathrm{E}(e(t)^2) = 3 - 1 = 2, \\ R_{yu}(\tau) &= 0, \quad \forall \tau \in \mathbb{Z} \setminus \{-1, 0, 1\} \end{aligned}$$

and

$$\begin{split} R_y(0) &= \mathrm{E}(y(t)^2) \\ &= \mathrm{E}((e(t)^2 + e(t-1)^4 + 2e(t)e(t-1)^2 - 2e(t) - 2e(t-1)^2 - 2)) \\ &\quad \cdot (e(t)^2 + e(t-1)^4 + 2e(t)e(t-1)^2 - 2e(t) - 2e(t-1)^2 - 2)) \\ &= \mathrm{E}(e(t)^4) + 2\mathrm{E}(e(t)^2e(t-1)^4) - 4\mathrm{E}(e(t)^2e(t-1)^2) - 4\mathrm{E}(e(t)^2) \\ &\quad + \mathrm{E}(e(t-1)^8) - 4\mathrm{E}(e(t-1)^6) - 4\mathrm{E}(e(t-1)^4) + 4\mathrm{E}(e(t)^2e(t-1)^4) \\ &\quad - 8\mathrm{E}(e(t)^2e(t-1)^2) + 4\mathrm{E}(e(t)^2) + 4\mathrm{E}(e(t-1)^4) + 8\mathrm{E}(e(t-1)^2) + 4 \\ &= 3 + 2 \cdot 3 - 4 - 4 + 105 - 4 \cdot 15 - 4 \cdot 3 + 4 \cdot 3 - 8 + 4 + 4 \cdot 3 + 8 + 4 \\ &= 66, \\ R_y(\pm 1) &= \mathrm{E}(y(t)y(t-1)) \\ &= \mathrm{E}((e(t)^2 + e(t-1)^4 + 2e(t)e(t-1)^2 - 2e(t) - 2e(t-1)^2 - 2) \\ &\quad \cdot (e(t-1)^2 + e(t-2)^4 + 2e(t-1)e(t-2)^2 - 2e(t-1) \\ &\quad - 2e(t-2)^2 - 2)) \\ &= \mathrm{E}(e(t)^2e(t-1)^2) + \mathrm{E}(e(t-1)^6) - 2\mathrm{E}(e(t-1)^4) \\ &\quad - 2\mathrm{E}(e(t-1)^2) + \mathrm{E}(e(t)^2e(t-2)^4) + \mathrm{E}(e(t-1)^4e(t-2)^4) \\ &\quad - 2\mathrm{E}(e(t-1)^2e(t-2)^4) - 2\mathrm{E}(e(t-1)^2e(t-2)^2) + 4\mathrm{E}(e(t-2)^2) \\ &\quad - 2\mathrm{E}(e(t-1)^4e(t-2)^2) + 4\mathrm{E}(e(t-1)^2e(t-2)^2) + 4\mathrm{E}(e(t-2)^2) \\ &\quad - 2\mathrm{E}(e(t)^2) - 2\mathrm{E}(e(t-1)^4) + 4\mathrm{E}(e(t-1)^2) + 4 \\ &= 1 + 15 - 2 \cdot 3 - 2 + 3 + 9 - 2 \cdot 3 - 2 \cdot 3 - 2 - 2 \cdot 3 + 4 + 4 - 2 \\ &\quad - 2 \cdot 3 + 4 + 4 = 8, \\ R_y(\tau) &= 0, \quad \forall \tau \in \mathbb{Z} \setminus \{-1, 0, 1\}. \end{split}$$

The z-spectra are thus

$$\Phi_u(z) = 3, 
\Phi_{yu}(z) = 2z + 8, 
\Phi_y(z) = 8z + 66 + 8z^{-1}.$$

Inserted in (4.2b) this gives

$$\Phi_{\zeta}(z) = \begin{pmatrix} 3 & 2+8z \\ 2+8z^{-1} & 8z+66+8z^{-1} \end{pmatrix}.$$

In order to compute the canonical spectral factorization  $\Phi_{\zeta}(z) = T(z)Q_{\zeta}T^{T}(z^{-1})$  we first pre- and postmultiply  $\Phi_{\zeta}(z)$  with matrices  $T_{1}(z)$  and  $T_{1}^{T}(z^{-1})$ , respectively. If  $T_{1}(z)$  is chosen as

$$T_1(z) = \begin{pmatrix} 1 & 0\\ -\frac{2}{3} - \frac{8}{3}z^{-1} & 1 \end{pmatrix},$$

the result is the following diagonal matrix

$$D(z) = T_1(z)\Phi_{\zeta}(z)T_1^T(z^{-1}) = \begin{pmatrix} 3 & 0\\ 0 & \frac{2}{3}\left(4z^{-1} + 65 + 4z\right) \end{pmatrix}.$$

The matrix element  $D_{22}(z) = \frac{2}{3} \cdot \left(4z^{-1} + 65 + 4z\right)$  can be factorized as

$$D_{22}(z) = \frac{8}{3} \left( z - z_0 \right) \left( 1 - \frac{1}{z_0 z} \right) = -\frac{8}{3z_0} \left( z - z_0 \right) \left( z^{-1} - z_0 \right),$$

where  $z_0 = (-65 + \sqrt{4161})/8$ . Hence, the matrix D(z) can be factorized as  $D(z) = T_2(z)T_2^T(z^{-1})$ , where

$$T_2(z) = \begin{pmatrix} \sqrt{3} & 0 \\ 0 & \kappa (z - z_0) \end{pmatrix},$$

and where  $\kappa = \sqrt{-\frac{8}{3z_0}}$ . The two matrices  $T_1(z)$  and  $T_2(z)$  defines a spectral factorization,  $\Phi_{\zeta}(z) = T_p(z)T_p^T(z^{-1})$  with

$$T_p(z) = T_1^{-1}(z)T_2(z) = \begin{pmatrix} \sqrt{3} & 0\\ \frac{2+8z^{-1}}{\sqrt{3}} & \kappa(z-z_0) \end{pmatrix}.$$

However, this is not the *canonical* spectral factorization as  $T_p(\infty) \neq I$ . Let

$$T_3(z) = \begin{pmatrix} \frac{1}{\sqrt{3}} & 0\\ -\frac{2}{3\kappa z} & \frac{1}{\kappa z} \end{pmatrix}$$

and let

$$T(z) = T_p(z)T_3(z) = \begin{pmatrix} 1 & 0\\ \left(\frac{8+2z_0}{3}\right)z^{-1} & 1-z_0z^{-1} \end{pmatrix}$$
$$= \begin{pmatrix} 1 & 0\\ \frac{\sqrt{4161-33}}{12}z^{-1} & 1+\frac{65-\sqrt{4161}}{8}z^{-1} \end{pmatrix},$$
$$Q_{\zeta} = T_3^{-1}(z)T_3^{-T}(z^{-1}) = \begin{pmatrix} \sqrt{3} & 0\\ \frac{2}{\sqrt{3}} & \kappa z \end{pmatrix} \begin{pmatrix} \sqrt{3} & \frac{2}{\sqrt{3}}\\ 0 & \kappa z^{-1} \end{pmatrix}$$
$$= \begin{pmatrix} 3 & 2\\ 2 & \frac{4}{3} + \kappa^2 \end{pmatrix} = \begin{pmatrix} 3 & 2\\ 2 & 23 + \frac{\sqrt{4161}}{3} \end{pmatrix}.$$

This gives  $\Phi_{\zeta}(z) = T(z)Q_{\zeta}T^{T}(z^{-1})$  and both T(z) and  $T^{-1}(z)$  are analytical on and outside the unit circle,  $T(\infty) = I$  and  $Q_{\zeta} \succ 0$ . Hence we have found the canonical spectral factorization of  $\Phi_{\zeta}(z)$ .

# MATLAB Code

### B.1 Example 5.2

MATLAB commands for Example 5.2:

```
N = 10000;
e=2*rand(N,1)-1;
of _____
% Minimum phase case
€ _____
ump=filter([1 .5],1,e);
ymp=ump.^3;
zmp=iddata(ymp,ump,1);
GOmpspa=spa(zmp, 30);
GOmp=tf([.85 .575],[1 .5],1)
figure(1)
clf
bode(G0mpspa,G0mp)
% _____
% Nonminimum phase case
§ _____
unmp=filter([.5 1],1,e);
```

```
ynmp=unmp.^3;
znmp=iddata(ynmp,unmp,1);
G0nmpspa=spa(znmp,30);
G0nmp=tf([.925 .425],[1 .5],1)
figure(2)
clf
```

### B.2 Example 7.2

bode (G0nmpspa, G0nmp)

MATLAB commands for Example 7.2:

```
N=100001;
e=randn(N,1);
u=filter([1 -.8 .1],[1 -.2],e);
v=u(1:end-1).^2.*atan(u(2:end));
w=randn(N-1,1);
y=filter(1,[1 .6 .1],v)+w;
z=iddata(y,u(2:end),1);
G=oe(z,[2 2 0],'lim',0,'cov','none')
```

### B.3 Example 7.3

MATLAB commands for Example 7.3:

```
N=100001;
e=randn(N,1);
u=filter([1 -.8 .1],[1 -.2],e);
n=filter(1,[1 .6 .1],u);
w=randn(N-1,1);
y=n(1:end-1).^2.*atan(n(2:end))+w;
z=iddata(y,u(2:end),1);
G=oe(z,[2 2 0],'lim',0,'cov','none')
```

#### PhD Dissertations Division of Automatic Control Linköpings universitet

**M. Millnert:** Identification and control of systems subject to abrupt changes. Thesis No. 82, 1982. ISBN 91-7372-542-0.

**A. J. M. van Overbeek:** On-line structure selection for the identification of multivariable systems. Thesis No. 86, 1982. ISBN 91-7372-586-2.

B. Bengtsson: On some control problems for queues. Thesis No. 87, 1982. ISBN 91-7372-593-5.

**S. Ljung:** Fast algorithms for integral equations and least squares identification problems. Thesis No. 93, 1983. ISBN 91-7372-641-9.

**H. Jonson:** A Newton method for solving non-linear optimal control problems with general constraints. Thesis No. 104, 1983. ISBN 91-7372-718-0.

**E. Trulsson:** Adaptive control based on explicit criterion minimization. Thesis No. 106, 1983. ISBN 91-7372-728-8.

**K. Nordström:** Uncertainty, robustness and sensitivity reduction in the design of single input control systems. Thesis No. 162, 1987. ISBN 91-7870-170-8.

**B. Wahlberg:** On the identification and approximation of linear systems. Thesis No. 163, 1987. ISBN 91-7870-175-9.

**S. Gunnarsson:** Frequency domain aspects of modeling and control in adaptive systems. Thesis No. 194, 1988. ISBN 91-7870-380-8.

**A. Isaksson:** On system identification in one and two dimensions with signal processing applications. Thesis No. 196, 1988. ISBN 91-7870-383-2.

**M. Viberg:** Subspace fitting concepts in sensor array processing. Thesis No. 217, 1989. ISBN 91-7870-529-0.

**K. Forsman:** Constructive commutative algebra in nonlinear control theory. Thesis No. 261, 1991. ISBN 91-7870-827-3.

**F. Gustafsson:** Estimation of discrete parameters in linear systems. Thesis No. 271, 1992. ISBN 91-7870-876-1.

**P. Nagy:** Tools for knowledge-based signal processing with applications to system identification. Thesis No. 280, 1992. ISBN 91-7870-962-8.

**T. Svensson:** Mathematical tools and software for analysis and design of nonlinear control systems. Thesis No. 285, 1992. ISBN 91-7870-989-X.

**S. Andersson:** On dimension reduction in sensor array signal processing. Thesis No. 290, 1992. ISBN 91-7871-015-4.

**H. Hjalmarsson:** Aspects on incomplete modeling in system identification. Thesis No. 298, 1993. ISBN 91-7871-070-7.

**I. Klein:** Automatic synthesis of sequential control schemes. Thesis No. 305, 1993. ISBN 91-7871-090-1.

**J.-E. Strömberg:** A mode switching modelling philosophy. Thesis No. 353, 1994. ISBN 91-7871-430-3.

**K. Wang Chen:** Transformation and symbolic calculations in filtering and control. Thesis No. 361, 1994. ISBN 91-7871-467-2.

**T. McKelvey:** Identification of state-space models from time and frequency data. Thesis No. 380, 1995. ISBN 91-7871-531-8.

**J. Sjöberg:** Non-linear system identification with neural networks. Thesis No. 381, 1995. ISBN 91-7871-534-2.

**R. Germundsson:** Symbolic systems – theory, computation and applications. Thesis No. 389, 1995. ISBN 91-7871-578-4.

**P. Pucar:** Modeling and segmentation using multiple models. Thesis No. 405, 1995. ISBN 91-7871-627-6.

**H. Fortell:** Algebraic approaches to normal forms and zero dynamics. Thesis No. 407, 1995. ISBN 91-7871-629-2.

 A. Helmersson: Methods for robust gain scheduling. Thesis No. 406, 1995. ISBN 91-7871-628-4.
 P. Lindskog: Methods, algorithms and tools for system identification based on prior knowledge. Thesis No. 436, 1996. ISBN 91-7871-424-8.

J. Gunnarsson: Symbolic methods and tools for discrete event dynamic systems. Thesis No. 477, 1997. ISBN 91-7871-917-8.

**M. Jirstrand:** Constructive methods for inequality constraints in control. Thesis No. 527, 1998. ISBN 91-7219-187-2.

**U. Forssell:** Closed-loop identification: Methods, theory, and applications. Thesis No. 566, 1999. ISBN 91-7219-432-4.

**A. Stenman:** Model on demand: Algorithms, analysis and applications. Thesis No. 571, 1999. ISBN 91-7219-450-2.

**N. Bergman:** Recursive Bayesian estimation: Navigation and tracking applications. Thesis No. 579, 1999. ISBN 91-7219-473-1.

**K. Edström:** Switched bond graphs: Simulation and analysis. Thesis No. 586, 1999. ISBN 91-7219-493-6.

**M. Larsson:** Behavioral and structural model based approaches to discrete diagnosis. Thesis No. 608, 1999. ISBN 91-7219-615-5.

**F. Gunnarsson:** Power control in cellular radio systems: Analysis, design and estimation. Thesis No. 623, 2000. ISBN 91-7219-689-0.

**V. Einarsson:** Model checking methods for mode switching systems. Thesis No. 652, 2000. ISBN 91-7219-836-2.

**M. Norrlöf:** Iterative learning control: Analysis, design, and experiments. Thesis No. 653, 2000. ISBN 91-7219-837-0.

**F. Tjärnström:** Variance expressions and model reduction in system identification. Thesis No. 730, 2002. ISBN 91-7373-253-2.

**J. Löfberg:** Minimax approaches to robust model predictive control. Thesis No. 812, 2003. ISBN 91-7373-622-8.

**J. Roll:** Local and piecewise affine approaches to system identification. Thesis No. 802, 2003. ISBN 91-7373-608-2.

**J. Elbornsson:** Analysis, estimation and compensation of mismatch effects in A/D converters. Thesis No. 811, 2003. ISBN 91-7373-621-X.

**O. Härkegård:** Backstepping and control allocation with applications to flight control. Thesis No. 820, 2003. ISBN 91-7373-647-3.

**R. Wallin:** Optimization algorithms for system analysis and identification. Thesis No. 919, 2004. ISBN 91-85297-19-4.

**D. Lindgren:** Projection methods for classification and identification. Thesis No. 915, 2005. ISBN 91-85297-06-2.

**R. Karlsson:** Particle Filtering for Positioning and Tracking Applications. Thesis No. 924, 2005. ISBN 91-85297-34-8.

**J. Jansson:** Collision Avoidance Theory with Applications to Automotive Collision Mitigation. Thesis No. 950, 2005. ISBN 91-85299-45-6.

**E. Geijer Lundin:** Uplink Load in CDMA Cellular Radio Systems. Thesis No. 977, 2005. ISBN 91-85457-49-3.